EXISTENCE OF SOLUTIONS AND APPROXIMATE CONTROLLABILITY OF SOME EVOLUTION EQUATIONS

Ph. D. THESIS

by

ARSHI MERAJ



DEPARTMENT OF MATHEMATICS INDIAN INSTITUTE OF TECHNOLOGY ROORKEE ROORKEE – 247 667 (INDIA) FEBRUARY, 2019

EXISTENCE OF SOLUTIONS AND APPROXIMATE CONTROLLABILITY OF SOME EVOLUTION EQUATIONS

A THESIS

Submitted in partial fulfilment of the requirements for the award of the degree

of

DOCTOR OF PHILOSOPHY

in

MATHEMATICS

by

ARSHI MERAJ



DEPARTMENT OF MATHEMATICS INDIAN INSTITUTE OF TECHNOLOGY ROORKEE ROORKEE - 247 667 (INDIA) FEBRUARY, 2019

©INDIAN INSTITUTE OF TECHNOLOGY ROORKEE, ROORKEE-2019 ALL RIGHTS RESERVED



INDIAN INSTITUTE OF TECHNOLOGY ROORKEE ROORKEE

CANDIDATE'S DECLARATION

I hereby certify that the work which is being presented in the thesis entitled "EXISTENCE OF SOLUTIONS AND APPROXIMATE CONTROLLABILITY OF SOME EVOLUTION EQUATIONS" in partial fulfillment of the requirements for the award of the Degree of Doctor of Philosophy and submitted in the Department of mathematics of the Indian Institute of Technology Roorkee, Roorkee is an authentic record of my own work carried out during a period from July, 2014 to July, 2019 under the supervision of Dr. D. N. Pandey, Associate Professor, Department of Mathematics, Indian Institute of Technology Roorkee, Roorkee.

The matter presented in this thesis has not been submitted by me for the award of any other degree of this or any other Institution.

(Arshi Meraj)

This is to certify that the above statement made by the candidate is correct to the best of my knowledge.

(D. N. Pandey) Supervisor

The Ph. D. Viva-Voce Examination of Arshi Meraj, Research Scholar, has been held on 08/07/2019.

Chairman, SRC

Signature of External Examiner

This is to certify that the student has made all the corrections in the thesis.

Signature of Supervisor (s)

Dated:July,2019

Head of the Department

Acknowledgement

"All praises and thanks are due to Almighty"

I am greatly indebted to whoever helped me in the accomplishment of my research work, especially to my supervisor Dr. Dwijendra Narain Pandey for his contribution of time, good behaviour, guidance, valuable suggestions, motivation and continuous support throughout my Ph. D. programme. I am also grateful to him for providing peaceful atmosphere and freedom to explore new things. He guided me in several ways to approach a research problem. I am indebted to Mrs. Richa Pandey for providing a positive environment and moral support during this period.

My thanks are due to Professor Nagarajan Sukavanam, Head of Department of Mathematics, IIT Roorkee, for providing valuable time for discussion and infrastructural facilities. I am indebted to all the faculty members, especially to my SRC members Professor Sunita Gakkhar, Dr. Ankik Kumar Giri and Dr. B. Venkata Manoj Kumar for their valuable time, suggestions and guidance to improve my skills and quality of research work. I am indebted to all non-teaching staff members of the department, hostels staff members and administration staff of IIT Roorkee, for their cooperation throughout my research work. My gratitude is due to "Ministry of Human Resource and Development, India" for providing financial assistance for my research work.

I am greatly indebted to my seniors Dr. Renu Chaudhary and Dr. Vikram Singh for supporting by all possible ways during my research work. I am grateful to all my seniors, juniors, friends and colleagues for their helping hands. I extend my gratitude to all my school and college teachers as well.

The never ending thanks to all my family members and relatives, especially to my parents Mr. Abdul Qaiyoom and Mrs. Razia Bano, husband Mr. Muhammed Shafi, parents-in-law, all sisters and 'Nani' for their unconditional love, care, continuous moral support and sincere wish for my success.

Arshi Meraj

Abstract

The work presented in this thesis deals with the investigation of the existence of mild solutions and approximate controllability of some fractional and integer order differential and integro-differential equations. To prove our results, we will use semigroup theory, evolution system, fixed point theorems, fractional calculus, measure of noncompactness, basic theory of functional analysis, and monotone iterative technique. The present work consists of the following eight chapters.

Chapter 1 contains a brief introduction to the problems which are discussed in the consecutive chapters and provides a motivational background to study the problems which are discussed in this thesis. Further, it contains a review of relevant literature and an outline of the thesis.

Chapter 2 contains some basic concepts of fractional calculus, functional analysis, semigroup theory and measure of noncompactness that will be required in the subsequent chapters.

Chapter 3 concerns with the study of a fractional nonlocal neutral integrodifferential equation having flux type integral boundary conditions. The existence and uniqueness results are proved by using Banach and Leray-Schauder nonlinear alternative fixed point theorems.

Chapter 4 contains fractional integro-differential equations having non-instantaneous impulses. The existence result is obtained by the help of fixed point theorem and

noncompact semigroup.

Chapter [5] consists of fractional nonlocal semilinear integro-differential equations having impulsive conditions for which the impulses are not instantaneous. The approximate controllability is proved via semigroup theory, Kuratowski measure of noncmpactness and ρ -set contractive fixed point technique, without imposing the condition of Lipschitz continuity on nonlinear term as well as the condition of compactness on impulsive functions and nonlocal function.

Chapter 6 contains deformable fractional differential equations. The results of existence and approximate controllability are obtained via semigroup theory, Schauder and Banach fixed point technique.

Chapter 7 considers non-autonomous semilinear differential equations having nonlocal conditions. The existence and uniqueness are obtained via monotone iterative method with the upper and lower solutions in an ordered complete norm space, using evolution system and measure of noncompactness.

Chapter 8 extends the results of chapter 7 for non-autonomous integro-differential equations having nonlocal conditions.

The relevant references are appended at the end.

Table of Contents

\mathbf{A}	cknowledgement	i
A	bstract	iii
Ta	able of Contents	v
N	omenclature	ix
1	Introduction	1
	1.1 Literature Survey	5
	1.1.1 Existence of solutions	5
	$1.1.2 \text{Controllability} \dots \dots$	10
	1.2 Organization of Thesis	14
2	Preliminaries	17
	2.1 Basic Concepts of Functional Analysis	17
	2.2 Semigroup Theory	20
	2.3 Basic Concepts of Fractional Calculus	22
	2.3.1 Solutions of Caputo Fractional Differential Equations	23
	2.4 Basic Concepts of Measure of Noncompactness	27
	2.5 Fixed Point Theorems	29
3	Fractional Neutral Integro-Differential Equations involving Nonlo-	
	cal Integral Boundary Conditions	31
	3.1 Introduction	31

	3.2 Preliminaries and Assumptions	. 32
	3.3 Existence and Uniqueness Results	. 35
	3.4 Examples	. 40
4	Fractional Non-Instantaneous Impulsive Integro-Differential Equ	a-
	tions with Nonlocal Conditions	43
	4.1 Introduction	. 43
	4.2 Preliminaries and Assumptions	. 44
	4.3 Main Result	. 46
	4.4 Example	. 51
5	Approximate Controllability of Nonlocal Non-Instantaneous Ir	
	pulsive Integro-Differential Equations of Fractional Order	53
	$5.1 \text{Introduction} \dots \dots \dots \dots \dots \dots \dots \dots \dots $. 53
	5.2 Preliminaries and Assumptions	. 55
	5.3 Main Results	. 58
	5.3.1 Existence of mild solutions	. 58
	5.3.2 Approximate controllability	. 65
	5.4 Example	. 66
6	Fractional Evolution Equations with Deformable Fractional Deriv	'a-
	tive	69
	6.1 Introduction	. 69
	6.2 Preliminaries	. 71
	6.3 Main Results	. 72
	6.3.1 Existence and uniqueness	. 72
	6.3.2 Approximate controllability	. 76
	6.4 Examples	. 85
		• • • • • •
7	Monotone Iterative Technique for Non-Autonomous Nonlocal D	
	ferential Equations	87

	7.1	Introdu	iction .		• •		•		•		•		•	•		•		•		•		•		•	•	87
	7.2	Prelimi	inaries		•••		•						•					•		•		•		•		88
	7.3	Main F	Results				• •		•		•		•	•						•				•		90
	7.4	Examp	le						•		•		•	•						•		•		•		97
0	Ман		Tt an a t	T	1			C	NT	~ ~ ~		4					~ 1	T.		1~		.	T	4		
ð	Mon	otone	Iterati	ve 1	ech	niq	ue	or	IN	on	-A	ut	or	101	mo	bu	S I	NC	n	10	ca	<u>.</u>	In	ιτe)gi	:0-

Differential Equations	99
8.1 Introduction	99
8.2 Preliminaries	100
8.3 Main Results	102
8.4 Example	106
List of Publications	109

List	of Publication	\mathbf{s}

Bibliography

110

Nomenclature

Notation	Description
\mathbb{R}	Set of Real numbers
N	Set of Natural numbers
\mathbb{Z}	Set of Integers
\mathbb{C}	Set of Complex numbers
$B_r(x,X)$	Ball centered at x with radius r in X
C([a,b],X)	Set of all continuous maps from $[a, b]$
	into X
$\mathcal{PC}([a,b],X)$	Set of all piecewise continuous maps
	from $[a, b]$ into X
$W^{m,p}([a,b],X)$	Sobolev space
$\Gamma(n)$	Euler's continuous gamma fuctions
J^q	Riemann-Liouville (in short RL) frac-
	tional integral operator of order \boldsymbol{q}
$^{L}\mathbf{D}^{q}$	Riemann-Liouville fractional differen-
	tial operator of order q
$^{C}\mathbf{D}^{q}$	Caputo fractional differential operator
	of order q
eta	Kuratowski measure of noncompact-
	ness
χ	Hausdroff measure of noncompactness
iff	if and only if

Chapter 1 Introduction

By the evolution equations, we mean the abstract formulation of the prototype of several problems of similar kind. In an infinite dimensional space of functions, an ordinary differential equation which is generally associated with a partial differential equation modeling a physical phenomenon may be considered as an evolution equation. Laser optics, reaction diffusion equations, climate models, control theory, neural networks, heat conduction and wave propagation in materials can be modeled as abstract evolution equations in a Hilbert space or in a Banach space. In modeling a real world problem as an evolution equation, initial conditions are managed as essential conditions while the boundary conditions are treated as natural conditions because those are fused with the domain of the operator and therefore do not appear in the abstract formulation. In this way, we able to focus on certain invariant properties of a class of such problems instead of examining individual problems.

Researchers in applied mathematics and engineering have found differential equations having nonlocal conditions invaluable because of their practical applications to various physical problems such as heat conduction, nonlocal reactive transport in underground water flows in porous media, biotechnology, population dynamics, investigation of pollution process in rivers, seas etc. Particularly, introduction of nonlocal conditions into a system can improve its qualitative and quantitative characteristics. These nonlocal conditions come into picture while taking measurements at regular intervals rather than continuously over the history period. Therefore, the analysis can be presented more practically by using nonlocal conditions instead of using classical initial conditions.

A differential equation in which the highest order derivative of the unknown function appears with and without deviations is called neutral differential equation. For example,

$$z'(t) - cz'(t - \varrho) - dz(t - \varrho) = 0, \ \varrho > 0, \ c \neq 0,$$
(1.0.1)

is a neutral differential equation of first order. Neutral differential equations appear in problems dealing with electric networks involving lossless transmission lines.

Fractional calculus, the generalization of traditional calculus, deals with the investigation and applications of derivatives and integrals of non-integer order. In particular, in this branch of mathematics, we study the methods of solving differential equations containing fractional derivatives of the unknown function, such equations are called fractional differential equations (in short FDEs). The analysis of nonlinear oscillations of an earthquake, continuum and statistical mechanics, relaxation in fluid polymers and the modeling of visco-plasticity are some of the fields in which employment of differential equations involving fractional derivative gives a more realistic analysis of the problem considered.

Due to involvement of integral operator in the definition of fractional derivative, fractional differential operator is a non-local operator. That is, a fractional derivative at a certain point in space or time consists of information about the function at previous points in space or time, respectively. Therefore, fractional differential equations describe the hereditary and memory properties of various materials and phenomena. For example, viscoelastic materials and polymers which are related to systems with memory can be efficiently described with fractional differential equations.

Various evolutionary processes such as population dynamics, orbital transfer of

satellites and sampled-data systems are characterized by the abrupt changes in their state. These abrupt changes occur for a very short interval of time and can be approximated in terms of instantaneous changes of state, i.e., impulses. Such processes can be appropriately modeled by impulsive differential equations. In past few years, the theory of impulsive differential equation have been emerged as a beneficial tool, which can precisely define mathematical model of various realistic situations, for example biological phenomena which involves optimal control models in economics, thresholds, bursting rhythm models in medicine.

Generally, the impulses start abruptly for very short duration of time that can be negligible in comparison to the overall process. But instantaneous impulses are failed to describe the certain dynamics of evolution processes in pharmacotherapy. The introduction of the drugs in the bloodstream and the resultant absorption to the body are gradual and continuous processes. To characterize these type of situations, Hernàndez and O'Regan [95] introduced new type of impulses which triggers abruptly and remains active during finite time. These type of impulsive conditions are called non-instantaneous impulsive conditions.

For most of the differential equations, it is difficult to find exact solutions in closed forms. To overcome this difficulty, many numerical and analytical techniques have been designed for example, the homotopy analysis method and the Adomian decomposition method have been applied to integrate various systems of fractional order. However, in recent years, considerable work has been done using monotone iterative technique (in short MIT), which is a productive procedure for proving existence results in a closed interval formed by the lower and upper solutions. In MIT, by choosing upper and lower solutions as two initial iterations, one may construct two monotone sequences which converge monotonically from above and below, respectively, to a solution of the problem. Ladde et al. [120] has described a comprehensive introduction to the monotone iterative techniques in their book "Monotone iterative techniques for nonlinear differential equations".

Process of influencing the behavior of a dynamical system to achieve a given goal can be described as the concept of control. Many physical systems can be controlled by manipulating their inputs based on the observation of the outputs, for example an airplane is controlled by the pilot's action based on instrument reading and visual observations. The control problem is to determine the necessary input to achieve a desired goal.

Consider the case of the vibrating system consisting of a single mass on linear spring , we obtain the differential equation of motion

$$\frac{d^2x}{dt^2} = -x$$

where x is displacement from equilibrium at time t. Can we introduce a control force u depending on x and $\frac{dx}{dt}$ so that every solution of

$$\frac{d^2x}{dt^2} = -x + u$$

returns to the rest equilibrium state x = 0, $\frac{dx}{dt} = 0$ after a finite duration? This is the problem of controllability.

Controllability theory originates from the famous work done by Kalman [107] which leads to very important conclusion regarding the behavior of linear and nonlinear dynamical systems. In case of infinite dimensional systems, basically there are two main concepts of controllability i.e. exact (complete) controllability and approximate controllability. Exact controllability steers the system to arbitrary desirable state while approximate controllability means that the system can be steered to arbitrary small neighborhood of desirable state. The exact controllability is stronger notion than approximate controllability, but approximate controllability is completely adequate in applications. However, in case of finite dimensional systems notions of exact and approximate controllability coincide.

1.1 Literature Survey

1.1.1 Existence of solutions

Many physical phenomena such as flow of fluid through fissured rocks [25], nonlinear oscillations of earthquake, relaxation phenomena in complex viscoelastic material [86], propagation of waves in viscoelastic media [136] can be described by fractional differential equations. Fractional derivative was first mentioned in a letter correspondence between Leibniz and L'Hospital in 1695. Later on, many famous mathematicians e.g. Euler, Laplace, Fourier, Abel, Grünwald, Riemann, Liouville, Caputo etc. provided a lot of contribution in this field. For basic theory of FDEs one may see the books [114; 140; 158; 167].

In [76], author studied existence, uniqueness and some properties of the solution of the equation

$$\frac{d^q x(t)}{dt^q} = f(t, x(t)), \quad t > 0, \tag{1.1.1}$$

where $q \in (0, 1)$. El-Sayed [77] studied existence, uniqueness, smoothness and continuation to the solution of initial value problem corresponding to (1.1.1) for q > 0. Delbosco [67] reduced equation (1.1.1) into an integral equation with weak singularity and applied basic techniques of nonlinear analysis to discuss the existence results. El-Sayed [78] studied the existence results for the diffusion wave equation of fractional order

$$D^{q}z(t) = Az(t), \quad t > 0, \quad q \in (0,2],$$
(1.1.2)

where A is generator of an analytic semigroup $\mathcal{S}(t)$. Equation (1.1.2), represents the diffusion equation when $q \to 1$ and the wave equation when $q \to 2$.

In [75], El-Borai introduced mild solution in terms of probability density function with Laplace transform to Cauchy problem in a Banach space X. El-Borai [74] studied the existence results for the FDE

$$D^{q}z(t) = Az(t) + F(t, B_{1}(t)z(t), \dots, B_{m}(t)z(t)) \quad t > 0, q \in (0, 1],$$
(1.1.3)

where A generates an analytic semigroup, function F satisfies uniformly Hölder continuity in t and $B_j(t), j = 1, 2, ... m$ are densely defined closed linear operators on X. In [153] Nyamoradi et al. investigated results for a fractional order differential inclusion via suitable fixed point theorems. In [203] and [204], Zhou obtained various results on solutions for fractional evolution equations. Later on, utilizing the concept of mild solution introduced by Zhou [203], many authors studied different type of fractional differential equations, see [1]; [8]; [19]; [28]; [40]; [43]; [45]; [65]; [71]; [72]; [83]; [96]; [135]; [146]; [162]; [190] and the references therein.

Recently, Khalil [112] introduced a new fractional derivative named as conformable fractional derivative which seems to be a natural extension of the classical derivative. In [112], authors gave the definition of conformable fractional derivative and corresponding fractional integral also they proved the basic properties of conformable derivative, Rolle's Theorem and Mean Value Theorem for conformable derivative. For more details one may see [2]; [3]; [14]; [15]; [30]; [93]; [101]; [154]; [193]; [199] and the references therein. In [205], authors defined a new derivative called as deformable derivative which is simpler than conformable derivative and ranges over wider class of functions.

The nonlocal problem was motivated by physical problems. It is used to represent mathematical models for evolution of various phenomena such as nonlocal neutral networks, nonlocal pharmacokinetics, nonlocal pollution and nonlocal combustion (see [138]). The nonlocal problem was first studied by Byszewski [37]. Deng [68] used the nonlocal conditions to describe diffusion phenomenon of an small amount of gas. Byszewski [37], Byszewski and Lakshmikantham [38] and Jackson [100] have generalized the Cauchy problem with classical initial condition to the Cauchy problem with nonlocal initial conditions. Later, many authors stududied the differential equations with nonlocal conditions, see [8; 16; 22; 28; 40; 45; 65; 70; 71; 152; 186; 187] and references therein.

Boundary value problems (in short BVPs) concerning fractional differential equations having integral boundary conditions are applicable in several fields such as thermoelasticity, chemical engineering, blood flow problem, underground water flow, cellular systems, heat transmission, plasma physics, population dynamics and so on. For more details regarding these boundary conditions, one may see the papers [6; 7; 9; 10; 35; 39; 50; 105; 106; 119].

Mouffak **[142]** considered neutral impulsive differential equations of first and second order to investigate the existence of solutions. He used Schaefer fixed point theorem to obtain the results. In **[5]**, Shruti considered a class of neutral differential equations with nonlocal conditions to investigate the existence of solution. She proved the results with the help of Schauder fixed point theorem. Zhou **[203]** obtained various results for the existence and uniqueness associated to neutral differential equations of fractional order. Alka **[41]** discussed the existence of solutions for fractional neutral delay differential system having nonlocal conditions by using resolvent operator, Banach and Krasnoselskii fixed point theorems. In **[202]**, Zhou investigated the existence, uniqueness and dependence of solution on initial value by iterative technique for linear neutral fractional differential equations with constant coefficient. Malik **[137]** considered non-autonomous neutral differential system of fractional order and studied the existence results via resolvent operators and Banach contraction principle. In **[97]** results are obtained for neutral delay differential equations via contraction mapping fixed point theorem and semigroup theory.

In the past decades, many researchers paid attention to study the differential equations with instantaneous impulses, which have been used to describe abrupt changes such as shocks, harvesting and natural disasters. Particularly, the theory of instantaneous impulsive equations have wide applications in control, mechanics, electrical engineering, biological and medical fields. In order to solve impulsive fractional differential equations there are two main approaches. The first approach (also called multiple base point approach) was introduced by Benchohra and Slimani [33] in which they considered a fractional Cauchy problem with impulsive effects, and developed the method to find the existence and uniqueness of solutions by using several fixed point theorems. The second approach (also called the single base point approach) was introduced by Fečkan et al. [81] in which they developed single base point approach to find existence results via fixed point techniques. Wang [160] considered semilinear fractional differential equation with impulsive conditions, and introduced the new concept of mild solution using semigroup, Laplace transform and probability density function. For more details, one may see the monographs [31], [122], [168], papers [42], [43], [44], [51], [63], [103], [125], [126], [170], [171], [183].

Hernández et al. [95] initially studied Cauchy problems of first order with noninstantaneous impulses. Pradeep et al. [117] considered fractional non-instantaneous impulsive system to investigate the existence results. In [116] results of [117] are extended for non-instantaneous impulsive integro-differential equations of fractional order. Muslim [147] considered second order differential equations with deviated argument having non-instantaneous impulses, and results are obtained by using strongly continuous cosine family of linear operators and Banach fixed point theorem. The recent results for evolution equations with non-instantaneous impulses can be found in [17]; [18]; [47]; [57]; [113]; [117]; [151]; [157]; [185]; [191]; [194] and the references therein.

Du [69] proposed a monotone iterative technique (in short MIT) for ordinary differential equations in an ordered Banach space X. He proved the existence of minimal and maximal mild solutions lying between the lower and upper solutions. In [98] and [99], Hristova and Bainov applied MIT to functional and impulsive differential equations. In [173], Sun improved the result of Du [69] by removing the measure of noncompactness condition on function f. In [174], Sun and Zhao improved the result of [173] further and investigated the results for integro-differential equations in a weakly sequentially complete Banach space X to find extremal solutions. In [89], Guo and Liu used MIT to find the existence of extremal solutions of mixed type impulsive integro-differential equation. Later, Li [124] improved the result of Guo and Liu [89] by using Bellman inequality.

Chen and Li **52** used monotone iterative technique for nonlocal differential equations. Chen and Mu **56** considered impulsive integro-differential equations and obtained results via monotone iterative method. Mu **143** applied MIT for fractional differential equations. In **145**, Mu and Li used MIT for impulsive differential equations of fractional order. Later on, the result has been extended for nonlocal condition by Mu **144**. Chen et al. **54** considered nonlocal impulsive semilinear system and applied perturbation method and MIT to obtain the results. In **53** a differential equation with mixed monotone nonlocal conditions of fractional order is considered to study the existence results via coupled lower and upper mild *L*-quasi solutions and monotone iterative technique. Kamaljeet **109** and Renu **49** used MIT for nonlocal differential equations having infinite delay of fractional order respectively. Chen et al. **61** used perturbation method and iterative technique for evolution equations involving non-instantaneous impulses. For more details on MIT, one may see **55 128**; **139**; **177**; **188**; **197**.

Yan [189] considered non-autonomous integro-differential equations involving nonlocal conditions to study the existence results via theory of evolution system, Banach and Schauder fixed point theorems (in short FPTs). Haloi et al. [92] studied existence, uniqueness and asymptotic stability of non-autonomous differential equations with deviated arguments via Banach fixed point theorem and theory of analytic semigroup. In 46, Alka et al. established the existence and uniqueness results for non-autonomous differential equations having iterated deviating arguments and instantaneous impulsive conditions. Borai [73] first derived the sufficient conditions for the existence of solution for a fractional non-autonomous Cauchy problem. Li 123 considered fractional non-autonomous integro-differential equations involving nonlocal condition. Chen et al. 58 generalized the results of 73 for fractional non-autonomous evolution equations with delay by using FPT with respect to ρ -set contractive operator. Nieto 32 considered second order non-autonomous evolution equations having nonlocal conditions, and established the sufficient conditions for the existence of solutions by using measure of noncompactness and Tikhonoff fixed point theorem. In <u>60</u>, Chen et al. considered fractional nonlocal non-autonomous differential equations, and established the existence of mild solutions via theory of evolution families and FPT with respect to ρ -set contractive operator. In [59], authors first studied the local existence of mild solutions, then obtained a blowup alternative result for fractional non-autonomous integro-differential equation of Volterra type.

1.1.2 Controllability

The theory of controllability is introduced by Kalaman (1960) [107]. For the basic theory of controllability in finite and infinite dimensional spaces one may see [20]; [26]; [34]; [62]; [195]. Fattorini (1966) [79] studied approximate controllability of linear control system x'(t) = Ax(t) + Bu(t) in infinite dimensional reflexive Banach space, by assuming that A is a linear densely defined closed operator which generates a C_0 semigroup. In [80], he discussed approximate controllability in separable Hilbert space by assuming that A is self adjoint and semibounded above operator.

Triggiani (1975) [179] considered infinite dimensional linear control system to study the approximate controllability in separable Banach space under the assumption that the operator A is bounded. He also proved that infinite dimensional linear control system can never be exactly controllable in finite time by using L_1 controls, if either the operator B is compact and the state space has a Schauder basis or range space of B is of finite dimension. In [180], he improved the results of [179] by removing the assumption that the state space has a basis. In [181], author proved that linear control system in infinite dimensional space can never achieve exact controllability in finite time by using L_1 control, if C_0 semigroup generated by A is compact.

Zhou (1983) [200] considered a class of semilinear system $x'(t) + Ax(t) = \mathcal{F}(x(t)) + Bu(t)$ in a Hilbert space, where -A generates a differentiable semigroup. He assumed that the function \mathcal{F} satisfies Lipschitz continuity, and obtained approximate controllability of the system by imposing an inequality condition on the range of operator B as well as approximate controllability of associated linear system. In [201], he studied approximate controllability for generalized system $x'(t) + Ax(t) = \mathcal{F}(x(t), u(t)) + Bu(t).$

Naito (1987) **[148]** proved approximate controllability for semilinear system under the assumptions that -A generates a C_0 semigroup, nonlinear function is uniformly bounded as well as Lipschitz continuous, and some condition on the range of operator B. In [149], he removed the condition of uniformly bounded from nonlinear function and introduced an inequality condition containing the parameters: K > 0 (Lipschitz constant of \mathcal{F}), T > 0 (control time), $M \ge 1$ (bound of C_0 semigroup), and $||P|| \ge 1$ (norm of projective type operator P introduced by estimating the control efficiency of operator B) with the assumption that $\mathcal{F}(0) = 0$.

Joshi et al. (1989) [104] considered non-autonomous semilinear control system in finite dimensional space. They discussed the controllability of the system by using contraction mapping principle, monotone operator theory, and nonlinearities of both Lipschitian and non-Lipschitzian types. George (1995) [85] investigated the approximate controllability of infinite dimensional non-autonomous semilinear control system by using monotone operator theory, evolution operator, and nonlinearities of both Lipschitian and non-Lipschitzian types. Bashirov et al. (1999) [29] determined new necessary and sufficient conditions for a linear system to be approximately and exactly (complete) controllable.

Dauer et al. (2002) **[64]** considered semilinear system to study approximate and complete controllability. The approximate controllability is proved via compact semigroup and Schauder fixed point technique, complete controllability is attained with the help of non-compact semigroup and Banach contraction principle. Mahmudov (2003) **[131]** obtained the approximate controllability for semilinear deterministic and stochastic systems by using compact semigroup, properties of symmetric operators, Banach and Schauder fixed point theorems with the assumption that associated linear system is controllable. Sakthivel et al. (2010) **[164]** considered semilinear impulsive control system with state delay, to investigate the approximate controllability. Haloi (2017) **[91]** considered non-autonomous delay system having deviated arguments. He used Krasnoselskii fixed point theorem and evolution system to prove the approximate controllability.

Balachandran et al. (2009) [21] studied controllability for fractional semilinear integro-differential equations by using Banach contraction principle, semigroup theory, and theory of fractional calculus. Tai et al. (2009) [175] considered fractional semilinear neutral impulsive integro-differential system having infinite delay. They determined sufficient conditions for the controllability of the system with the help of Krasnoselskii fixed point theorem, theory of fractional calculus and semigroup. Sakthivel et al. (2011) [166] discussed approximate controllability for a class of fractional semilinear system. They obtained the results by using theory of fractional power, semigroup, and Schauder fixed point theorem. In [165] (2012), authors studied exact controllability for semilinear fractional neutral delay differential system with local and nonlocal conditions by using Banach contraction principle, semigroup theory, and approximate controllability of corresponding linear system.

Kumar et al. (2013) **118** discussed the exact controllability of fractional semilinear system. To obtain the results, they used regular integral contractor, iterative technique, and the controllability of corresponding linear system. Mahmudov et al. (2014) **132** considered fractional semilinear integro-differential system, and established sufficient conditions for the approximate controllability by the help of Schauder fixed point theorem, theory of semigroup and fractional calculus. In [23] (2014), Balasubramaniam et al. generalized the results of [132] for fractional impulsive integro-differential equations with nonlocal conditions, by assuming the compactness of impulsive and nonlocal functions in a Hilbert space. The results are proved with the help of Darbo-Sodovskii fixed point theorem, theory of semigroup and fractional calculus. Zhang et al. (2015) **198** discussed the approximate controllability of fractional impulsive integro-differential equations in a Hilbert space with the help of Krasnoselskii fixed point theorem and comapact analytic semigroup theory. In [169] (2015), authors determined sufficient conditions for the controllability of fractional semilinear non-autonomous neutral differential system. They used Krasnoselskii fixed point theorem, theory of semigroup and fractional power of operators.

In [127] (2015), Liu et al. considered fractional semilinear control system with Riemann-Liouville fractional derivative first time. They derived the expression of mild solution, and established sufficient conditions for the approximate controllability via generalized Banach contraction theorem by assuming that the semigroup is differentiable. Dong et al. (2016) [84] studied approximate controllability of semilinear impulsive fractional evolution equations via approximate technique. Kamaljeet et al. (2016) [110] considered fractional neutral semilinear integro-differential equations having nonlocal conditions and finite delay. They proved approximate controllability of the system via fractional power of operators, semigroup, and Krasnoselskii fixed point technique. Urvashi et al. (2017) **[13]** dealt with fractional nonlocal semilinear delay differential system. They proved exact controllability by the help of Nussbaum fixed point theorem, fractional calculus and semi group theory. Wang et al. (2017) **[184]** studied the approximate controllability of Sobolev-type nonlocal fractional semilinear system. For more literature on controllability, one may see **[4] [27] [108] [48] [111]**; **[102] [115]**; **[129]**; **[130]**; **[133]**; **[134]**; **[141]**; **[155]**; **[159]**; **[161]**; **[172]**; **[176]**; **[182]**; **[192]** and the references therein.

1.2 Organization of Thesis

In this thesis, we divide our research work into three main parts.

In first part, we investigated the existence results to integro-differential equations of fractional order having nonlocal conditions. The main tools for this part are fixed point theorems, measure of noncompactness and semigroup theory.

In second part, we discussed the approximate controllability of differential systems with Caputo and deformable fractional derivative.

In third part, we studied the existence of extremal mild solutions to integer order non-autonomous nonlocal differential systems with the help of evolution system and monotone iterative technique.

The thesis contains following chapters.

Chapter 1 provides a brief introduction and motivational background to the problems considered in the consecutive chapters. Literature review associated to the work done in this area also has been discussed.

Chapter 2 contains some basic concepts of fractional calculus, semigroup theory, functional analysis and measure of noncompactness that are basically prerequisites for our work.

Chapter 3 concerns with the study of existence results for fractional neutral integro-differential equations having nonlocal flux type integral boundary conditions.

The contents of this chapter are published in Malaya Journal of Matematik.

Chapter ⁴ consists of fractional semilinear integro-differential equations having nonlocal and non-instantaneous impulsive conditions, and the existence result is obtained via noncompact semigroup and fixed point theorem.

The contents of this chapter are published in Arab Journal of Mathematical Sciences (Elsevier Publications).

Chapter $\mathbf{5}$ contains fractional semilinear integro-differential equations having nonlocal and impulsive conditions for which the impulses are not instantaneous. The existence result and approximate controllability are obtained by using semigroup theory, Kuratowski measure of noncmpactness and ρ -set contractive fixed point theorem, without imposing the condition of Lipschitz continuity on nonlinear term as well as the condition of compactness on impulsive functions and nonlocal function.

The contents of this chapter are accepted in **Journal of Fractional Calculus** and **Applications**.

Chapter 6 contains deformable fractional differential equations. The existence and approximate controllability results are obtained via semigroup theory, Banach and Schauder fixed point theorems.

The contents of this chapter are accepted in **Journal of Nonlinear Evolution** Equations and Applications.

Chapter 7 consists of non-autonomous semilinear differential equations having nonlocal conditions. The existence and uniqueness results are established via monotone iterative method, using evolution system and measure of noncompactness.

The contents of this chapter are published in **Demonstratio Mathematica** (DE GRUYTER).

Chapter 8 consists of non-autonomous integro-differential equations having nonlocal conditions. The monotone iterative method is applied to get the existence

and uniqueness results.

The contents of this chapter are accepted in **Filomat**.

Chapter 2 Preliminaries

In this chapter, we provide some basic concepts of functional analysis, fractional calculus, semigroup theory and measure of noncompactness which serve as prerequisites for subsequent chapters.

2.1 Basic Concepts of Functional Analysis

Let $(\mathbb{X}, \|\cdot\|_{\mathbb{X}})$ and $(\mathbb{Y}, \|\cdot\|_{\mathbb{Y}})$ be two Banach spaces. $(\mathbb{B}(\mathbb{X}, \mathbb{Y}), \|\cdot\|_{\mathbb{B}(\mathbb{X}, \mathbb{Y})})$ denotes the space of all linear bounded operators from \mathbb{X} to \mathbb{Y} and for $\mathbb{Y} = \mathbb{X}$, we write $\mathbb{B}(\mathbb{X})$. In similar fashion, $(\mathcal{L}(\mathbb{X}, \mathbb{Y}), \|\cdot\|_{\mathcal{L}(\mathbb{X}, \mathbb{Y})})$ stands for the space of all linear operators from \mathbb{X} to \mathbb{Y} . In the case $\mathbb{Y} = \mathbb{X}$, we write $\mathcal{L}(\mathbb{X})$. If A denotes the linear operator on \mathbb{X} , then the domain, null and range space of A are denoted by D(A), N(A) and R(A)respectively. Let us denote $L^p(\mathcal{I}, \mathbb{X})$ for Banach space of all Bochner-measurable functions $F: \mathcal{I} \to \mathbb{X}$ with the norm

$$||F||_{L^{p}(\mathcal{I},\mathbb{X})} := \begin{cases} (\int_{\mathcal{I}} ||F(s)||_{\mathbb{X}}^{p} ds)^{1/p}, & 1 \le p < \infty, \\ \sup_{t \in \mathcal{I}} ||F(t)||_{\mathbb{X}}, & p = \infty, \end{cases}$$
(2.1.1)

where $\mathcal{I} = (a, b)$ with $-\infty < a < b < \infty$.

The notations $C(\mathcal{I}, \mathbb{X})$ and $C^m(\mathcal{I}, \mathbb{X})$ stand for the space of all continuous and *m*-times continuously differentiable functions, respectively. Set $J = [0, T], T < \infty$. Then, $C(J, \mathbb{X})$ and $C^m(J, \mathbb{X})$ are the Banach spaces equipped with the norm

$$||F||_C := \sup_{t \in J} ||F(t)||_{\mathbb{X}}, \quad ||F||_{C^m} := \sup_{t \in J} \sum_{k=0}^m ||F^k(t)||_C, \quad (2.1.2)$$

respectively.

Now, we state some important inequalities :

• Hölder inequality: Let $p \in [1, \infty)$ and q is such that 1/p + 1/q = 1. Then

$$\|(FG)\|_{L^{1}(\mathcal{I},\mathbb{X})} \leq \|F\|_{L^{p}(\mathcal{I},\mathbb{X})} \|G\|_{L^{q}(\mathcal{I},\mathbb{X})}, \qquad (2.1.3)$$

where $F \in L^p(\mathcal{I}, \mathbb{X}), G \in L^q(\mathcal{I}, \mathbb{X}).$

• Young inequality: Assume $F \in L^p(\mathcal{I}, \mathbb{X})$, $G \in L^q(\mathcal{I}, \mathbb{X})$ and 1/q + 1/p = 1/r + 1 such that $1 \leq q, p, r \leq \infty$. Then

$$\|F * G\|_{L^{p}(\mathcal{I},\mathbb{X})} \leq \|F\|_{L^{p}(\mathcal{I},\mathbb{X})} \|G\|_{L^{q}(\mathcal{I},\mathbb{X})}, \qquad (2.1.4)$$

where " \ast " denotes the convolution.

• Gronwall's inequality: Let F and G be the non-negative continuous functions for each $t \ge t_0$ and β be a constant. Then the inequality

$$F(t) \le \beta + \int_{t_0}^t G(s)F(s)ds, \quad t \ge t_0,$$
 (2.1.5)

implies the following inequality

$$F(t) \le \beta \exp(\int_{t_0}^t G(s)ds), \quad t \ge t_0.$$
 (2.1.6)

• Generalized Gronwall's inequality [90]: Let F(t) and G(t) be locally integrable nonnegative functions on $0 \le t < T < \infty$ and $a \ge 0, \beta > 0$ be two constants with

$$F(t) \leqslant G(t) + a \int_0^t (t-\varrho)^{\beta-1} F(\varrho) d\varrho.$$
(2.1.7)

Then

$$F(t) \leqslant G(t) + \int_0^t \left[\sum_{n=1}^\infty \frac{(a\Gamma(\beta))^n}{\Gamma(n\beta)} (t-\varrho)^{n\beta-1} G(\varrho) \right] d\varrho, \quad 0 \leqslant t < T. \quad (2.1.8)$$

Definition 2.1.1. Suppose that I = (0,T), or $I = \mathbb{R}$, or $I = \mathbb{R}^+$, $m \in \mathbb{N}$ and $1 \leq p < \infty$. The Sobolev spaces $W^{m,p}$ is defined as

$$W^{m,p}(I,\mathbb{X}): = \{F: there \ exists \ z \in L^p(I,\mathbb{X}): F(t) = \sum_{k=0}^{m-1} c_k \frac{t^k}{k!} + \frac{t^{m-1}}{(m-1)!} * z(t), \ t \in I\},$$
(2.1.9)

where $z(t) = F^{m}(t), c_{k} = F^{k}(0)$. Also

$$W_0^{m,p}(I,\mathbb{X}) := \{ F \in W^{m,p}(I,\mathbb{X}) : F^k(0) = 0, \ k = 0, 1, \cdots, m-1 \}.$$
 (2.1.10)

Definition 2.1.2. The Laplace transform of a function $G \in L^1(\mathbb{R}^+, \mathbb{X})$ is defined as

$$\widehat{G}(\lambda) := \int_0^\infty e^{-\lambda t} G(t) dt.$$
(2.1.11)

Definition 2.1.3. Suppose that $\widetilde{\mathbb{X}}$ and \mathbb{X} be Banach spaces. A mapping $F : \widetilde{\mathbb{X}} \to \mathbb{X}$ is known as **Lipschitz continuous** if there exists a constant L > 0 such that

$$\|F(v_1) - F(v_2)\|_{\mathbb{X}} \le L \|v_1 - v_2\|_{\widetilde{\mathbb{X}}}, \text{ for all } v_1, v_2 \in \widetilde{\mathbb{X}}.$$
 (2.1.12)

Definition 2.1.4. A function $F : \widetilde{\mathbb{X}} \to \mathbb{X}$ is said to be a Hölder continuous if there exist nonnegative real constants C and $\theta \in (0, 1]$ such that

$$\|F(z_1) - F(z_2)\|_{\mathbb{X}} \le C \|z_1 - z_2\|_{\widetilde{\mathbb{X}}}^{\theta} \text{ for each } z_1, z_2 \in \widetilde{\mathbb{X}}.$$
 (2.1.13)

The number θ is called the exponent of the Hölder condition. If $\theta = 1$, then the function satisfies a Lipschitz condition. If $\theta = 0$, then the function is simply bounded.

Definition 2.1.5. A family of functions F defined on a set E in X is called **equicon**tinuous on E if for given $\epsilon > 0$, we can find a $\delta > 0$ in a way that

 $||f(z_0) - f(z)|| < \epsilon$ whenever $||z_0 - z|| < \delta$, $z_0, z \in E$ and $f \in F$. (2.1.14)

Definition 2.1.6. Suppose that $F : \mathbb{X} \to \mathbb{X}$ be a nonlinear operator. Then solution of the equation

$$F(z) = z, \quad z \in \mathbb{X}, \tag{2.1.15}$$

is called a fixed point of operator F.

Definition 2.1.7. A mapping F from a subset B of a normed space X into X is called a contraction mapping if we can find a number 0 < C < 1 in a way that

$$||F(z_1) - F(z_2)|| \le C ||z_1 - z_2||, \text{ for all } z_1, z_2 \in B.$$
(2.1.16)

Note that contraction mappings are continuous.

Definition 2.1.8. Suppose that \mathbb{X} and $\widetilde{\mathbb{X}}$ be normed linear spaces. An operator $\mathbb{T}: \widetilde{\mathbb{X}} \to \mathbb{X}$ is called **compact** if it maps every bounded subset of $\widetilde{\mathbb{X}}$ into a relatively compact subset of \mathbb{X} .

Theorem 2.1.1. (Arzela-Ascoli theorem) Let N be a compact metric space and $B \subset C(N)$. Then, B is totally bounded in C(N) if and only if B equicontinuous and pointwise bounded on N.

Theorem 2.1.2. Let $\widetilde{\mathbb{X}}$ and \mathbb{X} be two normed linear spaces. A linear operator $\mathbb{T}: \widetilde{\mathbb{X}} \to \mathbb{X}$ is compact if and only for every bounded sequence (z_n) in $\widetilde{\mathbb{X}}$ the sequence $(\mathbb{T}(z_n))$ in \mathbb{X} has a convergent subsequence.

2.2 Semigroup Theory

Definition 2.2.1. A one parameter family $\{S(t)\}_{t\geq 0}$, of bounded linear operators from Banach space X into X is called a semigroup if

- (1) $\mathcal{S}(0) = I$ (*I* is the identity operator on X).
- (2) $S(t_1 + t_2) = S(t_1)S(t_2), \forall t_1, t_2 \ge 0.$

Definition 2.2.2. A semigroup $\{S(t)\}_{t\geq 0}$ is called strongly continuous semigroup or C_0 -semigroup if

$$\lim_{t \downarrow 0} \mathcal{S}(t)z = z, \quad for \ every \ z \in \mathbb{X}.$$
(2.2.1)

Definition 2.2.3. If $\lim_{t_1\to t_2} || \mathcal{S}(t_1) - \mathcal{S}(t_2) || = 0$, the semigroup $\{\mathcal{S}(t)\}_{t\geq 0}$ is called uniformly continuous semigroup.

Definition 2.2.4. A linear operator A on X defined by

$$Az = \lim_{t \downarrow 0} \frac{\|S(t)z - z\|}{t}, \text{ for } z \in D(A),$$
(2.2.2)

is named as infinitesimal generator of a semigroup $\{S(t)\}_{t\geq 0}$ whenever the above limit exists. The domain of A is

$$D(A) = \{ z \in \mathbb{X} : \lim_{t \downarrow 0} \frac{\| \mathcal{S}(t)z - z \|}{t} \text{ exists} \}.$$

$$(2.2.3)$$

Theorem 2.2.1. If S(t) is a C_0 -semigroup, then we can find constants $\delta \geq 0$ and $M \geq 1$ in a way that

$$\|\mathcal{S}(t)\| \le M e^{\delta t}, \quad \forall t \ge 0.$$
(2.2.4)

Remark 2.2.2. If $\delta = 0$, S(t) is called uniformly bounded, and if M = 1, S(t) is called C_0 -semigroup of contractions.

Theorem 2.2.3. [156] Let S(t) be a C_0 -semigroup of bounded linear operators on \mathbb{X} which is generated by A. Then,

- (1) there is a > 0 such that $\|\mathcal{S}(t)\|$ is uniformly bounded on [0, a],
- (2) for each $z \in \mathbb{X}$, $\lim_{h \to 0} \frac{1}{h} \int_{t}^{t+h} \mathcal{S}(s) z ds = \mathcal{S}(t) z$,
- (3) for all $z \in \mathbb{X}$, $\int_0^t \mathcal{S}(s) z ds \in D(A)$ and

$$A(\int_0^t \mathcal{S}(s)zds) = \mathcal{S}(t)z - z, \qquad (2.2.5)$$

(4) for $z \in D(A)$, $S(t)z \in D(A)$, moreover

$$\frac{d}{dt}\mathcal{S}(t)z = \mathcal{S}(t)Az = A\mathcal{S}(t)z, \qquad (2.2.6)$$

(5) for all $z \in D(A)$,

$$\mathcal{S}(t)z - \mathcal{S}(s)z = \int_{s}^{t} \mathcal{S}(\tau)Azd\tau = \int_{s}^{t} A\mathcal{S}(\tau)zd\tau.$$
(2.2.7)

Corollary 2.2.4. [156] Suppose that A is the infinitesimal generator of a C_0 -semigroup. Then, A is a linear closed operator and D(A) is dense in X.

The resolvent set $\rho(A)$ of a linear operator A consists all $\lambda \in \mathbb{C}$ such that $(\lambda I - A)^{-1}$ exists and is a linear bounded operator on X. $R(\lambda, A) = (\lambda I - A)^{-1}, \lambda \in \rho(A)$ is known as family of resolvent operators of A.

Theorem 2.2.5. [156] (Hille-Yosida Theorem) A linear operator A is the infinitesimal generator of an strongly continuous semigroup S(t) ($t \ge 0$) satisfying $||S(t)|| \le Me^{\delta t}$, iff A is densely defined in X and closed, $\rho(A)$ contains (δ, ∞), and for $\lambda > \delta$

$$||R(\lambda, A)^k|| \le \frac{M}{(\lambda - \delta)^k}, \ \forall \ k = 1, 2, \dots$$
 (2.2.8)

Theorem 2.2.6. [156] (Hille-Yosida Theorem for Contraction Semigroups) Suppose that $A : D(A) \subset \mathbb{X} \to \mathbb{X}$ is a linear operator. Then, A is the infinitesimal generator of an strongly continuous semigroup of contraction iff A is densely defined in \mathbb{X} and closed, $\rho(A)$ contains \mathbb{R}^+ , and

$$\|(\lambda I - A)^{-1}\| \le \frac{1}{\lambda}, \text{ for all positive } \lambda.$$
(2.2.9)

Now, we state the following results which provide the representation of the semigroup generated by a bounded linear operator. **Theorem 2.2.7.** Suppose that A be a linear operator and densely defined in X which satisfies the following two conditions :

- (1) $\sum_{\mu} = \{\lambda : |arg\lambda| < \frac{\pi}{2} + \mu\} \cup \{0\} \subset \rho(A), \text{ for some } 0 < \mu < \pi/2;$
- (2) there is a constant \mathcal{L} in such a way that

$$(\lambda I - A)^{-1} \le \frac{\mathcal{L}}{|\lambda|}, \quad \lambda \in \sum_{\mu} \quad and \quad \lambda > 0.$$

Then, A generates a C_0 -semigroup $\mathcal{S}(t)$ fulfilling $||\mathcal{S}(t)|| \leq N$ for some constant N > 0 and

$$\mathcal{S}(t) = \frac{1}{2\pi i} \int_{\Gamma} e^{\varrho t} (\varrho I - A)^{-1} d\varrho,$$

where Γ is a smooth curve contained in \sum_{μ} starting from $\infty e^{-i\phi}$ to $\infty e^{i\phi}$ for some $\pi/2 < \phi < \pi/2 + \mu$, the integral converges in uniform operator topology for t > 0.

2.3 Basic Concepts of Fractional Calculus

In literature, there are more than fifteen definitions of fractional derivative. But the most commonly used definitions are the definitions given by Caputo and Riemaan-

Liouville, as these two definitions coincide with the ordinary derivative at integers.

Definition 2.3.1. For a function $F \in L^1(\mathbb{R}^+, \mathbb{X})$, the Riemann-Liouville (in short RL) fractional integral of order q > 0, is defined as

$$J^{q}F(t) = \frac{1}{\Gamma(q)} \int_{0}^{t} (t-\varrho)^{q-1} F(\varrho) d\varrho, \text{ for } t > 0.$$
 (2.3.1)

We set $J^0 = I$ (the identity operator).

Definition 2.3.2. For a function F, the Riemann-Liouville fractional derivative of order q > 0, is defined as

$${}^{L} \boldsymbol{D}^{q} F(t) := D^{m} J^{m-q} F(t) = \frac{1}{\Gamma(m-q)} \frac{d^{m}}{dt^{m}} \int_{0}^{t} (t-s)^{m-q-1} F(s) ds,$$

where t > 0, $m = \lceil q \rceil = \min\{z \in \mathbb{Z} : z \ge q\}$, $D^m = \frac{d^m}{dt^m}$. We set ${}^L D^0 = I$

Definition 2.3.3. The Caputo fractional derivative of fractional order q > 0 of a function F is defined as

$${}^{C}\boldsymbol{D}^{q}F(t) := J^{m-q}D^{m}F(t) = \frac{1}{\Gamma(m-q)}\int_{0}^{t}(t-s)^{m-q-1}F^{m}(s)ds,$$

for t > 0, $m = \lceil q \rceil$.

Comparing this definition with the Riemann-Liouville one, we see that the definition given by Caputo is more restrictive as it requires the absolute integrability of m^{th} order derivative of the function and therefore functions which are derivable in the Caputo sense are much "fewer" than those which are derivable in the Riemann-Liouville sense.

The main advantage of the Caputo derivative is that the form of initial conditions for FDEs are same as that of integer order differential equations and Caputo fractional derivative of constant function is zero, which does not hold for Riemann-Liouville definition.

2.3.1 Solutions of Caputo Fractional Differential Equations

Let us consider the homogenous problem

$$^{C}\mathbf{D}^{q}y(t) = 0, \ t > 0.$$
 (2.3.2)

Then solution of the above problem is

$$y(t) = d_0 + d_1 t + d_2 t^2 + \dots + d_{n-1} t^{n-1}, \qquad (2.3.3)$$

and

$$J^q \cdot {}^C \mathbf{D}^q y(t) = y(t), \qquad (2.3.4)$$

where $d_i \in \mathbb{R}$, $i = 1, \dots, n-1$ and $n = \lceil q \rceil$. For the nonhomogenous fractional differential equation

$$^{C}\mathbf{D}^{q}y(t) = F(t), \ t \in [0, b], \ b \in \mathbb{R}^{+}.$$
 (2.3.5)

From the equation (2.3.5), we get following integral equation

$$y(t) = d_0 + d_1 t + d_2 t^2 + \dots + d_{n-1} t^{n-1} + \frac{1}{\Gamma(q)} \int_0^t (t-\varrho)^{q-1} F(\varrho) d\varrho.$$
(2.3.6)

Now, we consider the following fractional order problem

$$^{C}\mathbf{D}^{q}y(t) = Ay(t), \ t \in (0,T],$$
(2.3.7)

$$y(0) = y_0, (2.3.8)$$

where 0 < q < 1, y takes its values in X, A is densely defined from $D(A) \subseteq X$ to X, and is a closed linear operator which generates C_0 -semigroup of bounded linear operator $\mathcal{S}(t), t \geq 0$.

The equation (2.3.7) is equivalent to the following

$$y(t) = y_0 + \frac{1}{\Gamma(q)} \int_0^t (t - \varrho)^{q-1} A y(\varrho) d\varrho.$$
 (2.3.9)

The solution to (2.3.7)-(2.3.8) is a function $y \in C([0,T], \mathbb{X})$ that satisfies the following

- (i) $y(t) \in D(A), t \in (0, T],$
- (*ii*) $^{C}\mathbf{D}^{q}y(t)$ exists and continuous on (0,T] with 0 < q < 1,
- (*iii*) y satisfies the equations (2.3.7)-(2.3.8).

Applying the Laplace transformation of the equation (2.3.9) on both sides

$$L[y(t)] = L[y_0] + L[\frac{1}{\Gamma(q)} \int_0^t (t-\varrho)^{q-1} Ay(\varrho) d\varrho]$$

= $L[y_0] + \frac{1}{\lambda^q} AL[y(t)]$
= $\lambda^{q-1} (\lambda^q I - A)^{-1} y_0 = \lambda^{q-1} \int_0^\infty e^{-\lambda^q \varrho} \mathcal{S}(\varrho) y_0 d\varrho.$ (2.3.10)

Let one-sided probability density

$$\Phi_q(\zeta) = \frac{1}{\pi} \sum_{m=1}^{\infty} (-1)^{m-1} \zeta^{-mq-1} \frac{\Gamma(mq+1)}{n!} \sin(m\pi q), \ \zeta \in (0,\infty), \qquad (2.3.11)$$

whose Laplace transform is given by

$$\int_0^\infty e^{-\lambda\zeta} \Phi_q(\zeta) d\zeta = e^{-\lambda^q}, \quad q \in (0,1).$$
(2.3.12)

Therefore, we get

$$\lambda^{q-1} \int_0^\infty e^{-\lambda^q s} \mathcal{S}(s) y_0 ds$$

= $\int_0^\infty q(\lambda t)^{q-1} e^{-(\lambda t)^q} \mathcal{S}(t^q) y_0 dt$, (put $s = t^q$)
= $\int_0^\infty \frac{-1}{\lambda} \frac{d}{dt} [e^{-(\lambda t)^q}] \mathcal{S}(t^q) y_0 dt$
= $\int_0^\infty \int_0^\infty \zeta \Phi_q(\zeta) e^{-(\lambda t\zeta)} \mathcal{S}(t^q) y_0 d\zeta dt$,
= $\int_0^\infty e^{-\lambda t} [\int_0^\infty \Phi_q(\zeta) \mathcal{S}(t^q/\zeta^q) y_0 d\zeta] dt.$ (2.3.13)

From (2.3.10) and (2.3.13), we get

$$L[y(t)] = \int_0^\infty e^{-\lambda t} \Big[\int_0^\infty \Phi_q(\zeta) \mathcal{S}(t^q/\zeta^q) y_0 d\zeta \Big] dt.$$
(2.3.14)

Taking inverse Laplace transformation of above equation

$$y(t) = \int_0^\infty \Phi_q(\zeta) \mathcal{S}(t^q/\zeta^q) y_0 d\zeta$$

=
$$\int_0^\infty \Psi_q(\zeta) \mathcal{S}(t^q\zeta) y_0 d\zeta$$

=
$$\mathcal{S}_q(t) y_0,$$
 (2.3.15)

where $\Psi_q(\zeta) = \frac{1}{q} \zeta^{-1-\frac{1}{q}} \Phi_q(\zeta^{-1/q})$ with $\Psi_q(\zeta) \ge 0$, and $\int_0^\infty \Psi_q(\zeta) d\zeta = 1$. Therefore, the solution of (2.3.7) is

$$y(t) = \mathcal{S}_q(t)y_0, \qquad (2.3.16)$$

where $S_q(t)$ is defined by

$$S_q(t)y = \int_0^\infty \Psi_q(\zeta) S(t^q \zeta) y d\zeta, \quad y \in D(A), \ t \ge 0.$$
(2.3.17)

The Laplace transform of Ψ_q is given by

$$L[\Psi_q(t)] = \int_0^\infty e^{-\lambda t} \Psi_q(t) dt = F_q(\lambda) = \sum_{m=0}^\infty \frac{(-\lambda)^m}{\Gamma(qm+1)} = E_q(-\lambda), \qquad (2.3.18)$$

for 0 < q < 1. Next, we consider the following inhomogeneous FDE

$$^{C}\mathbf{D}^{q}y(t) = Ay(t) + F(t), \quad t \in (0,T], \quad 0 \le T < \infty,$$
(2.3.19)

$$y(0) = y_0, (2.3.20)$$

where $F \in L^1([0,T], \mathbb{X})$.

Taking Laplace transformation of the equation (2.3.19) on both sides and getting

$$L[y(t)] = \lambda^{q-1} (\lambda^{q} I - A)^{-1} y_{0} + (\lambda^{q} I - A)^{-1} L[F(t)]$$

= $\lambda^{q-1} \int_{0}^{\infty} e^{-\lambda^{q} s} \mathcal{S}(s) y_{0} ds + \int_{0}^{\infty} e^{-\lambda^{q} s} \mathcal{S}(s) L[F(s)] ds.$ (2.3.21)

Now, we estimate

$$\begin{split} \int_0^\infty e^{-\lambda^q s} \mathcal{S}(s) L[F(s)] ds \\ &= \int_0^\infty \int_0^\infty q t^{q-1} e^{-(\lambda t)^q} \mathcal{S}(t^q) e^{-\lambda s} F(s) ds dt \\ &= \int_0^\infty \int_0^\infty \int_0^\infty q \Phi_q(\zeta) e^{-(\lambda t\theta)} \mathcal{S}(t^q) e^{-\lambda s} t^{q-1} F(s) d\zeta ds dt \\ &= \int_0^\infty \int_0^\infty \int_0^\infty q \Phi_q(\zeta) e^{-\lambda(t+s)} \mathcal{S}(t^q/\zeta^q) \frac{t^{q-1}}{\zeta^q} F(s) d\zeta ds dt \\ &= \int_0^\infty e^{-\lambda t} \Big[q \int_0^t \int_0^\infty \Phi_q(\zeta) \mathcal{S}(\frac{(t-s)^{q-1}}{\zeta^q}) F(s) \frac{(t-s)^q}{\zeta^q} d\zeta ds \Big] dt, \end{split}$$

Thus, we get

$$\begin{split} L[y(t)] &= \int_0^\infty e^{-\lambda t} \Big[\int_0^\infty \Phi_q(\zeta) \mathcal{S}(t^q/\zeta^q) y_0 d\zeta \Big] dt \\ &+ \int_0^\infty e^{-\lambda t} \Big[q \int_0^t \int_0^\infty \Phi_q(\zeta) \mathcal{S}(\frac{(t-s)^{q-1}}{\zeta^q}) F(s) \frac{(t-s)^q}{\zeta^q} d\zeta ds \Big] dt. \end{split}$$

Taking inverse Laplace transformation of the above equation and obtaining

$$y(t) = \int_0^\infty \Phi_q(\zeta) \mathcal{S}(t^q/\zeta^q) y_0 d\zeta +$$

+ $q \int_0^t \int_0^\infty \Phi_q(\zeta) \mathcal{S}(\frac{(t-s)^{q-1}}{\zeta^q}) F(s) \frac{(t-s)^q}{\zeta^q} d\zeta ds$
= $\int_0^\infty \Psi_q(\zeta) \mathcal{S}(t^q\zeta) y_0 d\zeta + q \int_0^t \int_0^\infty \zeta(t-s)^{q-1} \Psi_q(\zeta) \mathcal{S}((t-s)^{q-1}\zeta) F(s) d\zeta ds$
= $\mathcal{P}(t) y_0 + \int_0^t (t-s)^{q-1} \mathcal{R}(t-s) F(s) ds,$ (2.3.22)

where, the operator $\mathcal{P}(t)$ and $\mathcal{R}(t)$ are defined by

$$\mathcal{P}(t) = \int_0^\infty \Psi_q(\zeta) \mathcal{S}(t^q \zeta) d\zeta, \quad \mathcal{R}(t) = q \int_0^\infty \zeta \Psi_q(\zeta) \mathcal{S}(t^q \zeta) d\zeta.$$
(2.3.23)

Remark 2.3.1. $\Psi_q(\zeta) \ge 0, \zeta \in (0,\infty), \quad \int_0^\infty \Psi_q(\zeta) d\zeta = 1, \quad \int_0^\infty \zeta \Psi_q(\zeta) d\zeta = \frac{1}{\Gamma(1+q)}.$

Definition 2.3.4. A function $y \in C([0,T], \mathbb{X})$ is called a solution to the problem (2.3.19)-(2.3.20) if the following integral equation

$$y(t) = \mathcal{P}(t)y_0 + \int_0^t (t-\varrho)^{q-1} \mathcal{R}(t-\varrho) F(\varrho) d\varrho, \qquad (2.3.24)$$

is verified.

Lemma 2.3.2. [203] The operators $\{\mathcal{R}(t), t \ge 0\}$ and $\{\mathcal{P}(t), t \ge 0\}$ are linear bounded, and

- (i) $\|\mathcal{R}(t)z\| \leq \frac{qM}{\Gamma(1+q)} \|z\|$ and $\|\mathcal{P}(t)z\| \leq M \|z\|, z \in \mathbb{X}.$
- (ii) The families $\{\mathcal{R}(t) : t \ge 0\}$ and $\{\mathcal{P}(t) : t \ge 0\}$ are continuous strongly.
- (iii) If S(t) is compact, then $\mathcal{R}(t)$ and $\mathcal{P}(t)$ are compact for t > 0.
- (iv) If $S(t)(t \ge 0)$ is an equicontinuous, then $\mathcal{R}(t)$ and $\mathcal{P}(t)$ are continuous for t > 0 by the operator norm, which means that for $0 < t' < t'' \le a$

$$\|\mathcal{R}(t'') - \mathcal{R}(t')\| \to 0 \quad and \quad \|\mathcal{P}(t'') - \mathcal{P}(t')\| \to 0 \quad as \quad t'' \to t'.$$

2.4 Basic Concepts of Measure of Noncompactness

Assume that (\mathbb{X}, d) be a complete metric space, denote $\mathcal{B}_{\mathbb{X}}$ for the family of all bounded subsets of \mathbb{X} . Now, we have some notations which will be used in the subsequent chapters. If U is a subset of a metric space (\mathbb{X}, d) , then diam(U) =sup{ $d(y, y') : y, y' \in U$ } is called the diameter of U.

Definition 2.4.1. [178] Suppose that \mathbb{X} is a complete metric space. A function $\mu : \mathcal{B}_{\mathbb{X}} \to [0, \infty)$ is said to be a **measure of noncompactness** on \mathbb{X} , if it satisfies the following :

- (i) $\mu(W) = 0$ iff $W \in \mathcal{B}_{\mathbb{X}}$ is precompact [Regularity];
- (ii) $\mu(W) = \mu(\overline{W})$, where \overline{W} denotes the closure of $W \in \mathcal{B}_{\mathbb{X}}$ [Invariance under closure];
- (*iii*) $\mu(W_1 \cup W_2) = \max\{\mu(W_1), \mu(W_2)\}, \forall W_1, W_2 \in \mathcal{B}_{\mathbb{X}} \ [Semi-additivity].$

Proposition 2.4.1. [178] For bounded sets $W, W_1, W_2 \in \mathcal{B}_{\mathbb{X}}$, a measure of noncompactness function μ fulfills the following conditions

- (i) $\mu(W_1) \leq \mu(W_2)$, when $W_1 \subset W_2$, [Monotonicity];
- (*ii*) $\mu(W_1 \cap W_2) \le \min\{\mu(W_1), \mu(W_2)\};$
- (iii) $\mu(W) = 0$ for each finite set W, [Non-singularity].

Definition 2.4.2. [178] The function $\beta : \mathcal{B}_{\mathbb{X}} \to [0, \infty)$ as following

 $\beta(U) := \inf\{\kappa > 0 : U \subset \bigcup_{k=1}^{n} U_k, \ U_k \subset \mathbb{X}, \ diam(U_k) < \kappa, \ k = 1, 2, \cdots, n \in \mathbb{N}\},\$

is known as Kuratowski measure of noncompactness.

Definition 2.4.3. The Hausdorff measure of noncompactness $\chi : \mathcal{B}_{\mathbb{X}} \to [0,\infty)$ is defined as

$$\chi(U) := \inf\{\kappa > 0 : U \subset \bigcup_{k=1}^{n} B_{r_k}(x_k), \ x_k \in \mathbb{X}, r_k < \kappa, \ k = 1, \cdots, n \in \mathbb{N}\},\$$

where $B_{r_k}(x_k) = \{x \in \mathbb{X} : d(x, x_k) < r_k\}.$

Proposition 2.4.2. Let μ denotes both Hausdorff and Kurtatowski measure of noncompactness. Then for sets $W, W_1, W_2 \in \mathcal{B}_X$, we have

- (i) W is relatively compact iff $\mu(W) = 0$;
- (ii) $\mu(W_1 + W_2) \le \mu(W_1) + \mu(W_2)$ (Algebraic semi-additivity);
- (*iii*) $\mu(W_1) \leq \mu(W_2)$ when $W_1 \subset W_2$;
- (iv) $\mu(\alpha \cdot W) = |\alpha| \cdot \mu(W)$, α is a number (Semi-homogeneity);
- (v) $\mu(W+z) = \mu(W)$ for each $z \in \mathbb{X}$ (Translation invariance);
- (vi) $\mu(W) = \mu(\overline{W}) = \mu(conv(W))$, where \overline{W} and conv(W) denote the closure and convex hull of W respectively.

Lemma 2.4.3. [24] Let X and E be Banach spaces and $\mathcal{F} : D(\mathcal{F}) \subset \mathbb{E} \to X$ satisfies Lipschitz continuity with constant L, then $\mu(\mathcal{F}(V)) \leq L\mu(V)$ for any bounded subset $V \subset D(\mathcal{F})$.

Lemma 2.4.4. [24] For any $W \subset C([a, b], \mathbb{X})$, set $W(t) = \{w(t) : w \in W\}$. If W is equicontinuous and bounded, then $\beta(W(t))$ is continuous on [a, b], moreover $\beta(W) = \sup_{t \in [a, b]} \beta(W(t))$.

Lemma 2.4.5. [94] If $\{w_n\}_{n=1}^{\infty} \subset C([a, b], \mathbb{X})$ is a bounded sequence, then $\beta(\{w_n(t)\}_{n=1}^{\infty}) \in L^1([a, b], \mathbb{X})$ and

$$\beta\left(\left\{\int_0^t w_n(s)ds\right\}_{n=1}^\infty\right) \leqslant 2\int_0^t \beta(\{w_n(s)\}_{n=1}^\infty ds.$$

Lemma 2.4.6. [36] If W is bounded subset of X, then there exists a sequence $\{w_n\}_{n=1}^{\infty} \subset W$, such that $\beta(W) \leq 2\beta(\{w_n\}_{n=1}^{\infty})$.

2.5 Fixed Point Theorems

Theorem 2.5.1. (Banach fixed point theorem) Let F be a contraction mapping from X into X, where X is a complete metric space. Then, X has a unique point zsuch that F(z) = z.

Theorem 2.5.2. (Leray-Schauder nonlinear alternative) [87] Let B is a nonempty, convex and closed subset of a Banach space X, and U is an open subset of B such that $0 \in U$. If $F : \overline{U} \to B$ is a compact and continuous map. Then only one of the following holds :

- (i) F has a fixed point in \overline{U} ,
- (ii) there is a point $u_0 \in \partial U$ (boundary of U) for which $u_0 = \varepsilon F u_0$ holds, for some $0 < \varepsilon < 1$.

Definition 2.5.1. [66] Suppose that S be a nonempty subset of the Banach space \mathbb{X} . A continuous map $F: S \to \mathbb{X}$ is called ρ -set contractive if for some constant $\rho \in [0, 1)$, we have

 $\beta(F(\Omega)) \leq \rho\beta(\Omega)$, for all bounded subsets Ω in S.

Theorem 2.5.3. (ρ -set contraction mapping fixed point theorem) [66] Let Ω be a closed bounded convex subset of a Banach space \mathbb{X} , and the operator $F : \Omega \to \Omega$ is ρ -set contractive, then there exists a fixed point of F in Ω .

Theorem 2.5.4. (Schauder) If Ω is a convex closed bounded subset of a Banach space \mathbb{X} , and $F : \Omega \to \Omega$ is completely continuous. Then F has a fixed point in Ω .

For more details on fixed point theorems, we refer **[12**; **66**; **87**; **88**; **150**], and the references therein.

Chapter 3

Fractional Neutral Integro-Differential Equations involving Nonlocal Integral Boundary Conditions

۵

3.1 Introduction

This chapter concerns with the following fractional integro-differential equations of neutral type having nonlocal flux type integral boundary conditions :

$${}^{C}\mathbf{D}^{q}[x(t) - g(t, x(t))] = f(t, x(t), \int_{0}^{t} \Upsilon(t, \varrho, x(\varrho))d\varrho), \ 0 < t < 1, \ 1 < q \leq 2,$$
$$x'(0) = \alpha \int_{0}^{\xi} x'(\varrho)d\varrho, \ x(1) = \gamma \phi(x'(\eta)),$$
$$0 \leq \xi, \eta \leq 1, \ \xi \neq \frac{1}{\alpha},$$
(3.1.1)

where f and g are real valued functions defined over $[0,1] \times \mathbb{R} \times \mathbb{R}$ and $[0,1] \times \mathbb{R}$ respectively, $\Upsilon : \mathbb{D} \times \mathbb{R} \to \mathbb{R}, \phi : \mathbb{R} \to \mathbb{R}$, and $\mathbb{D} = \{(\tau_1, \tau_2) : 0 \leq \tau_2 < \tau_1 \leq 1\}$. In (3.1.1), the first of nonlocal boundary condition relates the flux at the initial point 0 with the continuous distribution of flux on an interval of arbitrary length ξ , and

The contents of this chapter are published in **Malaya Journal of Matematik** as Arshi Meraj and Dwijendra N Pandey: "Existence result for neutral fractional integro-differential equations with nonlocal integral boundary conditions", Vol. 6, No. 1, 21-27, 2018.

the second condition expresses that the value of unknown function at final point is proportional to a nonlinear function ϕ of flux.

Motivated by the work done in [6; 9; 10; 50] for various type of integral boundary value problems, we considered the above mentioned problem (3.1.1), to investigate the existence results. The remaining part of the chapter is arranged as following: The next section contains a lemma concerning the solution of linear system associated to (3.1.1), and introduce the assumptions which are required to obtain our main results. Section 3.3 consists of the proof of the main results via Leray-Schauder nonlinear alternative and Banach contraction theorem. Finally, in section 3.4, we will present some examples to illustrate our results.

3.2 Preliminaries and Assumptions

Lemma 3.2.1. For \mathbb{R} -valued function f continuous on [0,1], and $g \in C^1([0,1],\mathbb{R})$, solution of the following fractional linear BVP

$${}^{C}\boldsymbol{D}^{q}[x(t) - g(t)] = f(t), \quad t \in (0,1), \ 1 < q \leq 2$$

$$x'(0) = \alpha \int_{0}^{\xi} x'(\varrho) d\varrho, \quad x(1) = \gamma \phi(x'(\eta)), \ 0 \leq \xi, \eta \leq 1, \ (3.2.2)$$

is given by

$$\begin{aligned} x(t) &= \int_{0}^{t} \frac{(t-\varrho)^{q-1}}{\Gamma(q)} f(\varrho) d\varrho + g(t) - g(1) - \int_{0}^{1} \frac{(1-\varrho)^{q-1}}{\Gamma(q)} f(\varrho) d\varrho \\ &+ \frac{t-1}{1-\alpha\xi} \bigg[\alpha(g(\xi) - g(0)) - g'(0) + \alpha \int_{0}^{\xi} \int_{0}^{\varrho} \frac{(\varrho-\tau)^{q-2}}{\Gamma(q-1)} f(\tau) d\tau d\varrho \bigg] \\ &+ \gamma \phi \bigg(\frac{\alpha}{(1-\alpha\xi)} (g(\xi) - g(0)) - \frac{1}{(1-\alpha\xi)} g'(0) + g'(\eta) \\ &+ \frac{\alpha}{1-\alpha\xi} \int_{0}^{\xi} \int_{0}^{\varrho} \frac{(\varrho-\tau)^{q-2}}{\Gamma(q-1)} f(\tau) d\tau d\varrho + \int_{0}^{\eta} \frac{(\eta-\varrho)^{q-2}}{\Gamma(q-1)} f(\varrho) d\varrho \bigg). (3.2.3) \end{aligned}$$

Proof. From [158], it is known that the solution of (3.2.1) is

$$x(t) = c_1 + c_2 t + g(t) - g(0) + \int_0^t \frac{(t-\varrho)^{q-1}}{\Gamma(q)} f(\varrho) d\varrho.$$
(3.2.4)

On applying the conditions (3.2.2), we find that

$$c_{2} = \frac{1}{(1 - \alpha\xi)} \left(\alpha(g(\xi) - g(0)) - g'(0) + \alpha \int_{0}^{\xi} \int_{0}^{\varrho} \frac{(\varrho - \tau)^{q-2}}{\Gamma(q-1)} f(\tau) d\tau d\varrho \right) \quad (3.2.5)$$

and

$$c_1 = \gamma \phi \left(c_2 + g'(\eta) + \int_0^\eta \frac{(\eta - \varrho)^{q-2}}{\Gamma(q-1)} f(\varrho) d\varrho \right) + g(0) - g(1) - c_2$$
$$- \int_0^1 \frac{(1 - \varrho)^{q-1}}{\Gamma(q)} f(\varrho) d\varrho,$$

where c_2 is given by (3.2.5). Substituting the values of c_1 and c_2 in (3.2.4), we get (3.2.3).

Let us denote $C := C([0, 1], \mathbb{R})$ equipped with the norm $||x|| = \sup_{t \in [0, 1]} |x(t)|$, and $\mathcal{P} := C^1([0, 1], \mathbb{R})$ equipped with norm $||v||_{C^1} = \sup_{t \in [0, 1]} \{|v(t)|, |v'(t)|\}.$

By the help of Lemma 3.2.1, we define $F: \mathcal{P} \to \mathcal{P}$ as

$$(Fv)(t) = \int_{0}^{t} \frac{(t-\varrho)^{q-1}}{\Gamma(q)} f(\varrho, v(\varrho), Kv(\varrho)) d\varrho + g(t, v(t)) - g(1, v(1)) - \int_{0}^{1} \frac{(1-\varrho)^{q-1}}{\Gamma(q)} f(\varrho, v(\varrho), Kv(\varrho)) d\varrho + \frac{t-1}{1-\alpha\xi} \bigg[\alpha(g(\xi, v(\xi)) - g(0, v(0))) - g'(0, v(0)) + \alpha \int_{0}^{\xi} \int_{0}^{\varrho} \frac{(\varrho-\tau)^{q-2}}{\Gamma(q-1)} f(\tau, v(\tau), Kv(\tau)) d\tau d\varrho \bigg] + \gamma \phi \bigg(\frac{\alpha}{(1-\alpha\xi)} (g(\xi, v(\xi)) - g(0, v(0))) - \frac{1}{(1-\alpha\xi)} g'(0, v(0)) + g'(\eta, v(\eta)) + \int_{0}^{\eta} \frac{(\eta-\varrho)^{q-2}}{\Gamma(q-1)} f(\varrho, v(\varrho), Kv(\varrho)) d\varrho + \frac{\alpha}{1-\alpha\xi} \int_{0}^{\xi} \int_{0}^{\varrho} \frac{(\varrho-\tau)^{q-2}}{\Gamma(q-1)} f(\tau, v(\tau), Kv(\tau)) d\tau d\varrho \bigg),$$
(3.2.6)

where $Kv(t) := \int_0^t \Upsilon(t, \varrho, v(\varrho)) d\varrho$.

Now, we state the required assumptions :

- (A1) There exist continuous nondecreasing functions $\psi_1, \psi_2, \psi_3 : \mathbb{R}^+ \to \mathbb{R}^+$ and continuous positive functions p_1, p_2, p_3 defined on [0, 1] satisfying :
- (i) $|f(\tau, y, z)| \leq p_1(\tau)\psi_1(||y||), \quad \forall (\tau, y, z) \in [0, 1] \times \mathbb{R} \times \mathbb{R},$
- (ii) $|g(\tau, z)| \leq p_2(\tau)\psi_2(||z||), |g'(\tau, z)| \leq p_3(\tau)\psi_3(||z||), \quad \forall (\tau, z) \in [0, 1] \times \mathbb{R}.$
- (A2) The function ϕ is continuous, and $|\phi(v)| \leq |v|, \quad \forall v \in \mathbb{R}$.

- (A3) There is a positive constant \mathfrak{M} satisfying $\frac{\mathfrak{M}}{\|p\|\psi(\mathfrak{M})\Lambda_1} > 1$, where $\|p\| = \max\{\|p_i\| : i = 1, 2, 3\}$, $\Lambda_1 = \frac{1}{\Gamma(q+1)} \left[2 + |\gamma|q\eta^{q-1} + (1+|\gamma|)\frac{|\alpha|\xi^q}{|1-\alpha\xi|} \right] + 2 + |\gamma| + \frac{1+|\gamma|}{|1-\alpha\xi|} + 2(1+|\gamma|)\frac{|\alpha|}{|1-\alpha\xi|}$ and $\psi(r) = \max\{\psi_1(r), \psi_2(r), \psi_3(r)\}$.
- (B1) The function f is continuous and satisfying :

$$|f(\tau, v_1, y_1) - f(\tau, v_2, y_2)| \leqslant L_1\{|v_1 - v_2| + |y_1 - y_2|\}, \quad \tau \in [0, 1], v_1, v_2, y_1, y_2 \in \mathbb{R},$$

for some positive constant L_1 .

(B2) The function g is continuously differentiable and satisfying :

$$|g(\tau, v_1) - g(\tau, v_2)| \leq L_2 |v_1 - v_2|, |g'(\tau, v_1) - g'(\tau, v_2)| \leq L_3 |v_1 - v_2|,$$

$$\tau \in [0, 1], v_1, v_2 \in \mathbb{R},$$

for some positive constants L_2, L_3 .

(B3) The function Υ is continuous and the following holds :

$$|\Upsilon(\tau, \upsilon, y_1) - \Upsilon(\tau, \upsilon, y_2)| \leq L_4 |y_1 - y_2|, \quad \tau, \upsilon \in [0, 1], y_1, y_2 \in \mathbb{R},$$

for some positive constant L_4 .

(B4) Let $\rho = \max\{\rho_1, \rho_2\} < 1$ where

$$\rho_{1} = \left(2 + (1 + |\gamma|) \frac{|\alpha|\xi^{q}}{|1 - \alpha\xi|} + |\gamma|q\eta^{q-1}\right) \frac{L_{1}(1 + L_{4})}{\Gamma(q+1)} \\ + \left(2 + 2(1 + |\gamma|) \frac{|\alpha|}{|1 - \alpha\xi|}\right) L_{2} + \left(|\gamma| + (1 + |\gamma|) \frac{1}{|1 - \alpha\xi|}\right) L_{3},$$

and

$$\rho_2 = \left(q + \frac{|\alpha|\xi^q}{|1 - \alpha\xi|}\right) \frac{L_1(1 + L_4)}{\Gamma(q + 1)} + 2\frac{|\alpha|}{|1 - \alpha\xi|} L_2 + \left(1 + \frac{1}{|1 - \alpha\xi|}\right) L_3.$$

3.3 Existence and Uniqueness Results

The following theorem is proved via Leray-Schauder nonlinear alternative.

Theorem 3.3.1. Suppose that the function f, Υ are continuous and g is continuously differentiable. Assume that the assumptions (A1)-(A3) hold. Then, there exists a solution to the boundary value problem (3.1.1).

Proof. Observe that the operator F defined in (3.2.6) is continuous. Now, we check that the image of a bounded set in \mathcal{P} under F is a bounded set in \mathcal{P} . Let $B_r = \{v \in \mathcal{P} : ||v||_{C^1} \leq r\}$ (r > 0), then for $v \in B_r$,

$$\begin{split} |(Fv)(t)| &\leqslant \int_{0}^{t} \frac{(t-\varrho)^{q-1}}{\Gamma(q)} |f(\varrho, v(\varrho), Kv(\varrho))| d\varrho + |g(t, v(t))| + |g(1, v(1))| \\ &+ \int_{0}^{1} \frac{(1-\varrho)^{q-1}}{\Gamma(q)} |f(\varrho, v(\varrho), Kv(\varrho))| d\varrho \\ &+ \frac{|t-1|}{|1-\alpha\xi|} \Big[|\alpha| (|g(\xi, v(\xi))| + |g(0, v(0))|) + |g'(0, v(0))| \\ &+ |\alpha| \int_{0}^{\xi} \int_{0}^{\varrho} \frac{(\varrho-\tau)^{q-2}}{\Gamma(q-1)} |f(\tau, v(\tau), Kv(\tau))| d\tau d\varrho \Big] \\ &+ |\gamma| \Big(\frac{|\alpha|}{|1-\alpha\xi|} (|g(\xi, v(\xi))| + |g(0, v(0))|) \\ &+ \frac{1}{|1-\alpha\xi|} |g'(0, v(0))| + |g'(\eta, v(\eta))| \\ &+ \int_{0}^{\eta} \frac{(\eta-\varrho)^{q-2}}{\Gamma(q-1)} |f(\varrho, v(\varrho), Kv(\varrho))| d\varrho \\ &+ \frac{|\alpha|}{|1-\alpha\xi|} \int_{0}^{\xi} \int_{0}^{\varrho} \frac{(\varrho-\tau)^{q-2}}{\Gamma(q-1)} |f(\tau, v(\tau), Kv(\tau))| d\tau d\varrho \Big) \\ &\leqslant 2 ||p_1||\psi_1(r)\frac{1}{\Gamma(q+1)} + 2 ||p_2||\psi_2(r) \\ &+ \frac{1}{|1-\alpha\xi|} \Big(2|\alpha|||p_2||\psi_2(r) + ||p_3||\psi_3(r) + |\alpha|||p_1||\psi_1(r)\frac{\xi^q}{\Gamma(q+1)} \Big) \\ &+ |\gamma| \Big(2 \frac{|\alpha|}{|1-\alpha\xi|} ||p_2||\psi_2(r) + \frac{1}{|1-\alpha\xi|} ||p_3||\psi_3(r) + ||p_3||\psi_3(r) \\ &+ ||p_1||\psi_1(r)\frac{\eta^{q-1}}{\Gamma(q)} + ||p_1||\psi_1(r)\frac{|\alpha|\xi^q}{(|1-\alpha\xi|)\Gamma(q+1)} \Big). \end{split}$$

Choosing $\psi(r) = \max\{\psi_1(r), \psi_2(r), \psi_3(r)\}$, we get

$$|(Fv)(t)| \leqslant ||p||\psi(r)\Lambda_1.$$
(3.3.1)

On differentiating equation (3.2.6) with respect to t, we get

$$(Fv)'(t) = \int_{0}^{t} \frac{(t-\varrho)^{q-2}}{\Gamma(q-1)} f(\varrho, v(\varrho), Kv(\varrho)) d\varrho +g'(t, v(t)) + \frac{1}{1-\alpha\xi} \bigg[\alpha(g(\xi, v(\xi)) - g(0, v(0))) - g'(0, v(0)) +\alpha \int_{0}^{\xi} \int_{0}^{\varrho} \frac{(\varrho-\tau)^{q-2}}{\Gamma(q-1)} f(\tau, v(\tau), Kv(\tau)) d\tau d\varrho \bigg],$$
(3.3.2)

by (3.3.2), for each $v \in B_r$ we have

$$\begin{aligned} |(Fv)'(t)| &\leqslant \|p_1\|\psi_1(r)\frac{1}{\Gamma(q)} + \|p_3\|\psi_3(r) + \frac{1}{|1-\alpha\xi|} \left(2|\alpha|\|p_2\|\psi_2(r) + \|p_3\|\psi_3(r) + |\alpha|\|p_1\|\psi_1(r)\frac{\xi^q}{\Gamma(q+1)}\right) \\ &\quad + |\alpha|\|p_1\|\psi_1(r)\frac{\xi^q}{\Gamma(q+1)}\right) \\ &\leqslant \|p\|\psi(r)\left(1 + \frac{1}{\Gamma(q)} + \frac{1+2|\alpha|}{|1-\alpha\xi|} + \frac{|\alpha|\xi^q}{(|1-\alpha\xi|)\Gamma(q+1)}\right), \end{aligned}$$

by denoting $\Lambda_2 = \left(1 + \frac{1}{\Gamma(q)} + \frac{1+2|\alpha|}{|1-\alpha\xi|} + \frac{|\alpha|\xi^q}{(|1-\alpha\xi|)\Gamma(q+1)}\right)$, we have

$$|(Fv)'(t)| \leq ||p||\psi(r)\Lambda_2,$$
 (3.3.3)

observe that $\Lambda_2 \leq \Lambda_1$. Thus by (3.3.1) and (3.3.3), we have

$$||Fv||_{C^1} \leq ||p||\psi(r)\Lambda_1.$$
 (3.3.4)

Next, we check that bounded sets are taken into equicontinuous sets of \mathcal{P} by F. For

 $t_1 < t_2$ in [0,1] and $v \in B_r$

$$\begin{split} |(Fv)(t_{2}) - (Fv)(t_{1})| &\leq \left| \int_{0}^{t_{2}} \frac{(t_{2} - \varrho)^{q-1}}{\Gamma(q)} f(\varrho, v(\varrho), Kv(\varrho)) d\varrho \right| \\ &- \int_{0}^{t_{1}} \frac{(t_{1} - \varrho)^{q-1}}{\Gamma(q)} f(\varrho, v(\varrho), Kv(\varrho)) d\varrho \right| \\ &+ |g(t_{2}, v(t_{2})) - g(t_{1}, v(t_{1}))| \\ &+ \frac{|t_{2} - t_{1}|}{|1 - \alpha\xi|} |\alpha(g(\xi, v(\xi)) - g(0, v(0))) - g'(0, v(0)) \\ &+ \alpha \int_{0}^{\xi} \int_{0}^{\varrho} \frac{(\varrho - \tau)^{q-2}}{\Gamma(q-1)} f(\tau, v(\tau), Kv(\tau)) d\tau d\varrho | \\ &\leqslant \frac{1}{\Gamma(q)} \int_{0}^{t_{1}} [(t_{2} - \varrho)^{q-1} - (t_{1} - \varrho)^{q-1}] |f(\varrho, v(\varrho), Kv(\varrho))| d\varrho \\ &+ \frac{1}{\Gamma(q)} \int_{t_{1}}^{t_{2}} (t_{2} - \varrho)^{q-1} |f(\varrho, v(\varrho), Kv(\varrho))| d\varrho \\ &+ |g(t_{2}, v(t_{2})) - g(t_{1}, v(t_{1}))| \\ &+ |\frac{|t_{2} - t_{1}|}{|1 - \alpha\xi|} |\alpha(g(\xi, v(\xi)) - g(0, v(0))) - g'(0, v(0)) \\ &+ \alpha \int_{0}^{\xi} \int_{0}^{\varrho} \frac{(\varrho - \tau)^{q-2}}{\Gamma(q-1)} f(\tau, v(\tau), Kv(\tau)) d\tau d\varrho \Big|. \end{split}$$
(3.3.5)

Therefore, $|(Fv)(t_2) - (Fv)(t_1)| \to 0$ independently of $v \in B_r$ as $t_2 \to t_1$, similarly for the derivative term

$$\begin{aligned} |(Fv)'(t_{2}) - (Fv)'(t_{1})| &\leq \left| \int_{0}^{t_{2}} \frac{(t_{2} - \varrho)^{q-2}}{\Gamma(q-1)} f(\varrho, v(\varrho), Kv(\varrho)) d\varrho \right| \\ &- \int_{0}^{t_{1}} \frac{(t_{1} - \varrho)^{q-2}}{\Gamma(q-1)} f(\varrho, v(\varrho), Kv(\varrho)) d\varrho \right| \\ &+ |g'(t_{2}, v(t_{2})) - g'(t_{1}, v(t_{1}))| \\ &\leqslant \frac{1}{\Gamma(q-1)} \int_{0}^{t_{1}} [(t_{2} - \varrho)^{q-2} - (t_{1} - \varrho)^{q-2}] \\ &|f(\varrho, v(\varrho), Kv(\varrho))| d\varrho \\ &+ \frac{1}{\Gamma(q-1)} \int_{t_{1}}^{t_{2}} (t_{2} - \varrho)^{q-2} |f(\varrho, v(\varrho), Kv(\varrho))| d\varrho \\ &+ |g'(t_{2}, v(t_{2})) - g'(t_{1}, v(t_{1}))|. \end{aligned}$$
(3.3.6)

So, $|(Fv)'(t_2) - (Fv)'(t_1)| \to 0$ independently of $v \in B_r$ as $t_2 \to t_1$, therefore by (3.3.5), (3.3.6) and Arzela-Ascoli's theorem it follows that F is completely continuous.

Now, let $v = \lambda F v$ where $\lambda \in (0, 1)$, then $\|v\|_{C^1} < \|Fv\|_{C^1}$. Using (3.3.4), we have $\|v\|_{C_1} \leq \|p\|\psi(\|v\|_{C_1})\Lambda_1$. Consequently, we have $\frac{\|v\|_{C_1}}{\|p\|\psi(\|v\|_{C_1})\Lambda_1} \leq 1$. By assumption

(A3), we have positive constant \mathfrak{M} with $||v||_{C_1} \neq \mathfrak{M}$. Suppose that $S = \{v \in \mathcal{P} : ||v||_{C^1} < \mathfrak{M}\}$. Observe that $F : \overline{S} \to \mathcal{P}$ is continuous and compact. Moreover, there is no $v \in \partial S$, satisfying $v = \lambda F v$ for some $\lambda \in (0, 1)$. Thus, Leray-Schauder nonlinear alternative theorem implies that F has a fixed point $v \in \overline{S}$, hence a solution to the problem (3.1.1).

Theorem 3.3.2. Assume that the hypotheses (A2), (B1)-(B4) hold. Then, the system (3.1.1) has a unique solution.

Proof. For $t \in [0, 1]$ and $v, w \in \mathcal{P}$, we estimate

$$\begin{split} |(Fv)(t) - (Fw)(t)| &\leqslant \int_{0}^{t} \frac{(t-\varrho)^{q-1}}{\Gamma(q)} |f(\varrho, v(\varrho), Kv(\varrho)) - f(\varrho, w(\varrho), Kw(\varrho))|d\varrho \\ &+ |g(t, v(t)) - g(t, w(t))| + |g(1, v(1)) - g(1, w(1))| \\ &+ \int_{0}^{1} \frac{(1-\varrho)^{q-1}}{\Gamma(q)} |f(\varrho, v(\varrho), Kv(\varrho)) - f(\varrho, w(\varrho), Kw(\varrho))|d\varrho \\ &+ \frac{1}{|1-\alpha\xi|} \bigg[|\alpha| (|g(\xi, v(\xi)) - g(\xi, w(\xi))| \\ &+ |\alpha| (|g(0, v(0)) - g(0, w(0))|) + |g'(0, v(0)) - g'(0, w(0))| \\ &+ |\alpha| \int_{0}^{\xi} \int_{0}^{\varrho} \frac{(\varrho - \tau)^{q-2}}{\Gamma(q-1)} |f(\tau, v(\tau), Kv(\tau)) \\ &- f(\tau, w(\tau), Kw(\tau))|d\tau d\varrho \bigg] \\ &+ |\gamma| \bigg(\frac{|\alpha|}{|1-\alpha\xi|} (|g(\xi, v(\xi)) - g(\xi, w(\xi))| \\ &+ |g(0, v(0)) - g(0, w(0))|) + \frac{1}{|1-\alpha\xi|} |g'(0, v(0)) \\ &- g'(0, w(0))| + |g'(\eta, v(\eta)) - g'(\eta, w(\eta))| \\ &+ \int_{0}^{\eta} \frac{(\eta - \varrho)^{q-2}}{\Gamma(q-1)} |f(\varrho, v(\varrho), Kv(\varrho)) \\ &- f(\varrho, w(\varrho), Kw(\varrho))|d\varrho \\ &+ \frac{|\alpha|}{|1-\alpha\xi|} \int_{0}^{\xi} \int_{0}^{\varrho} \frac{(\varrho - \tau)^{q-2}}{\Gamma(q-1)} |f(\tau, v(\tau), Kv(\tau)) \\ &- f(\tau, w(\tau), Kw(\tau))|d\tau d\varrho \bigg) \end{split}$$

$$\leq \left[\frac{2}{\Gamma(q+1)} L_{1}(1+L_{4}) + 2L_{2} + \frac{1}{|1-\alpha\xi|} \left(2|\alpha|L_{2}+L_{3}+|\alpha|L_{1}(1+L_{4})\frac{\xi^{q}}{\Gamma(q+1)} \right) + |\gamma| \left(2\frac{|\alpha|}{|1-\alpha\xi|} L_{2} + \frac{1}{|1-\alpha\xi|} L_{3} + L_{3} + L_{1}(1+L_{4})\frac{\eta^{q-1}}{\Gamma(q} + L_{1}(1+L_{4})\frac{|\alpha|\xi^{q}}{(|1-\alpha\xi|)\Gamma(q+1)} \right) \\ = \left] \|v-w\| \\ \leq \rho_{1} \|v-w\|.$$

$$(3.3.7)$$

Similarly, for the derivative term we have

$$\begin{split} |(Fv)'(t) - (Fw)'(t)| &\leq \int_{0}^{t} \frac{(t-\varrho)^{q-2}}{\Gamma(q-1)} |f(\varrho, v(\varrho), Kv(\varrho)) - f(\varrho, w(\varrho), Kw(\varrho))| d\varrho \\ &+ |g'(t, v(t)) - g'(t, w(t))| + \frac{1}{|1-\alpha\xi|} \left[|\alpha| |g(\xi, v(\xi)) - g(\xi, w(\xi))| + |\alpha| |g(0, v(0)) - g(0, w(0))| + |g'(0, v(0)) - g((0, w(0)))| + |\alpha| \int_{0}^{\xi} \int_{0}^{\varrho} \frac{(\varrho - \tau)^{q-2}}{\Gamma(q-1)} |f(\tau, v(\tau), Kv(\tau))| - f(\tau, w(\tau), Kw(\tau))| d\tau d\varrho \right] \\ &\leq \left[\frac{1}{\Gamma(q)} L_{1}(1 + L_{4}) + L_{3} + \frac{1}{|1-\alpha\xi|} \left(2|\alpha|L_{2} + L_{3} + |\alpha|L_{1}(1 + L_{4}) \frac{\xi^{q}}{\Gamma(q+1)} \right) \right] \|v - w\| \\ &\leq \rho_{2} \|v - w\|. \end{split}$$
(3.3.8)

By (3.3.7) and (3.3.8), we estimate

$$||Fv - Fw||_{C^1} < \rho ||v - w||_{C^1}, \quad v, w \in \mathcal{P},$$

since $\rho < 1$ by assumption (B4), consequently F is a contraction. By Banach contraction fixed point theorem, we conclude that there exists a unique solution to the problem (3.1.1).

3.4 Examples

Example(1): Consider the following fractional BVP

$$\begin{cases} {}^{C}\mathbf{D}^{\frac{3}{2}}[x(t) - \frac{e^{-t}}{35(1+11e^{t})}x(t)] = \frac{1}{(t+4)^{2}}|x(t)| + \frac{1}{16}\int_{0}^{t}\frac{e^{-\varrho}}{9}\frac{|x(t)|}{1+|x(t)|}d\varrho, \quad t \in (0,1); \\ x'(0) = \frac{1}{2}\int_{0}^{\frac{1}{3}}x'(\varrho)d\varrho, \quad x(1) = \frac{1}{3}\phi(x'(\frac{3}{4})). \end{cases}$$
(3.4.1)

Here, $q = \frac{3}{2}$, $\alpha = \frac{1}{2}$, $\xi = \frac{1}{3}$, $\gamma = \frac{1}{3}$, $\eta = \frac{3}{4}$ and

$$\phi(v) = \begin{cases} \sqrt{|v|}, & |v| \ge 1; \\ v^2, & |v| < 1. \end{cases}$$
(3.4.2)

With the given values, we find that $\Lambda_1 \approx 9.45$. Clearly,

$$f(t, x, Kx) = \frac{1}{(t+4)^2} |x(t)| + \frac{1}{16} Kx(t),$$

where $Kx(t) = \int_0^t \frac{e^{-s}}{9} \frac{|x(t)|}{1+|x(t)|} ds$ and $g(t,x) = \frac{e^{-t}}{35(1+11e^t)} x(t)$.

$$\begin{aligned} f(t,x,Kx)| &\leqslant \frac{1}{(t+4)^2} |x(t)| + \frac{1}{16} \int_0^t \frac{1}{9} e^{-s} ds \\ &\leqslant \frac{1}{16} (\|x\|_{C^1} + \frac{1}{9}), \\ |g(t,x)| &\leqslant \frac{e^{-t}}{35(1+11e^t)} \|x\|_{C^1}, \\ g'(t,x) &= \frac{1}{35} \left[\frac{e^{-t}(x'-x) - 22x + 11x'}{(1+11e^t)^2} \right], \end{aligned}$$

therefore, we have

$$|g'(t,x)| \leq \frac{1}{(1+11e^t)^2} ||x||_{C^1}.$$

Hence, $p_1(t) = \frac{1}{16}$, $p_2(t) = \frac{e^{-t}}{35(1+11e^t)}$, $p_3(t) = \frac{1}{(1+11e^t)^2}$, $\psi_1(r) = r + \frac{1}{9}$, $\psi_2(r) = r$, $\psi_3(r) = r$, and $||p|| = \frac{1}{16}$, $\psi(r) = r + \frac{1}{9}$. Now using the condition in (A3) that is $\frac{\mathfrak{M}}{||p||\psi(\mathfrak{M})\Lambda_1} > 1$, we find that $\mathfrak{M} > 0.1603$. Hence, Theorem 3.3.1 tells that the BVP (3.4.1) has a solution.

Example(2): Consider the following fractional BVP

$$\begin{cases} {}^{C}\mathbf{D}^{\frac{3}{2}}[x(t) - \frac{1}{9}e^{-t}x(t)] = \frac{1}{(t+6)^{2}}\frac{|x(t)|}{1+|x(t)|} + \frac{1}{36}\int_{0}^{t}e^{\frac{-1}{5}x(s)}ds, \quad t \in (0,1); \\ x'(0) = \frac{1}{2}\int_{0}^{\frac{1}{3}}x'(s)ds, \quad x(1) = \frac{1}{3}\phi(x'(\frac{3}{4})). \end{cases}$$
(3.4.3)

Here, $q = \frac{3}{2}$, $\alpha = \frac{1}{2}$, $\xi = \frac{1}{3}$, $\gamma = \frac{1}{3}$, $\eta = \frac{3}{4}$ and ϕ is given as (3.4.2). Clearly, $f(t, x, Kx) = \frac{1}{(t+6)^2} \frac{|x(t)|}{1+|x(t)|} + \frac{1}{36} Kx(t)$, where $Kx(t) = \int_0^t e^{\frac{-1}{5}x(s)} ds$ and $g(t, x) = \frac{1}{9}e^{-t}x(t)$.

$$\begin{aligned} |\Upsilon(t,s,v(s)) - \Upsilon(t,s,w(s))| &= |e^{\frac{-1}{5}v(s)} - e^{\frac{-1}{5}w(s)}| \\ &\leqslant \frac{1}{5} ||v - w||_{C^1}, \end{aligned}$$

$$\begin{split} |f(t,v,Kv) - f(t,w,Kw)| &\leqslant \frac{1}{(t+6)^2} \frac{\|v-w\|}{(1+\|v\|)(1+\|w\|)} + \frac{1}{36} \|Kv - Kw\| \\ &\leqslant \frac{1}{36} \Big[\|v-w\|_{C^1} + \|Kv - Kw\|_{C^1} \Big], \\ |g(t,v) - g(t,w)| &\leqslant \frac{1}{9} \|v-w\|_{C^1}, \\ g'(t,v) &= \frac{1}{9} e^{-t} (v'-v), \\ |g'(t,v) - g'(t,w)| &\leqslant \frac{1}{9} \Big[\|v'-w'\| + \|v-w\| \Big] \\ &\leqslant \frac{2}{9} \|v-w\|_{C^1}. \end{split}$$

Therefore $L_1 = \frac{1}{36}$, $L_2 = \frac{1}{9}$, $L_3 = \frac{2}{9}$, $L_4 = \frac{1}{5}$, and we get $\rho_1 = 0.893 < 1$ and $\rho_2 = 0.662 < 1$. Hence, by Theorem 3.3.2, we have a unique solution to the BVP (3.4.3).

Chapter 4

Fractional Non-Instantaneous Impulsive Integro-Differential Equations with Nonlocal Conditions

۵

4.1 Introduction

In [95], Hernández et al. introduced first order non-instantaneous impulsive differential equations and the concepts of solutions for it. They established the existence of solutions by the help of compact semigroup and fixed point theorems. Further, Kumar et al. [117] generalized the problem of [95] for fractional order non-instantaneous impulsive evolution equations. Chen et al. [57] established the existence of mild solutions for first order non-instantaneous impulsive differential equations by using noncompact semigroup, Kuratowski measure of noncompactness and ρ -set contraction mapping fixed point theorem.

Therefore inspired by the above works, we considered the following abstract integro-differential systems of fractional order with non-instantaneous impulses and

The contents of this chapter are accepted in **Arab Journal of Mathematical Sciences** as Arshi Meraj, Dwijendra N Pandey: "Existence of mild solutions for fractional non-instantaneous impulsive integro-differential equations with nonlocal conditions", DOI: 10.1016/j.ajmsc.2018.11.002.

nonlocal conditions in a Banach space $\mathbb X$:

$${}^{C}\mathbf{D}^{q}x(t) + Ax(t) = f\left(t, x(t), \int_{0}^{t} \Re(t, s)x(s)ds\right), \quad t \in \bigcup_{k=0}^{m} (s_{k}, t_{k+1}],$$

$$x(t) = \gamma_{k}(t, x(t)), \quad t \in \bigcup_{k=1}^{m} (t_{k}, s_{k}],$$

$$x(0) + g(x) = x_{0},$$
(4.1.1)

where 0 < q < 1, A is closed linear operator defined on $D(A) \subset \mathbb{X}$, -A generates an equicontinuous and uniformly bounded C_0 semigroup $\mathcal{S}(t)(t \ge 0)$ on \mathbb{X} , J = [0, a], $0 < t_1 < t_2 < \ldots < t_m < t_{m+1} := a, s_0 := 0$ and $s_k \in (t_k, t_{k+1})$ for each k = $1, 2, \ldots, m$, f and g are \mathbb{X} -valued functions defined over $J \times \mathbb{X} \times \mathbb{X}$ and $\mathcal{PC}(J, \mathbb{X})$ respectively, $\gamma_k : (t_k, s_k] \times \mathbb{X} \to \mathbb{X}$ are non-instantaneous impulsive functions for all $k = 1, 2, \ldots, m$, \mathfrak{K} belongs to $C(\mathbb{D}, \mathbb{R}^+)$ with $\mathbb{D} := \{(\tau, \upsilon) : 0 \leq \upsilon < \tau \leq a\}$ and $x_0 \in \mathbb{X}$.

We conclude this section by summarizing the contents of this chapter. Section 4.2 consists of some basic notations, definitions, preliminary lemmas and assumptions. Section 4.3 contains the proof of main result without assuming that semigroup is noncompact. Section 4.4 consists of an example.

4.2 Preliminaries and Assumptions

Suppose that X be a Banach space and $\mathcal{PC}(J, \mathbb{X}) = \{v : J \to \mathbb{X} : v \text{ is continuous at } t \neq t_k, v(t_{k-}) = v(t_k) \text{ and } v(t_{k+}) \text{ exists for all } k = 1, 2, \dots, m\}$, which is a Banach space, we use θ to denote the zero function in $\mathcal{PC}(J, \mathbb{X})$. For each finite constant r > 0, let $B_r = \{v \in \mathcal{PC}(J, \mathbb{X}) : ||v|| \leq r\}$. Denote $\mathfrak{F}v(t) := \int_0^t \mathfrak{K}(t, \varrho)v(\varrho)d\varrho$, and let $G^* = \sup_{t \in J} \int_0^t \mathfrak{K}(t, \varrho)d\varrho < \infty$. Let $M = \sup_{t \in J} ||\mathcal{S}(t)||_{\mathbb{B}(\mathbb{X})}$, note that $M \ge 1$. A C_0 semigroup $\mathcal{S}(t)(t \ge 0)$ is called equicontinuous if for every t > 0, the operator $\mathcal{S}(t)$ is continuous by the operator norm.

Definition 4.2.1. ([117]) A function $v \in \mathcal{PC}(J, \mathbb{X})$ is known as a mild solution to

the system (4.1.1) if $v(0) = x_0 - g(v)$, $v(t) = \gamma_k(t, v(t))$ for all $t \in \bigcup_{k=1}^m (t_k, s_k]$, and

$$v(t) = \begin{cases} \mathcal{P}(t)(x_0 - g(v)) + \int_0^t (t - \varrho)^{q-1} \mathcal{R}(t - \varrho) f(\varrho, v(\varrho), \mathfrak{F}v(\varrho)) d\varrho, & t \in (0, t_1]; \\ \mathcal{P}(t - s_k) \gamma_k(s_k, v(s_k)) + \int_{s_k}^t (t - \varrho)^{q-1} \mathcal{R}(t - \varrho) f(\varrho, v(\varrho), \mathfrak{F}v(\varrho)) d\varrho, \\ & t \in (s_k, t_{k+1}], \ k = 1, 2, \dots, m. \end{cases}$$

Let us introduce the assumptions which are needed to prove our main result :

- (H1) The function $f(t, \cdot, \cdot)$ is continuous on $\mathbb{X} \times \mathbb{X}$ for each $t \in J$, and $f(\cdot, z, v)$ is Lebesgue measurable on J for each $(z, v) \in \mathbb{X} \times \mathbb{X}$.
- (H2) For some nondecreasing continuous function $\psi : [0, \infty) \to (0, \infty)$, and function $\phi \in L^{\frac{1}{q_1}}(J, \mathbb{R}^+)$ where $q_1 \in (0, q)$, the following holds

$$\|f(t, z, y)\| \leq \phi(t)\psi(\|z\|), \quad \forall z, y \in \mathbb{X}; t \in J.$$

(H3) g is continuous, and

$$\|g(z) - g(v)\| \leqslant \alpha^* \|z - v\|, \quad \forall z, v \in \mathcal{PC}(J, \mathbb{X}),$$

for some constant $\alpha^* > 0$.

(H4) $\gamma_k : [t_k, s_k] \times \mathbb{X} \to \mathbb{X}$ are continuous and there exist constants $K_{\gamma_k} > 0$, $k = 1, 2, \ldots, m$ such that

$$\|\gamma_k(t,x) - \gamma_k(t,y)\| \leqslant K_{\gamma_k} \|x - y\|, \quad \forall x, y \in \mathbb{X}; t \in [t_k, s_k].$$

(H5) For any countable sets $D_1, D_2 \subset X$, assume that

$$\beta(f(t, D_1, D_2)) \leqslant L_k \beta(D_1) + N_k \beta(D_2), \quad \forall t \in (s_k, t_{k+1}], \ k = 0, 1, 2, \dots, m,$$

for some positive constants L_k and N_k , $k = 0, 1, 2, \ldots, m$.

Let us denote :

$$K = \max_{k=1,2,\dots,m} K_{\gamma_k}, \quad K^* = \max\{K, \alpha^*\},$$
$$L = \max_{k=0,1,2,\dots,m} (L_k + N_k G^*) (t_{k+1} - s_k)^q.$$
(4.2.1)

4.3Main Result

Theorem 4.3.1. If the semigroup $\mathcal{S}(t)(t \ge 0)$ is equicontinuous, the functions $g(\theta)$ and $\gamma_k(\cdot,\theta)$ are bounded for $k = 1, 2, \ldots, m$, and the assumptions (H1)-(H5) are satisfied. Then the system (4.1.1) has a PC- mild solution, provided that

$$\max\{\Lambda_1, \Lambda_2\} < 1, \tag{4.3.1}$$

where $\Lambda_1 = M(\alpha^* + K)$ and $\Lambda_2 = M(K^* + \frac{4L}{\Gamma(q+1)})$.

Proof. Let us define an operator F on $\mathcal{PC}(J, \mathbb{X})$ as

$$(Fx)(t) = (F_1x)(t) + (F_2x)(t), \qquad (4.3.2)$$

where

$$(F_1x)(t) = \begin{cases} \mathcal{P}(t)(x_0 - g(x)), & t \in [0, t_1];\\ \gamma_k(t, x(t)), & t \in (t_k, s_k], \ k = 1, 2, \dots, m,\\ \mathcal{P}(t - s_k)\gamma_k(s_k, x(s_k)), & t \in (s_k, t_{k+1}], \ k = 1, 2, \dots, m. \end{cases}$$
(4.3.3)

$$(F_2 x)(t) = \begin{cases} \int_{s_k}^t (t-\varrho)^{q-1} \mathcal{R}(t-\varrho) f(\varrho, x(\varrho), \mathfrak{F} x(\varrho)) d\varrho, \\ t \in (s_k, t_{k+1}], k = 0, 1, 2, \dots, m, \\ 0, & \text{otherwise.} \end{cases}$$
(4.3.4)

It is easy to see that F is well defined. From the Definition 4.2.1, one can easily see

that the *PC*-mild solution of the system (4.1.1) is a fixed point *F*. Let $x \in B_R$ for some R > 0, $q_2 = \frac{q-1}{1-q_1} \in (-1,0)$ and $M_1 = \psi(R) \|\phi\|_{L^{\frac{1}{q_1}}(J,\mathbb{R}^+)}$, by using Hölder inequality and (H2), we obtain

$$\int_{0}^{t} \|(t-\varrho)^{q-1} f(\varrho, x(\varrho), \mathfrak{F} x(\varrho))\| d\varrho \leqslant \left(\int_{0}^{t} (t-\varrho)^{q_2} d\varrho \right)^{1-q_1} \psi(R) \|\phi\|_{L^{\frac{1}{q_1}}(J,\mathbb{R}^+)} \\ \leqslant \frac{M_1}{(1+q_2)^{1-q_1}} a^{(1+q_2)(1-q_1)}.$$
(4.3.5)

For our convenience, we divide the proof into following steps :

Step I: We prove that $F(B_R) \subset B_R$, for some R > 0. If this is not true, then for each r > 0, there will exist $x_r \in B_r$ and $t_r \in J$ such that $||(Fx_r(t_r))|| > r$. If $t_r \in [0, t_1]$, then by (4.3.2), (4.3.5), and (H3) we have

$$\begin{aligned} \|(Fx_r)(t_r)\| &\leq M(\|x_0\| + \alpha^* \|x_r - \theta\| + \|g(\theta)\|) + \frac{MM_1}{\Gamma(q)(1+q_2)^{1-q_1}} a^{(1+q_2)(1-q_1)} \\ &\leq M(\alpha^* r + \|x_0\| + \|g(\theta)\|) + \frac{MM_1}{\Gamma(q)(1+q_2)^{1-q_1}} a^{(1+q_2)(1-q_1)}. \end{aligned}$$
(4.3.6)

If $t_r \in (t_k, s_k]$, $k = 1, 2, \ldots, m$, then by (4.3.3) and (H4), we obtain

$$\begin{aligned} \|(Fx_r)(t_r)\| &= \|\gamma_k(t_r, x_r(t_r))\| \\ &\leqslant K_{\gamma_k} \|x_r(t_r)\| + \|\gamma_k(t_r, \theta)\| \\ &\leqslant K_{\gamma_k} r + \alpha, \end{aligned}$$
(4.3.7)

where $\alpha = \max_{k=1,2,\ldots,m} \{ \sup_{t \in J} \| \gamma_k(t,\theta) \| \}$. If $t_r \in (s_k, t_{k+1}], k = 1, 2, \ldots, m$, then by (4.3.2), (4.3.5), and (H4) we have

$$\| (Fx_r)(t_r) \| \leq M(K_{\gamma_k}r + \alpha) + M \int_{s_k}^{t_r} (t_r - s)^{q-1} \| f(s, x_r(s), \mathfrak{F}x_r(s)) \| ds$$

$$\leq M(K_{\gamma_k}r + \alpha) + \frac{MM_1}{\Gamma(q)(1+q_2)^{1-q_1}} a^{(1+q_2)(1-q_1)}$$
(4.3.8)

Combining (4.3.6), (4.3.7) and (4.3.8) with the fact $r < ||(Fx_r)(t_r)||$, we obtain

$$r < \|(Fx_r)(t_r)\| \le M(\alpha^* r + \|x_0\| + \|g(\theta)\|) + M(Kr + \alpha) + \frac{MM_1}{\Gamma(q)(1+q_2)^{1-q_1}} a^{(1+q_2)(1-q_1)}.$$
(4.3.9)

Dividing by r to the both sides of (4.3.9) and taking limit $r \to \infty$, we get

$$1 \leqslant M(\alpha^* + K), \tag{4.3.10}$$

which gives contradiction to (4.3.1).

Step II: We claim that $F_1 : B_R \to B_R$ is Lipschitz continuous. For $t \in [0, t_1]$ and $v, z \in B_R$, using (4.3.3) and (H3) we have

$$\|(F_1v)(t) - (F_1z)(t)\| \leq M \|g(v) - g(z)\| \leq M\alpha^* \|v - z\|.$$
(4.3.11)

If $t \in (t_k, s_k]$, k = 1, 2, ..., m and $v, z \in B_R$, by (4.3.3) and the assumption (H4), we obtain

$$\|(F_1v)(t) - (F_1z)(t)\| \leqslant K_{\gamma_k} \|v(t) - z(t)\| \leqslant MK \|v - z\|.$$
(4.3.12)

For $t \in (s_k, t_{k+1}]$, $k = 1, 2, \ldots, m$ and $v, z \in B_R$, using (H4), we get

$$\| (F_1 v)(t) - (F_1 z)(t) \| \leq M \| \gamma_k(s_k, v(s_k)) - \gamma_k(s_k, z(s_k)) \|$$

$$\leq M K \| v - z \|.$$
 (4.3.13)

From (4.3.11), (4.3.12) and (4.3.13), we obtain

$$||F_1v - F_1z|| \leq MK^* ||v - z||.$$
(4.3.14)

Step III: In this step, we prove that F_2 is continuous on B_R . Let $\{v_n\}$ be a convergent sequence in B_R with $\lim_{n\to\infty} v_n = v$. Since, f is continuous with respect to second and third variables, for each $\rho \in J$, we have

$$\lim_{n \to \infty} f(\varrho, v_n(\varrho), \mathfrak{F}v_n(\varrho)) = f(\varrho, v(\varrho), \mathfrak{F}v(\varrho)).$$
(4.3.15)

So, we can conclude that

$$\sup_{\varrho \in J} \|f(\varrho, v_n(\varrho), \mathfrak{F}v_n(\varrho)) - f(\varrho, v(\varrho), \mathfrak{F}v(\varrho))\| \to 0 \quad \text{as} \quad n \to \infty.$$
(4.3.16)

For $\rho \in [s_k, t]$ and $t \in (s_k, t_{k+1}], k = 0, 1, 2, \dots, m$, by (4.3.15) and (4.3.16), we get

$$\begin{aligned} \|(F_{2}v_{n})(t) - (F_{2}v)(t)\| &\leqslant \frac{M}{\Gamma(q)} \int_{s_{k}}^{t} (t-\varrho)^{q-1} \|f(\varrho, v_{n}(\varrho), \mathfrak{F}v_{n}(\varrho)) \\ &\quad -f(\varrho, v(\varrho), \mathfrak{F}v(\varrho)) \|d\varrho \\ &\leqslant \frac{Ma^{q}}{\Gamma(q+1)} \sup_{\varrho \in J} \|f(\varrho, v_{n}(\varrho), \mathfrak{F}v_{n}(\varrho)) - f(\varrho, v(\varrho), \mathfrak{F}v(\varrho))\| \\ &\rightarrow 0 \quad \text{as} \quad n \to \infty. \end{aligned}$$

$$(4.3.17)$$

Hence,

$$||F_2v_n - F_2v|| \to 0 \quad \text{as} \quad n \to \infty, \tag{4.3.18}$$

which means that F_2 is continuous on B_R .

Step IV: Now, we show $\{F_2v : v \in B_R\}$ is equicontinuous. For any $v \in B_R$ and $s_k \leq t < \tau \leq t_{k+1}$ for k = 0, 1, 2, ..., m

$$\begin{split} \|(F_{2}v)(\tau) - (F_{2}v)(t)\| &= \|\int_{s_{k}}^{\tau} (\tau - \varrho)^{q-1} \mathcal{R}(\tau - \varrho) f(\varrho, v(\varrho), \mathfrak{F}v(\varrho)) d\varrho \\ &- \int_{s_{k}}^{t} (t - \varrho)^{q-1} \mathcal{R}(t - \varrho) f(\varrho, v(\varrho), \mathfrak{F}v(\varrho)) d\varrho \| \\ &\leqslant \|\int_{t}^{\tau} (\tau - \varrho)^{q-1} \mathcal{R}(\tau - \varrho) f(\varrho, v(\varrho), \mathfrak{F}v(\varrho)) d\varrho \| \\ &+ \|\int_{s_{k}}^{t} [(\tau - \varrho)^{q-1} - (t - \varrho)^{q-1}] \mathcal{R}(\tau - \varrho) \\ &f(\varrho, v(\varrho), \mathfrak{F}v(\varrho)) d\varrho \| \\ &+ \|\int_{s_{k}}^{t} (t - \varrho)^{q-1} [\mathcal{R}(\tau - \varrho) - \mathcal{R}(t - \varrho)] \\ &f(\varrho, v(\varrho), \mathfrak{F}v(\varrho)) d\varrho \| \\ &= I_{1} + I_{2} + I_{3}, \end{split}$$

where,

$$I_{1} = \| \int_{t}^{\tau} (\tau - \varrho)^{q-1} \mathcal{R}(\tau - \varrho) f(\varrho, v(\varrho), \mathfrak{F}v(\varrho)) d\varrho \|,$$

$$I_{2} = \| \int_{s_{k}}^{t} [(\tau - \varrho)^{q-1} - (t - \varrho)^{q-1}] \mathcal{R}(\tau - \varrho) f(\varrho, v(\varrho), \mathfrak{F}v(\varrho)) d\varrho \|,$$

$$I_{3} = \| \int_{s_{k}}^{t} (t - \varrho)^{q-1} [\mathcal{R}(\tau - \varrho) - \mathcal{R}(t - \varrho)] f(\varrho, v(\varrho), \mathfrak{F}v(\varrho)) d\varrho \|.$$

Now, we only need to check that I_1 , I_2 and I_3 tend to 0 independently of $v \in B_R$ when $\tau \to t$. By (4.3.5), we have

$$I_1 \leqslant \frac{M_1 M}{\Gamma(q)(1+q_2)^{1-q_1}} (\tau - t)^{(1+q_2)(1-q_1)} \to 0 \text{ as } \tau \to t.$$

For I_2 , by (H2), Lemma 2.3.2, Hölder inequality, and 196, we get that

$$\begin{split} I_2 &\leqslant \frac{M}{\Gamma(q)} \bigg(\int_{s_k}^t [(\tau-\varrho)^{q-1} - (t-\varrho)^{q-1}]^{\frac{1}{1-q_1}} d\varrho \bigg)^{1-q_1} \psi(R) \|\phi\|_{L^{\frac{1}{q_1}}(J,\mathbb{R})} \\ &\leqslant \frac{M_1 M}{\Gamma(q)} \bigg(\int_{s_k}^t [(t-\varrho)^{q_2} - (\tau-\varrho)^{q_2}] d\varrho \bigg)^{1-q_1} \\ &\leqslant \frac{M_1 M}{\Gamma(q)(1+q_2)^{1-q_1}} [(t-s_k)^{1+q_2} + (\tau-s_k)^{1+q_2} - (\tau-t)^{1+q_2}]^{1-q_1} \\ &\leqslant \frac{M_1 M}{\Gamma(q)(1+q_2)^{1-q_1}} (\tau-t)^{(1+q_2)(1-q_1)} \to 0 \quad \text{as} \quad \tau \to t. \end{split}$$

For $t = s_k$, $I_3 = 0$. For $t > s_k$ and $\epsilon > 0$, by (H2), equicontinuity of $\mathcal{S}(t)$, and Lemma 2.3.2, we estimate

$$\begin{split} I_{3} &\leqslant \| \int_{s_{k}}^{t-\epsilon} (t-\varrho)^{q-1} [\mathcal{R}(\tau-\varrho) - \mathcal{R}(t-\varrho)] f(\varrho, v(\varrho), \mathfrak{F}v(\varrho)) d\varrho \| \\ &+ \| \int_{t-\epsilon}^{t} (t-\varrho)^{q-1} [\mathcal{R}(\tau-\varrho) - \mathcal{R}(t-\varrho)] f(\varrho, v(\varrho), \mathfrak{F}v(\varrho)) d\varrho \| \\ &\leqslant \int_{s_{k}}^{t-\epsilon} \| (t-\varrho)^{q-1} f(\varrho, v(\varrho), \mathfrak{F}v(\varrho)) \| d\varrho \sup_{\varrho \in [s_{k}, t-\epsilon]} \| \mathcal{R}(\tau-\varrho) - \mathcal{R}(t-\varrho) \| \\ &+ \frac{2M}{\Gamma(q)} \int_{t-\epsilon}^{t} \| (t-\varrho)^{q-1} f(\varrho, v(\varrho), \mathfrak{F}v(\varrho)) \| d\varrho \\ &\leqslant \frac{M_{1}}{(1+q_{2})^{1-q_{1}}} ((t-s_{k})^{1+q_{2}} - \epsilon^{1+q_{2}})^{1-q_{1}} \sup_{\varrho \in [s_{k}, t-\epsilon]} \| \mathcal{R}(\tau-\varrho) - \mathcal{R}(t-\varrho) \| \\ &+ \frac{2M_{1}M}{\Gamma(q)(1+q_{2})^{1-q_{1}}} \epsilon^{(1+q_{2})(1-q_{1})} \to 0 \quad \text{as} \quad \tau \to t. \end{split}$$

As a result, $||(F_2v)(\tau) - (F_2v)(t)|| \to 0$ independently of $v \in B_R$ as $\tau \to t$, hence $\{F_2(B_R)\}$ is equicontinuous.

Step V: We show that $F : B_R \to B_R$ is a ρ -set contractive map. For any bounded set $D \subset B_R$, by Lemma 2.4.6, we have a countable set $D_0 = \{v_n\} \subset D$ such that

$$\beta(F_2(D)) \leqslant 2\beta(F_2(D_0)).$$
 (4.3.19)

Since $F_2(D_0) \subset F_2(B_R)$ is bounded and equicontinuous, by Lemma 2.4.4, we get

$$\beta(F_2(D_0)) = \max_{t \in [s_k, t_{k+1}], k=0, 1, 2, \dots, m} \beta(F_2(D_0)(t)).$$
(4.3.20)

For every $t \in [s_k, t_{k+1}]$, $k = 0, 1, 2, \ldots, m$, by the assumption (H5), Lemma 2.4.5, and (4.2.1), we get

$$\beta(F_{2}(D_{0})(t)) = \beta\left(\left\{\int_{s_{k}}^{t}(t-\varrho)^{q-1}\mathcal{R}(t-\varrho)f(\varrho,v_{n}(\varrho),\mathfrak{F}v_{n}(\varrho))d\varrho\right\}\right)$$

$$\leqslant \frac{2M}{\Gamma(q)}\int_{s_{k}}^{t}(t-\varrho)^{q-1}\beta(\left\{f(\varrho,v_{n}(\varrho),\mathfrak{F}v_{n}(\varrho))\right\})d\varrho$$

$$\leqslant \frac{2M}{\Gamma(q)}\int_{s_{k}}^{t}(t-\varrho)^{q-1}[L_{k}\beta(D_{0}(\varrho))+N_{k}\beta(\mathfrak{F}D_{0}(\varrho))]d\varrho.(4.3.21)$$

Meanwhile, we have

$$\beta(\mathfrak{F}D_0(\varrho)) \leqslant \beta(\mathfrak{F}D_0) \leqslant \|\mathfrak{F}\|\beta(D_0) \leqslant G^*\beta(D_0) \leqslant G^*\beta(D).$$
(4.3.22)

Therefore,

$$\beta(F_2(D_0)(t)) \leqslant \frac{2M}{\Gamma(q+1)} (L_k + N_k G^*) (t_{k+1} - s_k)^q \beta(D)$$

$$\leqslant \frac{2ML}{\Gamma(q+1)} \beta(D).$$
(4.3.23)

From (4.3.19) and (4.3.23), we obtain

$$\beta(F_2(D)) \leqslant \frac{4ML}{\Gamma(q+1)}\beta(D). \tag{4.3.24}$$

From Lemma 2.4.3 and (4.3.14), for any bounded set $D \subset B_R$,

$$\beta(F_1(D)) \leqslant MK^*\beta(D). \tag{4.3.25}$$

Therefore, by (4.3.24) and (4.3.25), we obtain

$$\beta(F(D)) \leqslant \beta(F_1(D)) + \beta(F_2(D))$$

$$\leqslant M\left(K^* + \frac{4L}{\Gamma(q+1)}\right)\beta(D) = \Lambda_2\beta(D).$$
(4.3.26)

Now combining (4.3.26) with (4.3.1) and Definition 2.5.1, we get that $F: B_R \to B_R$ is a ρ -set-contractive map with $\rho = \Lambda_2$. Hence, Theorem 2.5.3 implies that F has a fixed point in B_R , hence a *PC*-mild solution to (4.1.1).

4.4 Example

Consider a partial differential system with non-instantaneous impulses and nonlocal conditions of fractional order as :

$$\begin{cases} {}^{C}\mathbf{D}^{\frac{1}{2}}v(t,y) + \frac{\partial^{2}}{\partial y^{2}}v(t,y) = \frac{1}{25}\frac{e^{-t}}{1+\epsilon^{t}}v(t,y) + \int_{0}^{t}\frac{1}{50}e^{-s}v(s,y)ds, \\ y \in (0,1), t \in (0,\frac{1}{3}] \cup (\frac{2}{3},1]; \\ v(t,0) = v(t,1) = 0, \quad t \in [0,1]; \\ v(t,y) = \frac{e^{-(t-\frac{1}{3})}}{4}\frac{|v(t,y)|}{1+|v(t,y)|}, \quad y \in (0,1), t \in (\frac{1}{3},\frac{2}{3}]; \\ v(0,y) + \sum_{i=1}^{2}\frac{1}{3^{i}}v(\frac{1}{i},y) = x_{0}(y), \quad y \in [0,1]. \end{cases}$$
(4.4.1)

Let $\mathbb{X} = L^2[0, 1], x_0(y) \in \mathbb{X}$ and Av = v'' with $D(A) = \{v \in \mathbb{X} : v, v' \text{ are absolutely continuous}$ and $v'' \in \mathbb{X}, v(0) = v(1) = 0\}$. By [156], -A is generates an equicontinuous C_0 semigroup $\mathcal{S}(t)(t \ge 0)$ on \mathbb{X} , with $\|\mathcal{S}(t)\| \le 1$, for any $t \ge 0$. Let $a = t_2 = 1, t_0 =$ $s_0 = 0, t_1 = \frac{1}{3}, s_1 = \frac{2}{3}$. By putting

$$\begin{aligned} v(t) &= v(t, \cdot), \\ f(t, v(t), \mathfrak{F}v(t)) &= \frac{1}{25} \frac{e^{-t}}{1 + e^{t}} v(t, \cdot) + \int_{0}^{t} \frac{1}{50} e^{-s} v(\varrho, \cdot) d\varrho, \\ \mathfrak{F}v(t) &= \int_{0}^{t} \frac{1}{50} e^{-\varrho} v(\varrho, \cdot) d\varrho, \\ \gamma_{1}(t, v(t)) &= \frac{e^{-(t - \frac{1}{3})}}{4} \frac{|v(t, \cdot)|}{1 + |v(t, \cdot)|}, \\ g(v) &= \sum_{i=1}^{2} \frac{1}{3^{i}} v(\frac{1}{i}, \cdot), \end{aligned}$$

the parabolic partial differential equation (4.4.1) can be transformed into the abstract form (4.1.1) for m = 1. Observe that the assumptions (H1)-(H5) and condition (4.3.1) hold with

$$q = \frac{1}{2}, \ M = 1, \ \phi(t) = \frac{1}{25} \frac{e^{-t}}{1 + e^{t}} + \frac{1}{50}, \ \psi(r) = r,$$
$$\alpha^* = \frac{4}{9}, \ K = K_{\gamma_1} = \frac{1}{4}, \ L = 0.02, \ \Lambda_1 = 0.69 < 1, \ \Lambda_2 = 0.53 < 1.$$

Therefore, Theorem 4.3.1 is applicable, so the system (4.4.1) has a *PC*-mild solution.

Chapter 5

Approximate Controllability of Nonlocal Non-Instantaneous Impulsive Integro-Differential Equations of Fractional Order

۵

5.1 Introduction

Controllability is one of the most important issue in mathematical control theory and engineering. The problem of controllability for various kinds of differential, integrodifferential equations and impulsive differential equations are studied. In case of controllability, the literature on abstract impulsive differential equations consists of problems involving instantaneous impulses. In [23], Balasubramaniam studied the approximate controllability of impulsive integro-differential system of fractional order having nonlocal conditions, by assuming the compactness of impulsive and nonlocal functions in a Hilbert space. Zhang [198] considered fractional impulsive integro-differential equations in a Hilbert space to study the approximate controllability. Dong et al. [84] also considered fractional impulsive evolution equations

The contents of this chapter are accepted in **Journal of Fractional Calculus and Applications** as Arshi Meraj, Dwijendra N Pandey: "Approximate controllability of fractional integrodifferential evolution equations with nonlocal and non-instantaneous impulsive conditions".

having nonlocal conditions, and studied the approximate controllability via approximate technique.

This chapter deals with the approximate controllability of a certain class of abstract fractional evolution equations of the form :

$${}^{C}\mathbf{D}^{q}x(t) = Ax(t) + f(t, x(t), \mathcal{H}x(t)) + Bu(t), \quad t \in \bigcup_{k=0}^{m} (s_{k}, t_{k+1}],$$

$$x(t) = \gamma_{k}(t, x(t)), \quad t \in \bigcup_{k=1}^{m} (t_{k}, s_{k}],$$

$$x(0) + g(x) = x_{0},$$
(5.1.1)

where 0 < q < 1, J = [0, b], the variable x assumes values in a reflexive separable Banach space \mathbb{X} , A is linear closed operator defined on $D(A) \subset \mathbb{X}$, and generates a C_0 semigroup $\mathcal{S}(t)(t \ge 0)$ on \mathbb{X} , $u \in L^2(J, U)$ is control function, B is \mathbb{X} -valued bounded linear operator defined on U, which is a Banach space. $0 < t_1 < t_2 < \ldots < t_m < t_{m+1} := b, s_0 := 0, s_k \in (t_k, t_{k+1}); k = 1, 2, \ldots, m, f$ and g are given \mathbb{X} -valued functions defined on $J \times \mathbb{X} \times \mathbb{X}$ and $\mathcal{PC}(J, \mathbb{X})$ respectively, $\gamma_k : (t_k, s_k] \times \mathbb{X} \to \mathbb{X}$ are non-instantaneous impulsive functions for all $k = 1, 2, \ldots, m, \ \mathcal{H}x(t) := \int_0^t h(t, \varrho, x(\varrho)) d\varrho, h : \mathbb{D} \times \mathbb{X} \to \mathbb{X}$ is continuous function where $\mathbb{D} := \{(\tau, v) : 0 \le v < \tau \le b\}$ and $x_0 \in \mathbb{X}$.

To the best of our knowledge, there is no work yet reported on the approximate controllability of fractional non-instantaneous impulsive integro-differential equations. Therefore inspired by this fact, we consider the system (5.1.1) to investigate the approximate controllability via Kuratowski measure of noncompactness and ρ -set contraction mapping fixed point theorem without assuming the compactness condition on impulsive and nonlocal functions.

The remaining part of the chapter is arranged as following: Section 5.2 consists of basic definitions, notations and theorems. Section 5.3 concerns with the main results. Finally, section 5.4 deals with an example to illustrate our results.

5.2 Preliminaries and Assumptions

Let X be a separable reflexive Banach space, and $\mathcal{PC}(J, \mathbb{X}) = \{v : J \to \mathbb{X} : v \text{ is continuous at } t \neq t_k, v(t_{k-}) = v(t_k) \text{ and } v(t_{k+}) \text{ exists for all } k = 1, 2, \dots, m\},$ which is a complete normed space with supremum norm. Let $B_r = \{v \in \mathcal{PC}(J, \mathbb{X}) : \|v\| \leq r\} (r > \theta)$, we use θ to denote the zero function in $\mathcal{PC}(J, \mathbb{X})$. Let $M = \sup_{t \in J} \|\mathcal{S}(t)\|_{\mathbb{B}(\mathbb{X})}$, note that $M \ge 1$.

Definition 5.2.1. ([117]) A function $v \in \mathcal{PC}(J, \mathbb{X})$ is known as a mild solution of the problem (5.1.1) if for any $u \in L^2(J,U)$, v satisfies $v(0) = x_0 - g(v)$, $v(t) = \gamma_k(t,v(t))$ for all $t \in \bigcup_{k=1}^m (t_k, s_k]$, and

$$v(t) = \begin{cases} \mathcal{P}(t)(x_0 - g(v)) + \int_0^t (t - \varrho)^{q-1} \mathcal{R}(t - \varrho) [f(\varrho, v(\varrho), \mathcal{H}v(\varrho)) \\ + Bu(\varrho)] d\varrho, \ t \in (0, t_1]; \\ \mathcal{P}(t - s_k) \gamma_k(s_k, v(s_k)) + \int_{s_k}^t (t - \varrho)^{q-1} \mathcal{R}(t - \varrho) [f(\varrho, v(\varrho), \mathcal{H}v(\varrho)) \\ + Bu(\varrho)] d\varrho, \ t \in \cup_{k=1}^m (s_k, t_{k+1}]. \end{cases}$$
(5.2.1)

Let $x^b(x_0, u)$ be the value of state at time b for the initial value x_0 and control u. Denote $\mathbf{R}(b, x_0) = \{x^b(x_0, u) : u \in L^2(J, U)\}$, which is reachable set for (5.1.1) at time b.

Definition 5.2.2. ([132]) If $\overline{\mathbf{R}}(b, x_0) = \mathbb{X}$, the system (5.1.1) is called approximately controllable on J.

Now, consider the linear control system of fractional order

$$^{C}D^{q}x(t) = Ax(t) + Bu(t), \quad t \in J,$$

 $x(0) = x_{0}.$ (5.2.2)

The controllability and resolvent operators associated with linear system (5.2.2) are mentioned below :

$$\Gamma_0^b = \int_0^b (b-\varrho)^{q-1} \mathcal{R}(b-\varrho) B B^* \mathcal{R}^*(b-\varrho) d\varrho, \qquad (5.2.3)$$

$$R(\lambda, \Gamma_0^b) = (\lambda I + \Gamma_0^b)^{-1}, \, \lambda > 0, \qquad (5.2.4)$$

respectively, where $(\cdot)^*$ denotes the adjoint operator of (\cdot) . Observe that the operator Γ_0^b is linear and bounded. Now, we state the following hypothesis :

(H0) $\lambda R(\lambda, \Gamma_0^b) \to 0$ as $\lambda \to 0^+$ in the strong operator topology.

Theorem 5.2.1. ([131]) Let \mathbb{Z} is a reflexive separable complete norm space and \mathbb{Z}^* denotes dual space of \mathbb{Z} , $\Gamma : \mathbb{Z}^* \to \mathbb{Z}$ is a symmetric map. Then the statements given below are equivalent :

- (i) Γ is positive map, which means $\langle v^*, \Gamma v^* \rangle > 0$ for all nonzero $v^* \in \mathbb{Z}^*$.
- (ii) $\lambda(\lambda I + \Gamma \mathfrak{J})^{-1}(v)$ converges to zero strongly as $\lambda \to 0^+$, $\forall v \in \mathbb{Z}$, here \mathfrak{J} is the duality map from $\mathbb{Z} \to \mathbb{Z}^*$.

Lemma 5.2.2. ([132]) The linear control system (5.2.2) is approximately controllable on J iff (H0) satisfies.

Proof. The system (5.2.2) is approximately controllable on J iff $\langle v, \Gamma_0^b v \rangle > 0$, for each nonzero $v \in X$ (see Theorem 4.1.7 of [62]), so the lemma is straightforward consequence of Theorem 5.2.1.

Remark 5.2.3. Notice that the system (5.2.2) is approximately controllable on J iff $\langle v, \Gamma_0^b v \rangle = \int_0^b (b-\varrho)^{q-1} ||B^* \mathcal{R}^*(b-\varrho)v||^2 d\varrho > 0$, for all nonzero $v \in \mathbb{X}$, this is further equivalent to $B^* \mathcal{R}^*(b-\varrho)v = 0, \ 0 \leq \varrho < b \Longrightarrow v = 0$.

Now, let us state the basic assumptions which are useful to prove our main results:

(H1) S(t)(t > 0) is a compact semigroup.

- (H2) The function $f(t, \cdot, \cdot)$ is continuous on $\mathbb{X} \times \mathbb{X}$, for every fixed $t \in J$, and $f(\cdot, v, z)$ is Lebesgue measurable on J, for all $(v, z) \in \mathbb{X} \times \mathbb{X}$.
- (H3) There exist a nondecreasing continuous function ψ from \mathbb{R}^+ to \mathbb{R}^+ , a constant $0 < q_1 < q$, and a function $\phi \in L^{\frac{1}{q_1}}(J, \mathbb{R}^+)$ satisfying

$$\|f(t,v,z)\| \leqslant \phi(t)\psi(\|v\|), \quad \forall v, z \in \mathbb{X}; t \in J.$$

(H4) The function g is continuous and

$$||g(v) - g(z)|| \leq \alpha ||v - z||, \quad \forall v, z \in \mathcal{PC}(J, \mathbb{X}),$$

for some constant $\alpha > 0$.

$$\|\gamma_k(t,v) - \gamma_k(t,z)\| \leqslant K_{\gamma_k} \|v - z\|, \quad \forall v, z \in \mathbb{X}; t \in [t_k, s_k],$$

for some constants $K_{\gamma_k} > 0$, $k = 1, 2, \ldots, m$.

For our convenience, we use the following notations :

$$K = \max_{k=1,2,\dots,m} K_{\gamma_k}, \quad K_1 = \max\{K,\alpha\}, \quad M_B := \|B\|,$$

$$\overline{M} = \frac{b^q M^2 (M_B)^2}{q\lambda(\Gamma(q))^2}, \quad q_2 = \frac{q-1}{1-q_1} \in (-1,0), \quad M_1 = \psi(R) \|\phi\|_{L^{\frac{1}{q_1}}(J,\mathbb{R}^+)},$$

$$M_b = \frac{MM_1}{\Gamma(q)(1+q_2)^{1-q_1}} b^{(1+q_2)(1-q_1)}.$$
(5.2.5)

For any $v \in \mathcal{PC}(J, \mathbb{X})$, we choose the control function for the nonlinear system (5.1.1) as given below :

$$u(t) = u_{\lambda}(t, v) = B^* \mathcal{R}^*(b-t) R(\lambda, \Gamma_0^b) p(v), \qquad (5.2.6)$$

where

$$p(v) = \begin{cases} x^b - \mathcal{P}(b)(x_0 - g(v)) - \int_0^b (b - \varrho)^{q-1} \mathcal{R}(b - \varrho) f(\varrho, v(\varrho), \mathcal{H}v(\varrho)) d\varrho, \ t \in (0, t_1], \\ x^b - \mathcal{P}(b - s_k) \gamma_k(s_k, v(s_k)) - \int_{s_k}^b (b - \varrho)^{q-1} \mathcal{R}(b - \varrho) f(\varrho, v(\varrho), \mathcal{H}v(\varrho)) d\varrho, \\ t \in \bigcup_{k=1}^m (s_k, t_{k+1}]. \end{cases}$$

By using the control function (5.2.6), for any $\lambda > 0$ define the operator F_{λ} on $\mathcal{PC}(J, \mathbb{X})$ as following :

$$(F_{\lambda}v)(t) = (\Phi_{\lambda}v)(t) + (\widetilde{\Phi_{\lambda}}v)(t), \qquad (5.2.7)$$

where

$$(\Phi_{\lambda}v)(t) = \begin{cases} \mathcal{P}(t)(x_0 - g(v)), & t \in [0, t_1], \\ \gamma_k(t, v(t)), & t \in \bigcup_{k=1}^m (t_k, s_k], \\ \mathcal{P}(t - s_k)\gamma_k(s_k, v(s_k)), & t \in \bigcup_{k=1}^m (s_k, t_{k+1}], \end{cases}$$
(5.2.8)

$$(\widetilde{\Phi_{\lambda}}v)(t) = \begin{cases} \int_{s_{k}}^{t} (t-\varrho)^{q-1} \mathcal{R}(t-\varrho) [f(\varrho, v(\varrho), \mathcal{H}v(\varrho)) + Bu_{\lambda}(\varrho, v)] d\varrho, \\ t \in \bigcup_{k=0}^{m} (s_{k}, t_{k+1}], \\ 0, & \text{otherwise.} \end{cases}$$
(5.2.9)

5.3 Main Results

5.3.1 Existence of mild solutions

Theorem 5.3.1. Let the hypotheses (H1)-(H5) are satisfied, the functions $g(\theta)$ and $\gamma_k(\cdot, \theta)$ are bounded for k = 1, 2, ..., m. Then the system (5.1.1) has a PC- mild solution, provided that

$$\rho := M K_1 < 1. \tag{5.3.1}$$

Proof. First let us observe that, for $v \in B_R$ (R > 0) with the help of (H3) and Hölder inequality, we have

$$\int_{0}^{t} \|(t-\varrho)^{q-1} f(\varrho, v(\varrho), \mathcal{H}v(\varrho))\| d\varrho \leqslant \left(\int_{0}^{t} (t-\varrho)^{q_2} d\varrho \right)^{1-q_1} \psi(R) \|\phi\|_{L^{\frac{1}{q_1}}(J,\mathbb{R}^+)} \\ \leqslant \frac{M_1}{(1+q_2)^{1-q_1}} b^{(1+q_2)(1-q_1)}.$$
(5.3.2)

We divide the proof into following steps :

Step 1: For any $\lambda > 0$, we want to show that there exists a constant $R = R(\lambda) > 0$, satisfying $F_{\lambda}(B_R) \subset B_R$. Let $v \in B_r$ for any positive constant r. If $t \in [0, t_1]$, then by using (5.2.6) and (5.3.2), we have

$$u_{\lambda}(t,v) = B^{*}\mathcal{R}^{*}(b-t)R(\lambda,\Gamma_{0}^{b})\left[x^{b}-\mathcal{P}(b)(x_{0}-g(v))\right]$$
$$-\int_{0}^{b}(b-\varrho)^{q-1}\mathcal{R}(b-\varrho)f(\varrho,v(\varrho),\mathcal{H}v(\varrho))d\varrho\right]$$
$$\|u_{\lambda}(t,v)\| \leq \frac{MM_{B}}{\lambda\Gamma(q)}\left[\|x^{b}\|+M(\|x_{0}\|+\alpha\|v-\theta\|+\|g(\theta)\|)+M_{b}\right]$$
$$\leq \frac{MM_{B}}{\lambda\Gamma(q)}\left[\|x^{b}\|+M(\alpha r+\|x_{0}\|+\|g(\theta)\|)+M_{b}\right], \quad (5.3.3)$$

and from (5.2.7), (5.3.3), we obtain

$$(F_{\lambda}v)(t) = \mathcal{P}(t)(x_{0} - g(v)) + \int_{0}^{t} (t - \varrho)^{q-1}\mathcal{R}(t - \varrho)f(\varrho, v(\varrho), \mathcal{H}v(\varrho))d\varrho + \int_{0}^{t} (t - \varrho)^{q-1}\mathcal{R}(t - \varrho)Bu_{\lambda}(\varrho, v)d\varrho |(F_{\lambda}v)(t)|| \leq M(\alpha r + ||x_{0}|| + ||g(\theta)||) + M_{b} + \int_{0}^{t} (t - \varrho)^{q-1}||\mathcal{R}(t - \varrho)|| ||Bu_{\lambda}(\varrho, v)||d\varrho \leq M(\alpha r + ||x_{0}|| + ||g(\theta)||) + M_{b} + \frac{b^{q}M^{2}(M_{B})^{2}}{q\lambda(\Gamma(q))^{2}} \Big[||x^{b}|| + M(\alpha r + ||x_{0}|| + ||g(\theta)||) + M_{b}\Big].$$
(5.3.4)

For $t \in (t_k, s_k]$; $k = 1, 2, \ldots, m$, then by (5.2.7) and (H5), we estimate

$$\begin{aligned} \|(F_{\lambda}v)(t)\| &= \|\gamma_{k}(t,v(t))\| \\ &\leqslant K_{\gamma_{k}}\|v(t)\| + \|\gamma_{k}(t,\theta)\| \\ &\leqslant Kr + \alpha^{*} \leqslant M(Kr + \alpha^{*}), \end{aligned}$$
(5.3.5)

where $\alpha^* = \max_{k=1,2,...,m} \{ \sup_{t \in J} \| \gamma_k(t,\theta) \| \}$. If $t \in (s_k, t_{k+1}]; k = 1, 2, ..., m$ then (5.2.6), (5.2.7) and (5.3.2) yield the following estimations

$$\|u_{\lambda}(t,v)\| \leq \frac{MM_B}{\lambda\Gamma(q)} \Big[\|x^b\| + M(Kr + \alpha^*) + M_b \Big], \qquad (5.3.6)$$

$$\|(F_{\lambda}v)(t)\| \leq M(Kr + \alpha^{*}) + M_{b} + \frac{b^{q}M^{2}(M_{B})^{2}}{q\lambda(\Gamma(q))^{2}} \left[\|x^{b}\| + M(Kr + \alpha^{*}) + M_{b}\right].$$
(5.3.7)

Combining (5.3.4), (5.3.5) and (5.3.7), we obtain

$$\|(F_{\lambda}v)(t)\| \leq M_{b} + M(Kr + \alpha^{*}) + M(\alpha r + \|x_{0}\| + \|g(\theta)\|) + \overline{M}M(Kr + \alpha^{*}) + \overline{M}\left[\|x^{b}\| + M(\alpha r + \|x_{0}\| + \|g(\theta)\|) + M_{b}\right].$$
(5.3.8)

Then, we get that for large enough R > 0, $F_{\lambda}(B_R) \subset B_R$ holds.

Step 2: We show that $\Phi_{\lambda} : B_R \to B_R$ is Lipschitz continuous. Let $v, z \in B_R$, for $t \in [0, t_1]$, using (5.2.8) and (H4)

$$\|(\Phi_{\lambda}v)(t) - (\Phi_{\lambda}z)(t)\| \leq M \|g(v) - g(z)\| \leq M\alpha \|v - z\|,$$
 (5.3.9)

for $t \in (t_k, s_k]$, k = 1, 2, ..., m, by (5.2.8) and the assumption (H5)

$$\|(\Phi_{\lambda}v)(t) - (\Phi_{\lambda}z)(t)\| \leqslant K_{\gamma_k} \|v(t) - z(t)\| \leqslant MK \|v - z\|,$$
(5.3.10)

for $t \in (s_k, t_{k+1}]$, k = 1, 2, ..., m, using (H5), we have

$$\begin{aligned} \|(\Phi_{\lambda}v)(t) - (\Phi_{\lambda}z)(t)\| &\leqslant M \|\gamma_k(s_k, v(s_k)) - \gamma_k(s_k, z(s_k))\| \\ &\leqslant MK \|v - z\|. \end{aligned}$$
(5.3.11)

From (5.3.9), (5.3.10) and (5.3.11), we obtain

$$\|\Phi_{\lambda}v - \Phi_{\lambda}z\| \leqslant MK_1 \|v - z\|.$$
(5.3.12)

Step 3: Let $\{v_n\}$ be a sequence in B_R such that $\lim_{n\to\infty} v_n = v$ in B_R . Since, f is continuous with respect to second and third variables, for each $\rho \in J$, we have

$$\lim_{n \to \infty} f(\varrho, v_n(\varrho), \mathcal{H}v_n(\varrho)) = f(\varrho, v(\varrho), \mathcal{H}v(\varrho)).$$
(5.3.13)

So, we can conclude that

$$\sup_{\varrho \in J} \|f(\varrho, v_n(\varrho), \mathcal{H}v_n(\varrho)) - f(\varrho, v(\varrho), \mathcal{H}v(\varrho))\| \to 0 \quad \text{as} \quad n \to \infty.$$
(5.3.14)

For $t \in (s_k, t_{k+1}]$, (H5) and (5.3.14) yield the following

$$\begin{aligned} \|p(v_n) - p(v)\| &\leqslant M \|\gamma_k(s_k, v_n(s_k)) - \gamma_k(s_k, v(s_k))\| \\ &+ \frac{M}{\Gamma(q)} \int_{s_k}^b (b - \varrho)^{q-1} \|f(\varrho, v_n(\varrho), \mathcal{H}v_n(\varrho)) - f(\varrho, v(\varrho), \mathcal{H}v(\varrho))\| d\varrho \\ &\leqslant M \|\gamma_k(s_k, v_n(s_k)) - \gamma_k(s_k, v(s_k))\| \\ &+ \frac{Mb^q}{\Gamma(q+1)} \sup_{\varrho \in J} \|f(\varrho, v_n(\varrho), \mathcal{H}v_n(\varrho)) - f(\varrho, v(\varrho), \mathcal{H}v(\varrho))\| \\ &\to 0 \quad \text{as} \quad n \to \infty. \end{aligned}$$
(5.3.15)

Therefore, (5.2.6) and (5.3.15) imply that

$$||u_{\lambda}(\varrho, v_n) - u_{\lambda}(\varrho, v)|| \to 0 \quad \text{as} \quad n \to \infty,$$
 (5.3.16)

also (5.2.9), (5.3.14) and (5.3.16) yield

$$\begin{split} \|(\widetilde{\Phi_{\lambda}}v_{n})(t) - (\widetilde{\Phi_{\lambda}}v)(t)\| &\leqslant \frac{M}{\Gamma(q)} \int_{s_{k}}^{t} (t-\varrho)^{q-1} \|f(\varrho, v_{n}(\varrho), \mathcal{H}v_{n}(\varrho)) \\ &-f(\varrho, v(\varrho), \mathcal{H}v(\varrho)) \|d\varrho \\ &+ \frac{M}{\Gamma(q)} \int_{s_{k}}^{t} (t-\varrho)^{q-1} \|B\| \|u_{\lambda}(\varrho, v_{n}) - u_{\lambda}(\varrho, v)\| d\varrho \\ &\leqslant \frac{Mb^{q}}{\Gamma(q+1)} \sup_{\varrho \in J} \|f(\varrho, v_{n}(\varrho), \mathcal{H}v_{n}(\varrho)) - f(\varrho, v(\varrho), \mathcal{H}v(\varrho))\| \\ &+ \frac{b^{q} M M_{B}}{\Gamma(q+1)} \sup_{\varrho \in J} \|u_{\lambda}(\varrho, v_{n}) - u_{\lambda}(\varrho, v)\| \\ &\to 0 \quad \text{as} \quad n \to \infty, \end{split}$$

$$(5.3.17)$$

which means that $\widetilde{\Phi_{\lambda}}$ is continuous in B_R .

Step 4: We claim that $\widetilde{\Phi_{\lambda}} : B_R \to B_R$ is compact, this result will be proved by using Arzela-Ascoli theorem. For this we need to prove : (i): For any $t \in J$, the set $\{(\widetilde{\Phi_{\lambda}}v)(t) : v \in B_R\}$ is relatively compact in X. For $t \notin (s_k, t_{k+1}], k = 0, 1, 2, ..., m$, obviously the set $\{(\widetilde{\Phi_{\lambda}}v)(t) : v \in B_R\} = \{0\}$ which is compact in X. Let $t \in (s_k, t_{k+1}], k = 0, 1, 2, ..., m$ be fixed. For any $\varepsilon \in (s_k, t)$ and $\delta > 0$, we define an operator $\widetilde{\Phi_{\lambda}}^{\varepsilon,\delta}$ on B_R as following :

$$\begin{split} (\widetilde{\Phi_{\lambda}}^{\varepsilon,\delta}v)(t) &= q \int_{s_{k}}^{t-\varepsilon} \int_{\delta}^{\infty} \varpi(t-\varrho)^{q-1} \Psi_{q}(\varpi) \mathcal{S}((t-\varrho)^{q}\varpi) [f(\varrho, v(\varrho), \mathcal{H}v(\varrho)) \\ &+ Bu_{\lambda}(\varrho, v)] d\varpi d\varrho \\ &= \mathcal{S}(\varepsilon^{q}\delta) q \int_{s_{k}}^{t-\varepsilon} \int_{\delta}^{\infty} \varpi(t-\varrho)^{q-1} \Psi_{q}(\varpi) \mathcal{S}((t-\varrho)^{q}\varpi - \varepsilon^{q}\delta) \\ &\quad [f(\varrho, v(\varrho), \mathcal{H}v(\varrho)) + Bu_{\lambda}(\varrho, v)] d\varpi d\varrho \\ &:= \mathcal{S}(\varepsilon^{q}\delta) y(t, \varepsilon). \end{split}$$

Since $y(t,\varepsilon)$ is bounded on B_R and $\mathcal{S}(\varepsilon^q \delta)(\varepsilon^q \delta > 0)$ is compact on \mathbb{X} , we conclude that the set $\{(\widetilde{\Phi_{\lambda}}^{\varepsilon,\delta}v)(t) : v \in B_r\}$ is relatively compact in \mathbb{X} . Now, observe that

$$\begin{split} \|(\widetilde{\Phi_{\lambda}}v)(t) - (\widetilde{\Phi_{\lambda}}^{\varepsilon,\delta}v)(t)\| &= q \left\| \int_{s_{k}}^{t} \int_{0}^{\delta} \varpi(t-\varrho)^{q-1}\Psi_{q}(\varpi)\mathcal{S}((t-\varrho)^{q}\varpi) \right. \\ &\left[f(\varrho, v(\varrho), \mathcal{H}v(\varrho)) + Bu_{\lambda}(\varrho, v) \right] d\varpi d\varrho \\ &+ \int_{s_{k}}^{t} \int_{\delta}^{\infty} \varpi(t-\varrho)^{q-1}\Psi_{q}(\varpi)\mathcal{S}((t-\varrho)^{q}\varpi) \\ &\left[f(\varrho, v(\varrho), \mathcal{H}v(\varrho)) + Bu_{\lambda}(\varrho, v) \right] d\varpi d\varrho \\ &- \int_{s_{k}}^{t-\varepsilon} \int_{\delta}^{\infty} \varpi(t-\varrho)^{q-1}\Psi_{q}(\varpi)\mathcal{S}((t-\varrho)^{q}\varpi) \\ &\left[f(\varrho, v(\varrho), \mathcal{H}v(\varrho)) + Bu_{\lambda}(\varrho, v) \right] d\varpi d\varrho \right\| \\ &= q \left\| \int_{s_{k}}^{t} \int_{0}^{\delta} \varpi(t-\varrho)^{q-1}\Psi_{q}(\varpi)\mathcal{S}((t-\varrho)^{q}\varpi) \\ &\left[f(\varrho, v(\varrho), \mathcal{H}v(\varrho)) + Bu_{\lambda}(\varrho, v) \right] d\varpi d\varrho \\ &+ \int_{t-\varepsilon}^{t} \int_{\delta}^{\infty} \varpi(t-\varrho)^{q-1}\Psi_{q}(\varpi)\mathcal{S}((t-\varrho)^{q}\varpi) \\ &\left[f(\varrho, v(\varrho), \mathcal{H}v(\varrho)) + Bu_{\lambda}(\varrho, v) \right] d\varpi d\varrho \\ &\left. \right\| \\ &\leqslant q(I_{1}+I_{2}), \end{split}$$
(5.3.18)

where

$$I_{1} = \| \int_{s_{k}}^{t} \int_{0}^{\delta} \varpi(t-\varrho)^{q-1} \Psi_{q}(\varpi) \mathcal{S}((t-\varrho)^{q} \varpi) [f(\varrho, v(\varrho), \mathcal{H}v(\varrho)) \\ + Bu_{\lambda}(\varrho, v)] d\varpi d\varrho \|,$$

$$I_{2} = \| \int_{t-\epsilon}^{t} \int_{\delta}^{\infty} \varpi(t-\varrho)^{q-1} \Psi_{q}(\varpi) \mathcal{S}((t-\varrho)^{q} \varpi) [f(\varrho, v(\varrho), \mathcal{H}v(\varrho)) \\ + Bu_{\lambda}(\varrho, v)] d\varpi d\varrho \|.$$

Now, by (5.3.2) and (5.3.6) we have

$$I_{1} \leqslant M\left(\int_{0}^{\delta} \varpi \Psi_{q}(\varpi) d\varpi\right) \left[\int_{s_{k}}^{t} (t-\varrho)^{q-1} \|f(\varrho, v(\varrho), \mathcal{H}v(\varrho))\| d\varrho + M_{B} \|u_{\lambda}\| \frac{b^{q}}{q}\right]$$

$$\leqslant M\left(\int_{0}^{\delta} \varpi \Psi_{q}(\varpi) d\varpi\right) \left[\frac{M_{1}}{(1+q_{2})^{1-q_{1}}} b^{(1+q_{2})(1-q_{1})} + \frac{M(M_{B})^{2}}{\lambda \Gamma(q)} [\|x^{b}\| + M(Kr + \alpha^{*}) + M_{b}] \frac{b^{q}}{q}\right].$$
(5.3.19)

Similarly using Remark 2.3.1, we can obtain

$$I_{2} \leqslant M\left(\int_{\delta}^{\infty} \varpi \Psi_{q}(\varpi) d\varpi\right) \left[\frac{M_{1}}{(1+q_{2})^{1-q_{1}}} \epsilon^{(1+q_{2})(1-q_{1})} + \frac{M(M_{B})^{2}}{\lambda \Gamma(q)} \right]$$

$$[||x^{b}|| + M(Kr + \alpha^{*}) + M_{b}] \frac{\epsilon^{q}}{q}$$

$$\leqslant \frac{M}{\Gamma(q+1)} \left[\frac{M_{1}}{(1+q_{2})^{1-q_{1}}} \epsilon^{(1+q_{2})(1-q_{1})} + \frac{M(M_{B})^{2}}{\lambda \Gamma(q)} \right]$$

$$[||x^{b}|| + M(Kr + \alpha^{*}) + M_{b}] \frac{\epsilon^{q}}{q} .$$
(5.3.20)

Therefore by (5.3.18), (5.3.19), and (5.3.20) we conclude that

$$\|(\widetilde{\Phi_{\lambda}}v)(t) - (\widetilde{\Phi_{\lambda}}^{\varepsilon,\delta}v)(t)\| \to 0 \quad as \quad \varepsilon \to 0, \delta \to 0.$$

This implies that the set $\{(\widetilde{\Phi_{\lambda}}v)(t) : v \in B_R\}$ is relatively compact in \mathbb{X} for $t \in (s_k, t_{k+1}], k = 0, 1, 2, \dots, m$. (ii): The family of functions $\{\widetilde{\Phi_{\lambda}}v : v \in B_R\}$ is equicontinuous. For any $v \in B_R$ and $s_k \leqslant t' < t'' \leqslant t_{k+1}$ for $k = 0, 1, 2, \dots, m$, we have

$$\begin{split} \|(\widetilde{\Phi_{\lambda}}v)(t'') - (\widetilde{\Phi_{\lambda}}v)(t')\| &\leq \left\| \int_{t'}^{t''} (t'' - \varrho)^{q-1} \mathcal{R}(t'' - \varrho) f(\varrho, v(\varrho), \mathcal{H}v(\varrho)) d\varrho \right\| \\ &+ \left\| \int_{t'}^{t''} (t'' - \varrho)^{q-1} \mathcal{R}(t'' - \varrho) Bu_{\lambda}(\varrho, v) d\varrho \right\| \\ &+ \left\| \int_{s_{k}}^{t'} [(t'' - \varrho)^{q-1} - (t' - \varrho)^{q-1}] \mathcal{R}(t'' - \varrho) \right. \\ & \left. f(\varrho, v(\varrho), \mathcal{H}v(\varrho)) d\varrho \right\| \\ &+ \left\| \int_{s_{k}}^{t'} [(t'' - \varrho)^{q-1} - (t' - \varrho)^{q-1}] \mathcal{R}(t'' - \varrho) \right. \\ & \left. Bu_{\lambda}(\varrho, v) d\varrho \right\| \\ &+ \left\| \int_{s_{k}}^{t'} (t' - \varrho)^{q-1} [\mathcal{R}(t'' - \varrho) - \mathcal{R}(t' - \varrho)] \right. \\ & \left. f(\varrho, v(\varrho), \mathcal{H}v(\varrho)) d\varrho \right\| \\ &+ \left\| \int_{s_{k}}^{t'} (t' - \varrho)^{q-1} [\mathcal{R}(t'' - \varrho) - \mathcal{R}(t' - \varrho)] \right. \\ & \left. Bu_{\lambda}(\varrho, v) d\varrho \right\| \\ &= \left. J_{1} + J_{2} + J_{3} + J_{4} + J_{5} + J_{6}, \end{split}$$

Now, we only need to check that J_1, J_2, J_3, J_4, J_5 and J_6 tends to 0 independently of $v \in B_R$ when $t'' \to t'$. By (5.3.2), we have

$$J_{1} \leqslant \frac{M_{1}M}{\Gamma(q)(1+q_{2})^{1-q_{1}}}(t''-t')^{(1+q_{2})(1-q_{1})} \to 0 \quad as \quad t'' \to t',$$

$$J_{2} \leqslant \frac{MM_{B}}{\Gamma(q+1)}(t''-t')^{q} ||u_{\lambda}|| \to 0 \quad as \quad t'' \to t'.$$

By (H3), Lemma 2.3.2, and Hölder inequality, we get that

$$J_{3} \leqslant \frac{M}{\Gamma(q)} \left(\int_{s_{k}}^{t'} [(t''-\varrho)^{q-1} - (t'-\varrho)^{q-1}]^{\frac{1}{1-q_{1}}} d\varrho \right)^{1-q_{1}} \psi(R) \|\phi\|_{L^{\frac{1}{q_{1}}}(J,\mathbb{R})}$$

$$\leqslant \frac{M_{1}M}{\Gamma(q)} \left(\int_{s_{k}}^{t'} [(t'-\varrho)^{q_{2}} - (t''-\varrho)^{q_{2}}] d\varrho \right)^{1-q_{1}}$$

$$\leqslant \frac{M_{1}M}{\Gamma(q)(1+q_{2})^{1-q_{1}}} [(t'-s_{k})^{1+q_{2}} + (t''-s_{k})^{1+q_{2}} - (t''-t')^{1+q_{2}}]^{1-q_{1}}$$

$$\leqslant \frac{M_{1}M}{\Gamma(q)(1+q_{2})^{1-q_{1}}} (t''-t')^{(1+q_{2})(1-q_{1})} \to 0 \quad as \quad t'' \to t',$$

and

$$J_4 \leqslant \frac{MM_B}{\Gamma(q+1)} \left[(t'' - s_k)^q - (t' - s_k)^q - (t'' - t')^q \right] \|u_\lambda\| \to 0 \quad as \quad t'' \to t'.$$

For $t' = s_k$, it is easy to see that $J_5 = 0$. For $t' > s_k$ and $\epsilon > 0$ (small enough), by Lemma 2.3.2 and (H3), we obtain

$$\begin{split} J_{5} &\leqslant \| \int_{s_{k}}^{t'-\epsilon} (t'-\varrho)^{q-1} [\mathcal{R}(t''-\varrho) - \mathcal{R}(t'-\varrho)] f(\varrho, v(\varrho), \mathcal{H}v(\varrho)) d\varrho \| \\ &+ \| \int_{t'-\epsilon}^{t'} (t'-\varrho)^{q-1} [\mathcal{R}(t''-\varrho) - \mathcal{R}(t'-\varrho)] f(\varrho, v(\varrho), \mathcal{H}v(\varrho)) d\varrho \| \\ &\leqslant \int_{s_{k}}^{t'-\epsilon} \| (t'-\varrho)^{q-1} f(\varrho, v(\varrho), \mathcal{H}v(\varrho)) \| d\varrho \\ &\sup_{\varrho \in [s_{k}, t'-\epsilon]} \| \mathcal{R}(t''-\varrho) - \mathcal{R}(t'-\varrho) \| \\ &+ \frac{2M}{\Gamma(q)} \int_{t'-\epsilon}^{t'} \| (t'-\varrho)^{q-1} f(\varrho, v(\varrho), \mathcal{H}v(\varrho)) \| d\varrho \\ &\leqslant \frac{M_{1}}{(1+q_{2})^{1-q_{1}}} ((t'-s_{k})^{1+q_{2}} - \epsilon^{1+q_{2}})^{1-q_{1}} \\ &\sup_{\varrho \in [s_{k}, t'-\epsilon]} \| \mathcal{R}(t''-\varrho) - \mathcal{R}(t'-\varrho) \| \\ &+ \frac{2M_{1}M}{\Gamma(q)(1+q_{2})^{1-q_{1}}} \epsilon^{(1+q_{2})(1-q_{1})} \to 0 \quad as \quad t'' \to t', \ \epsilon \to 0, \end{split}$$

similarly

$$J_{6} \leqslant \frac{M_{B}}{q} [(t'-s_{k})^{q}-\epsilon^{q}] \|u_{\lambda}\| \sup_{\varrho \in [s_{k},t'-\epsilon]} \|\mathcal{R}(t''-\varrho)-\mathcal{R}(t'-\varrho)\| \\ + \frac{2MM_{B}}{\Gamma(q+1)} \epsilon^{q} \|u_{\lambda}\| \to 0 \quad as \quad t'' \to t', \ \epsilon \to 0.$$

So, $\|(\widetilde{\Phi_{\lambda}}v)(t'') - (\widetilde{\Phi_{\lambda}}v)(t')\| \to 0$ independently of $v \in B_R$ as $t'' \to t'$, hence $\{\widetilde{\Phi_{\lambda}}(B_R)\}$ is equicontinuous. So, $\widetilde{\Phi_{\lambda}}$ is compact on B_R by Arzela-Ascoli theorem.

Step 5: We show that F_{λ} is ρ -set contractive map. For any bounded set $D \subset B_R$, by Lemma 2.4.6, we have a countable set $D_0 = \{v_n\} \subset D$ such that

$$\beta(\widetilde{\Phi_{\lambda}}(D)) \leqslant 2\beta(\widetilde{\Phi_{\lambda}}(D_0)).$$

Since $\widetilde{\Phi_{\lambda}}(D_0) \subset \widetilde{\Phi_{\lambda}}(B_R)$ is bounded and equicontinuous, by Lemma 2.4.4 we obtain

$$\beta(\widetilde{\Phi_{\lambda}}(D_0)) = \max_{t \in [s_k, t_{k+1}], k=0, 1, 2, \dots, m} \beta(\widetilde{\Phi_{\lambda}}(D_0)(t)).$$

By Proposition 2.4.2(i) and Step 4(i), we have $\beta(\widetilde{\Phi_{\lambda}}(D_0)(t)) = 0$ for all $t \in J$, therefore $\beta(\widetilde{\Phi_{\lambda}}(D)) = 0$. From (5.3.12) and Lemma 2.4.3, we know that for any bounded set $D \subset B_R$

$$\beta(\Phi_{\lambda}(D)) \leqslant MK_1\beta(D).$$

Thus, by Proposition 2.4.2(ii)

$$\beta(F_{\lambda}(D)) \leqslant \beta(\Phi_{\lambda}(D)) + \beta(\widetilde{\Phi_{\lambda}}(D))$$

$$\leqslant MK_{1}\beta(D) = \rho\beta(D).$$
(5.3.21)

Now combining (5.3.21) with (5.3.1) and Definition 2.5.1, we conclude that F_{λ} : $B_R \to B_R$ is a ρ -set-contractive map with $\rho = MK_1$. Hence from Theorem 2.5.3, it follows that there exists a fixed point of F_{λ} in B_R , hence a PC-mild solution of (5.1.1).

5.3.2 Approximate controllability

Theorem 5.3.2. Assume that the assumptions (H0)-(H5) are satisfied. Moreover, assume that the functions $f, g, \gamma_k (k = 1, 2, ..., m)$ are uniformly bounded by positive constants L_1, L_2 and $N_k (k = 1, 2, ..., m)$. Then, the system (5.1.1) is approximately controllable on J.

Proof. Let x_{λ} be a mild solution of the problem (5.1.1) under the control

$$u_{\lambda}(t, x_{\lambda}) = B^* \mathcal{R}^*(b-t) R(\lambda, \Gamma_0^b) p(x_{\lambda}),$$

and satisfies the equality

$$x_{\lambda}(b) = x^{b} - \lambda R(\lambda, \Gamma_{0}^{b}) p(x_{\lambda}), \qquad (5.3.22)$$

where

$$p(x_{\lambda}) = \begin{cases} x^{b} - \mathcal{P}(b)(x_{0} - g(x_{\lambda})) - \int_{0}^{b} (b - \varrho)^{q-1} \mathcal{R}(b - \varrho) f(\varrho, x_{\lambda}(\varrho), \mathcal{H}x_{\lambda}(\varrho)) d\varrho, \\ t \in (0, t_{1}], \\ x^{b} - \mathcal{P}(b - s_{k})\gamma_{k}(s_{k}, x_{\lambda}(s_{k})) - \int_{s_{k}}^{b} (b - \varrho)^{q-1} \mathcal{R}(b - \varrho) \\ f(\varrho, x_{\lambda}(\varrho), \mathcal{H}x_{\lambda}(\varrho)) d\varrho, \quad t \in \cup_{k=1}^{m} (s_{k}, t_{k+1}]. \end{cases}$$

Since, $\mathcal{P}(t)(t > 0)$ is compact, and g is uniformly bounded, we see that there exists a subsequence of $\{\mathcal{P}(b)g(x_{\lambda}) : \lambda > 0\}$, still denoted by it, converges to some $x_g \in \mathbb{X}$ as $\lambda \to 0$. Similarly there exists a subsequence of $\{\mathcal{P}(b-s_k)\gamma_k(s_k, x_{\lambda}(s_k)) : \lambda > 0\}$, still denoted by it, converges to some $x_{\gamma_k} \in \mathbb{X}$ as $\lambda \to 0$. Since f is uniformly bounded, we have

$$\int_0^b \|f(\varrho, x_\lambda(\varrho), \mathcal{H}x_\lambda(\varrho))\|^2 d\varrho \leqslant L_1^2 b.$$

Hence the sequence $f(\cdot, x_{\lambda}(\cdot), \mathcal{H}x_{\lambda}(\cdot))$ is bounded in $L^{2}(J, \mathbb{X})$. Then there exists a subsequence of $\{f(\cdot, x_{\lambda}(\cdot), \mathcal{H}x_{\lambda}(\cdot)) : \lambda > 0\}$, still denoted by it, converges weakly to some $f(\cdot) \in L^{2}(J, \mathbb{X})$. Define

$$\omega = \begin{cases} x^b - \mathcal{P}(b)(x_0) + x_g - \int_0^b (b-\varrho)^{q-1} \mathcal{R}(b-\varrho) f(\varrho) d\varrho, & t \in (0, t_1]; \\ x^b - x_{\gamma_k} - \int_{s_k}^b (b-\varrho)^{q-1} \mathcal{R}(b-\varrho) f(\varrho) d\varrho, & t \in \bigcup_{k=1}^m (s_k, t_{k+1}]. \end{cases}$$

It follows that for $t \in (0, t_1]$ and $t \in (s_k, t_{k+1}], k = 1, 2, \dots, m$,

$$||p(x_{\lambda}) - \omega|| \to 0 \quad \text{as} \quad \lambda \to 0^+,$$
 (5.3.23)

because of compactness of the operator (see [166])

$$l(\cdot) \to \int_0^{\cdot} (\cdot - s)^{q-1} \mathcal{R}(\cdot - \varrho) l(\varrho) d\varrho : L^2(J, \mathbb{X}) \to C(J, \mathbb{X}).$$

Then, from (5.3.22), (5.3.23), and (H0), we obtain

$$\begin{aligned} \|x_{\lambda}(b) - x^{b}\| &\leq \|\lambda R(\lambda, \Gamma_{0}^{b}) p(x_{\lambda})\| \\ &\leq \|\lambda R(\lambda, \Gamma_{0}^{b}) \omega\| + \|\lambda R(\lambda, \Gamma_{0}^{b})\| \|p(x_{\lambda}) - \omega\| \\ &\leq \|\lambda R(\lambda, \Gamma_{0}^{b}) \omega\| + \|p(x_{\lambda}) - \omega\| \to 0 \text{ as } \lambda \to 0^{+}. \end{aligned}$$

It concludes that the system (5.1.1) is approximately controllable on J.

5.4 Example

Consider a control system as following :

$$\begin{cases} {}^{C}\mathbf{D}^{\frac{1}{2}}v(t,y) = \frac{\partial^{2}}{\partial y^{2}}v(t,y) + u(t,y) + \frac{1}{25}\frac{e^{-t}}{1+e^{t}}\frac{|v(t,y)|}{1+|v(t,y)|} + \int_{0}^{t}\frac{1}{50}e^{-\varrho}\frac{|v(\varrho,y)|}{1+|v(\varrho,y)|}d\varrho, \\ y \in (0,1), t \in (0,\frac{1}{3}] \cup (\frac{2}{3},1], \\ v(t,0) = v(t,1) = 0, \quad t \in [0,1], \\ v(t,y) = \frac{e^{-(t-\frac{1}{3})}}{4}\frac{|v(t,y)|}{1+|v(t,y)|}, \quad y \in (0,1), t \in (\frac{1}{3},\frac{2}{3}], \\ v(0,y) + \sum_{i=1}^{2}\frac{1}{3^{i}}\frac{v(\frac{1}{i},y)}{1+v(\frac{1}{i},y)} = x_{0}(y), \quad y \in [0,1], \end{cases}$$
(5.4.1)

where $X = U = L^2[0, 1], J = [0, 1], x_0(y) \in X$. Define Av = v'' with

 $D(A) = \{ v \in \mathbb{X} : v, v' \text{ are absolutely continuous and } v'' \in \mathbb{X}, v(0) = v(1) = 0 \}.$

Then

$$Av = \sum_{n=1}^{\infty} -n^2 < v, \psi_n > \psi_n, \ v \in D(A),$$
(5.4.2)

where $\psi_n(y) = \sqrt{\frac{2}{\pi}} \sin(ny), 0 \leq y \leq 1, n = 1, 2, \dots$, and A is generator of a compact semigroup $\mathcal{S}(t)(t > 0)$ on X, and

$$\mathcal{S}(t)v = \sum_{n=1}^{\infty} e^{-n^2 t} < v, \psi_n > \psi_n, \ v \in \mathbb{X},$$
(5.4.3)

with $\|\mathcal{S}(t)\| \leq 1$, for any $t \geq 0$. Let $b = t_2 = 1, t_0 = s_0 = 0, t_1 = \frac{1}{3}, s_1 = \frac{2}{3}$. Put $v(t) = v(t, \cdot)$, means $v(t)(y) = v(t, y), t, y \in [0, 1]$. Let $u(t) = u(t, \cdot)$ is continuous, and $Bu(t) = u(t, \cdot)$. Further

$$\begin{split} f(t, v(t), \mathcal{H}v(t)) &= \frac{1}{25} \frac{e^{-t}}{1+e^t} \frac{|v(t, \cdot)|}{1+|v(t, \cdot)|} + \int_0^t \frac{1}{50} e^{-\varrho} \frac{|v(\varrho, \cdot)|}{1+|v(\varrho, \cdot)|} d\varrho, \\ \mathcal{H}v(t) &= \int_0^t \frac{1}{50} e^{-\varrho} \frac{|v(\varrho, \cdot)|}{1+|v(\varrho, \cdot)|} d\varrho, \\ \gamma_1(t, v(t)) &= \frac{e^{-(t-\frac{1}{3})}}{4} \frac{|v(t, \cdot)|}{1+|v(t, \cdot)|}, \\ g(v) &= \sum_{i=1}^2 \frac{1}{3^i} \frac{v(\frac{1}{i}, \cdot)}{1+v(\frac{1}{i}, \cdot)}. \end{split}$$

Then the system (5.4.1) can be rewritten into the abstract form of (5.1.1) for m = 1. Easily we can verify that the assumptions (H1)-(H5) as well as condition (5.3.1) hold with

$$q = \frac{1}{2}, \ M = 1, \ \phi(t) = \frac{1}{25} \frac{e^{-t}}{1+e^{t}} + \frac{1}{50}, \ \psi(r) = r,$$
$$\alpha = \frac{4}{9}, \ K_{\gamma_{1}} = \frac{1}{4}, \ \rho = \frac{4}{9} < 1.$$

Also f, g and γ_1 are uniformly bounded with $L_1 = \frac{3}{50}$, $L_2 = \frac{4}{9}$, $N_1 = \frac{1}{4}$ respectively. Moreover, (H0) also holds based on the argument in [132]. Thus by Theorem 5.3.2, the system (5.4.1) is approximately controllable.

Chapter 6

Fractional Evolution Equations with Deformable Fractional Derivative

0

6.1 Introduction

There are several generalization of the notions of derivative to fractional derivative so far, however only few of them become popular namely Riemann-Liouville, Caputo, Caputo-Fabrizio, Hilfer (Riemann-Liouville generalized), and Hadamard fractional derivatives. These definitions of fractional derivatives are based on integral form. Khalil [112], first introduced limit based definition of fractional derivative, and named it conformable fractional derivative, which looks a generalization of classical derivative and satisfies the classical properties that is linearity, product rule, quotient rule, Rolle's theorem, and mean value theorem. However this definition lacks to include zero and negative numbers. In [205], another limit based definition of fractional derivative is introduced by the authors and named as deformable fractional derivative, which is simpler than conformable fractional derivative and ranges

The contents of this chapter are accepted in **Journal of Nonlinear Evolution Equations and Applications** as Arshi Meraj, Dwijendra N Pandey: "Existence and uniqueness of mild solution and approximate controllability of fractional evolution equations with deformable fractional derivative".

over a wide class of functions, and also satisfies the classical properties.

Motivated by the fact that there is no work yet available on deformable fractional differential equations, in this chapter, first we discuss the existence and uniqueness of mild solution for the following fractional system

$$D^{\alpha}x(t) = Ax(t) + f(t, x(t)), \quad t \in J,$$

$$x(0) = x_0.$$
(6.1.1)

In recent years, controllability of various fractional differential equations, fractional integro-differential equations, fractional impulsive evolution equations has been studied where the fractional derivative is taken as either Caputo, Riemann-Liouville or Hilfer sense. However the controllability of fractional differential equations with local fractional derivative is still untreated topic. Motivated by this fact, next we establish the sufficient conditions for the approximate controllability of the following abstract deformable control system

$$D^{\alpha}x(t) = Ax(t) + f(t, x(t)) + Bu(t), \quad t \in J,$$

$$x(0) = x_0,$$
(6.1.2)

where D^{α} denotes the deformable fractional derivative with order $\alpha \in (0, 1)$ which is introduced by Zulfeqarr et. al. [205]. The operator A defined from $D(A) \subset \mathbb{X}$ to \mathbb{X} generates a C_0 semigroup $\mathcal{S}(t)(t \ge 0)$ on a Banach space \mathbb{X} , $x_0 \in \mathbb{X}$, J = [0, b], b > 0is a constant, $u \in L^2(J, U)$, $B : U \to \mathbb{X}$ is a bounded linear operator, U is a Hilbert space, and $f : J \times \mathbb{X} \to \mathbb{X}$ be a given function satisfying certain assumptions.

The remaining part of the chapter is organized as following: Section 6.2 contains some basic notations, definitions and theorems. Section 6.3 is further subdivided into two subsections. In first part, we will obtain the expression for mild solutions for the system (6.1.1) and discuss the sufficient conditions for the existence and uniqueness of mild solution. In second part, we will prove that the system (6.1.2) is approximately controllable on J. Finally, in section 6.4, we will present some examples to illustrate our results.

6.2 Preliminaries

Now, we recall definition of deformable fractional derivative, its fractional integral and some theorems, which are needed to prove our results. For more details regarding deformable fractional derivative, we refer [205].

Definition 6.2.1. The deformable fractional derivative of order α for a function $f:(a,b) \to \mathbb{R}$ is given as

$$D^{\alpha}f(t) = \lim_{\epsilon \to 0} \frac{(1 + \epsilon \gamma)f(t + \epsilon \alpha) - f(t)}{\epsilon},$$

where $\alpha + \gamma = 1, \ 0 \leq \alpha \leq 1$.

Remark 6.2.1. If $\alpha = 0$, $D^0 f(t) = f(t)$, and if $\alpha = 1$, Df(t) = f'(t).

Definition 6.2.2. The α -fractional integral of a continuous function f on [a, b] is defined as

$$I_a^{\alpha}f(t) = \frac{1}{\alpha}e^{\frac{-\gamma}{\alpha}t}\int_a^t e^{\frac{\gamma}{\alpha}s}f(s)ds, \quad where \ \alpha + \gamma = 1, \ \alpha \in (0,1].$$

Theorem 6.2.2. A differentiable function f at a point $t \in (a, b)$ is always α -differentiable at that point for any α . Moreover,

$$D^{\alpha}f(t) = \gamma f(t) + \alpha Df(t),$$

where $Df(t) = \frac{d}{dt}f(t)$.

Theorem 6.2.3. Let a function f is defined on (a, b), then f is α -differentiable for any $\alpha \in (0, 1]$ if and only if f is differentiable

Theorem 6.2.4. The operators D^{α} and I^{α} posses the following properties

(i)
$$D^{\alpha}(af + bg) = aD^{\alpha}f + bD^{\alpha}g$$
. (Linearity)

- (ii) $D^{\alpha_1}.D^{\alpha_2} = D^{\alpha_2}.D^{\alpha_1}.$ (Commutativity)
- (iii) $D^{\alpha}k = \gamma k$, where k is a constant.
- (iv) $D^{\alpha}(f.g) = (D^{\alpha}f).g + \alpha f.Dg.$
- (v) $I_a^{\alpha}(bf + cg) = bI_a^{\alpha}f + cI_a^{\alpha}g$. (Linearity)
- (vi) $I_a^{\alpha_1}I_a^{\alpha_2} = I_a^{\alpha_2}I_a^{\alpha_1}$. (Commutativity)

Theorem 6.2.5. (Inverse Property) If f is a function defined on [a, b] and is continuous also, then $I_a^{\alpha} f$ is α -differentiable in (a, b). In fact, we have

$$D^{\alpha}(I_a^{\alpha}f(t)) = f(t).$$

Conversely, suppose g is a continuous anti- α -derivative of f over (a,b), that is $g = D^{\alpha}f$, then

$$I_a^{\alpha}(D^{\alpha}f(t)) = I_a^{\alpha}(g(t)) = f(t) - e^{\frac{\gamma}{\alpha}(a-t)}f(a).$$

6.3 Main Results

6.3.1 Existence and uniqueness

Let $C(J, \mathbb{X})$ be the Banach space of all continuous maps from J to \mathbb{X} with supremum norm. Denote $M = \sup_{t \in J} \|\mathcal{S}(t)\|_{\mathbb{B}(\mathbb{X})}$, note that $M \ge 1$.

Lemma 6.3.1. Let A generates a C_0 -semigroup $S(t)(t \ge 0)$ on \mathbb{X} , then

$$D^{\alpha}\mathcal{S}(t)x = (\gamma I + \alpha A)\mathcal{S}(t)x, \ x \in D(A).$$

Proof. By [156], we know that $\mathcal{S}(t)x \in D(A)$ and $\frac{d}{dt}\mathcal{S}(t)x = A\mathcal{S}(t)x = \mathcal{S}(t)Ax$, for $x \in D(A)$. Now using Theorem 6.2.2, we get

$$D^{\alpha} \mathcal{S}(t) x = \gamma \mathcal{S}(t) x + \alpha \frac{d}{dt} \mathcal{S}(t) x$$

= $\gamma \mathcal{S}(t) x + \alpha A \mathcal{S}(t) x$
= $(\gamma I + \alpha A) \mathcal{S}(t) x.$

Consider the following linear deformable fractional Cauchy problem

$$D^{\alpha}x(t) = Ax(t), \quad t \in J,$$

 $x(0) = x_0.$ (6.3.1)

Definition 6.3.1. A function v is called solution to the problem (6.3.1), if the following hold

- (i) $v \in C(J, \mathbb{X})$, and $v(t) \in D(A)$ for all $t \in J$,
- (ii) $D^{\alpha}v$ exists and continuous on J,
- (iii) v satisfies (6.3.1).

Theorem 6.3.2. Let A be infinitesimal generator of a C_0 -semigroup $\mathcal{S}(t)(t \ge 0)$. If $x_0 \in D(A)$, then $e^{\frac{-\gamma}{\alpha}t} \mathcal{S}(\frac{t}{\alpha}) x_0$ is a solution to the problem (6.3.1).

Proof. Let $x(t) = e^{\frac{-\gamma}{\alpha}t} \mathcal{S}(\frac{t}{\alpha}) x_0$. Since $x_0 \in D(A)$, x(t) is differentiable. Now using Lemma 6.3.1, Theorem 6.2.2, and Theorem 6.2.4 we get

$$D^{\alpha}x(t) = \left[(D^{\alpha}e^{\frac{-\gamma}{\alpha}t})\mathcal{S}(\frac{t}{\alpha}) + \alpha e^{\frac{-\gamma}{\alpha}t}\frac{d}{dt}\mathcal{S}(\frac{t}{\alpha}) \right]x_{0}$$

$$= \left[(\gamma e^{\frac{-\gamma}{\alpha}t} + \alpha(\frac{-\gamma}{\alpha})e^{\frac{-\gamma}{\alpha}t})\mathcal{S}(\frac{t}{\alpha}) + e^{\frac{-\gamma}{\alpha}t}A\mathcal{S}(\frac{t}{\alpha}) \right]x_{0}$$

$$= Ae^{\frac{-\gamma}{\alpha}t}\mathcal{S}(\frac{t}{\alpha})x_{0}$$

$$= Ax(t).$$

Now, we consider the inhomogeneous initial value problem

$$D^{\alpha}x(t) = Ax(t) + f(t), \quad t \in J,$$

 $x(0) = x_0,$ (6.3.2)

where A generates a C_0 -semigroup, $x_0 \in \mathbb{X}$, and $f: J \to \mathbb{X}$ be a suitable function.

Theorem 6.3.3. Let x is a solution to the problem (6.3.2) and $f \in L^1(J, \mathbb{X})$, then x satisfies

$$x(t) = e^{\frac{-\gamma}{\alpha}t} \mathcal{S}(\frac{t}{\alpha}) x_0 + \frac{1}{\alpha} e^{\frac{-\gamma}{\alpha}t} \int_0^t e^{\frac{\gamma}{\alpha}\varrho} \mathcal{S}(\frac{t-\varrho}{\alpha}) f(\varrho) d\varrho.$$

Proof. Since x be a solution of the problem (6.3.2), so we have

$$D^{\alpha}x(t) = Ax(t) + f(t).$$
(6.3.3)

Also by Theorem 6.2.2, we know

$$D^{\alpha}x(t) = \gamma x(t) + \alpha x'(t). \tag{6.3.4}$$

From (6.3.3) and (6.3.4) easily we can conclude that

$$x'(t) = \frac{1}{\alpha} [(A - \gamma I)x(t) + f(t)].$$
(6.3.5)

Let $h(\varrho) = \mathcal{S}(\frac{t-\varrho}{\alpha})x(\varrho), \ 0 \leq \varrho \leq t$, since $x(\varrho) \in D(A)$ therefore h is differentiable and hence α -differentiable by Theorem 6.2.2. Now using Theorem 6.2.4, Lemma

6.3.1, and (6.3.5) we get

$$D^{\alpha}h(\varrho) = \left(D^{\alpha}S(\frac{t-\varrho}{\alpha})\right)x(\varrho) + \alpha S(\frac{t-\varrho}{\alpha})x'(\varrho)$$

$$= \left[\gamma S(\frac{t-\varrho}{\alpha}) + \alpha \frac{d}{d\varrho}S(\frac{t-\varrho}{\alpha})\right]x(\varrho) + S(\frac{t-\varrho}{\alpha})[(A-\gamma I)x(\varrho) + f(\varrho)]$$

$$= \left[\gamma S(\frac{t-\varrho}{\alpha}) - AS(\frac{t-\varrho}{\alpha})\right]x(\varrho) + AS(\frac{t-\varrho}{\alpha})x(\varrho) - \gamma S(\frac{t-\varrho}{\alpha})x(\varrho)$$

$$+ S(\frac{t-\varrho}{\alpha})f(\varrho)$$

$$= S(\frac{t-\varrho}{\alpha})f(\varrho).$$
(6.3.6)

By Theorem 6.2.5, we obtain

$$I^{\alpha}(D^{\alpha}h(t)) = h(t) - e^{\frac{-\gamma}{\alpha}t}h(0)$$

= $x(t) - e^{\frac{-\gamma}{\alpha}t}\mathcal{S}(\frac{t}{\alpha})x_0.$ (6.3.7)

Since $f \in L^1(J, \mathbb{X})$, so $\mathcal{S}(\frac{t-\varrho}{\alpha})f(\varrho)$ is integrable. Integrating (6.3.6)

$$I^{\alpha}(D^{\alpha}h(t)) = \frac{1}{\alpha} e^{\frac{-\gamma}{\alpha}t} \int_{0}^{t} e^{\frac{\gamma}{\alpha}\varrho} \mathcal{S}(\frac{t-\varrho}{\alpha}) f(\varrho) d\varrho.$$
(6.3.8)

From (6.3.7) and (6.3.8), we obtain

$$x(t) = e^{\frac{-\gamma}{\alpha}t} \mathcal{S}(\frac{t}{\alpha}) x_0 + \frac{1}{\alpha} e^{\frac{-\gamma}{\alpha}t} \int_0^t e^{\frac{\gamma}{\alpha}\varrho} \mathcal{S}(\frac{t-\varrho}{\alpha}) f(\varrho) d\varrho.$$

Theorem 6.3.4. Let A generates a C_0 -semigroup $\mathcal{S}(t)$ and $f \in C(J, \mathbb{X})$. If $f(\varrho) \in D(A)$ for $0 < \varrho < t$ and $Af(\varrho) \in L^1(J, \mathbb{X})$, then for every $x_0 \in D(A)$ the function $x: J \to \mathbb{X}$ defined by

$$x(t) = e^{\frac{-\gamma}{\alpha}t} \mathcal{S}(\frac{t}{\alpha}) x_0 + \frac{1}{\alpha} e^{\frac{-\gamma}{\alpha}t} \int_0^t e^{\frac{\gamma}{\alpha}\varrho} \mathcal{S}(\frac{t-\varrho}{\alpha}) f(\varrho) d\varrho, \qquad (6.3.9)$$

is a solution to the problem (6.3.2).

Proof. Let $u(t) = e^{\frac{-\gamma}{\alpha}t} \mathcal{S}(\frac{t}{\alpha}) x_0$ and $v(t) = \frac{1}{\alpha} e^{\frac{-\gamma}{\alpha}t} \int_0^t e^{\frac{\gamma}{\alpha}\varrho} \mathcal{S}(\frac{t-\varrho}{\alpha}) f(\varrho) d\varrho$, therefore x(t) given in (6.3.9) is rewritten as x(t) = u(t) + v(t). Since $x_0 \in D(A)$, u(t) is differentiable and by Theorem 6.3.2, we know $D^{\alpha}u(t) = Au(t)$. From the assumptions it is

easy to conclude that v(t) is differentiable, and

$$\begin{aligned} v'(t) &= \frac{1}{\alpha} \left(\frac{-\gamma}{\alpha}\right) e^{\frac{-\gamma}{\alpha}t} \int_{0}^{t} e^{\frac{\gamma}{\alpha}\varrho} \mathcal{S}\left(\frac{t-\varrho}{\alpha}\right) f(\varrho) d\varrho + \frac{1}{\alpha} e^{\frac{-\gamma}{\alpha}t} \frac{d}{dt} \int_{0}^{t} e^{\frac{\gamma}{\alpha}\varrho} \mathcal{S}\left(\frac{t-\varrho}{\alpha}\right) f(\varrho) d\varrho \\ &= \frac{-\gamma}{\alpha^{2}} e^{\frac{-\gamma}{\alpha}t} \int_{0}^{t} e^{\frac{\gamma}{\alpha}\varrho} \mathcal{S}\left(\frac{t-\varrho}{\alpha}\right) f(\varrho) d\varrho + \frac{1}{\alpha} e^{\frac{-\gamma}{\alpha}t} \left[\int_{0}^{t} e^{\frac{\gamma}{\alpha}\varrho} \frac{\partial}{\partial t} \mathcal{S}\left(\frac{t-\varrho}{\alpha}\right) f(\varrho) d\varrho \\ &+ e^{\frac{\gamma}{\alpha}t} f(t) \right] \\ &= \frac{-\gamma}{\alpha^{2}} e^{\frac{-\gamma}{\alpha}t} \int_{0}^{t} e^{\frac{\gamma}{\alpha}\varrho} \mathcal{S}\left(\frac{t-\varrho}{\alpha}\right) f(\varrho) d\varrho + \frac{1}{\alpha} e^{\frac{-\gamma}{\alpha}t} \left[\int_{0}^{t} e^{\frac{\gamma}{\alpha}\varrho} \frac{1}{\alpha} A \mathcal{S}\left(\frac{t-\varrho}{\alpha}\right) f(\varrho) d\varrho \\ &+ e^{\frac{\gamma}{\alpha}t} f(t) \right] \\ &= \frac{-\gamma}{\alpha^{2}} e^{\frac{-\gamma}{\alpha}t} \int_{0}^{t} e^{\frac{\gamma}{\alpha}\varrho} \mathcal{S}\left(\frac{t-\varrho}{\alpha}\right) f(\varrho) d\varrho + \frac{1}{\alpha^{2}} A e^{\frac{-\gamma}{\alpha}t} \int_{0}^{t} e^{\frac{\gamma}{\alpha}\varrho} \mathcal{S}\left(\frac{t-\varrho}{\alpha}\right) f(\varrho) d\varrho + \frac{1}{\alpha} f(t) \\ &= \frac{1}{\alpha} \left(-\gamma v(t) + A v(t) + f(t) \right). \end{aligned}$$

$$(6.3.10)$$

Now, by Theorem 6.2.2 and (6.3.10) we get

$$D^{\alpha}v(t) = \gamma v(t) + \alpha v'(t)$$

= $\gamma v(t) + \left(-\gamma v(t) + Av(t) + f(t)\right)$
= $Av(t) + f(t).$ (6.3.11)

So,

$$D^{\alpha}x(t) = D^{\alpha}u(t) + D^{\alpha}v(t) = Au(t) + Av(t) + f(t) = Ax(t) + f(t),$$
(6.3.12)

also, $x(0) = u(0) + v(0) = x_0$. Thus x(t) given by (6.3.9) is a solution to inhomogeneous problem (6.3.2).

Now, we will study the semilinear initial value problem (6.1.1).

Definition 6.3.2. A continuous solution of the following integral equation

$$x(t) = e^{\frac{-\gamma}{\alpha}t} \mathcal{S}(\frac{t}{\alpha}) x_0 + \frac{1}{\alpha} e^{\frac{-\gamma}{\alpha}t} \int_0^t e^{\frac{\gamma}{\alpha}\varrho} \mathcal{S}(\frac{t-\varrho}{\alpha}) f(\varrho, x(\varrho)) d\varrho, \qquad (6.3.13)$$

is named as mild solution to the problem (6.1.1).

Now, we state the basic assumptions required to prove the main result :

(Hf) The function f from $J \times \mathbb{X}$ to \mathbb{X} is continuous and satisfies

$$\|f(t,v) - f(t,z)\| \leq L \|v - z\|, \quad \forall v, z \in \mathbb{X},$$

for some positive constant L.

Theorem 6.3.5. Let the hypotheses (HA) and (Hf) are satisfied, then for every $x_0 \in \mathbb{X}$, the system (6.1.1) has a unique mild solution, provided that $\frac{ML}{\gamma} < 1$.

Proof. Let us define a map $F: C(J, \mathbb{X}) \to C(J, \mathbb{X})$ as

$$(Fx)(t) = e^{\frac{-\gamma}{\alpha}t} \mathcal{S}(\frac{t}{\alpha}) x_0 + \frac{1}{\alpha} e^{\frac{-\gamma}{\alpha}t} \int_0^t e^{\frac{\gamma}{\alpha}\varrho} \mathcal{S}(\frac{t-\varrho}{\alpha}) f(\varrho, x(\varrho)) d\varrho, \quad t \in J.$$

Let $z, v \in C(J, \mathbb{X})$, it follows readily by the definition of F

$$\begin{split} \|(Fz)(t) - (Fv)(t)\| &\leqslant \frac{1}{\alpha} e^{\frac{-\gamma}{\alpha}t} \int_0^t e^{\frac{\gamma}{\alpha}\varrho} \|\mathcal{S}(\frac{t-\varrho}{\alpha})\| \|f(\varrho, z(\varrho)) - f(\varrho, v(\varrho))\| d\varrho \\ &\leqslant \frac{ML}{\alpha} e^{\frac{-\gamma}{\alpha}t} \left(\int_0^t e^{\frac{\gamma}{\alpha}\varrho} d\varrho \right) \|z - v\| \\ &= \frac{ML}{\gamma} e^{\frac{-\gamma}{\alpha}t} \left[e^{\frac{\gamma}{\alpha}t} - 1 \right] \|z - v\| \\ &= \frac{ML}{\gamma} \left[1 - e^{\frac{-\gamma}{\alpha}t} \right] \|z - v\| \\ &\leqslant \frac{ML}{\gamma} \|z - v\|. \end{split}$$

Hence, F is contraction map. Banach contraction fixed point theorem implies that F has a unique fixed point $x \in C(J, \mathbb{X})$, hence a unique mild solution to the system (6.1.1).

6.3.2 Approximate controllability

Throughout this subsection, we consider X is a Hilbert space.

Definition 6.3.3. A function $v \in C(J, \mathbb{X})$ is known as a mild solution of (6.1.2), if for any $u \in L^2(J, \mathbb{X})$, the following holds

$$v(t) = e^{\frac{-\gamma}{\alpha}t} \mathcal{S}(\frac{t}{\alpha}) x_0 + \frac{1}{\alpha} e^{\frac{-\gamma}{\alpha}t} \int_0^t e^{\frac{\gamma}{\alpha}\varrho} \mathcal{S}(\frac{t-\varrho}{\alpha}) [f(\varrho, v(\varrho)) + Bu(\varrho)] d\varrho, \quad t \in J.$$

Let $x^b(x_0, u)$ be the state value of (6.1.2) at time b with respect to the initial value x_0 and control u. Let $\mathbf{R}(b, x_0) = \{x^b(x_0, u) : u \in L^2(J, U)\}$, which is named as reachable set to the system (6.1.2) at time b.

Definition 6.3.4. ([132]) If $\overline{\mathbf{R}(b, x_0)} = \mathbb{X}$, the system (6.1.2) is called approximately controllable on J.

Now, Consider the linear control system corresponding to (6.1.2)

$$D^{\alpha}x(t) = Ax(t) + Bu(t), \quad t \in J,$$

 $x(0) = x_0.$ (6.3.14)

Definition 6.3.5. (a) A controllability map for the system (6.3.14) on J is a linear bounded map $\mathfrak{B}^b : L^2(J, U) \to \mathbb{X}$, defined as

$$\mathfrak{B}^{b}u := \frac{1}{\alpha} e^{\frac{-\gamma}{\alpha}b} \int_{0}^{b} e^{\frac{\gamma}{\alpha}\varrho} \mathcal{S}(\frac{b-\varrho}{\alpha}) Bu(\varrho) d\varrho.$$
(6.3.15)

(b) The system (6.3.14) is called approximately controllable on J, if the range space of \mathfrak{B}^b is dense in \mathbb{X} i.e.

$$\overline{ran\mathfrak{B}^b} = \mathbb{X}$$

(c) The controllability gramian of (6.3.14) on J is defined by

$$\Gamma_0^b := \mathfrak{B}^b(\mathfrak{B}^b)^*. \tag{6.3.16}$$

Lemma 6.3.6. The controllability map and controllability gramian satisfy the following :

- (a) $(\mathfrak{B}^b)^* x(\varrho) = B^* \mathcal{S}^*(\frac{b-\varrho}{\alpha}) x$, for $\varrho \in J$, $x \in \mathbb{X}$.
- (b) $\Gamma_0^b \in \mathbb{B}(\mathbb{X})$, is symmetric, and has the representation

$$\Gamma_0^b = \frac{1}{\alpha} e^{\frac{-\gamma}{\alpha}b} \int_0^b e^{\frac{\gamma}{\alpha}\varrho} \mathcal{S}(\frac{b-\varrho}{\alpha}) BB^* \mathcal{S}^*(\frac{b-\varrho}{\alpha}) d\varrho, \qquad (6.3.17)$$

and $\Gamma_0^b \ge 0$, where $(\cdot)^*$ denotes the adjoint operator of (\cdot) .

Proof. (a) : The way of proof is based on [62] (Lemma 4.1.4, page 144). For $x \in \mathbb{X}$ and $u \in L^2(J, U)$

$$\begin{split} \langle u, (\mathfrak{B}^{b})^{*}x \rangle &= \langle \mathfrak{B}^{b}u, x \rangle \\ &= \langle \frac{1}{\alpha} e^{\frac{-\gamma}{\alpha}b} \int_{0}^{b} e^{\frac{\gamma}{\alpha}\varrho} \mathcal{S}(\frac{b-\varrho}{\alpha}) Bu(\varrho) d\varrho, x \rangle \\ &= \frac{1}{\alpha} e^{\frac{-\gamma}{\alpha}b} \int_{0}^{b} \langle e^{\frac{\gamma}{\alpha}\varrho} \mathcal{S}(\frac{b-\varrho}{\alpha}) Bu(\varrho), x \rangle d\varrho \\ &= \frac{1}{\alpha} e^{\frac{-\gamma}{\alpha}b} \int_{0}^{b} e^{\frac{\gamma}{\alpha}\varrho} \langle u(\varrho), B^{*} \mathcal{S}^{*}(\frac{b-\varrho}{\alpha}) x \rangle d\varrho, \end{split}$$

and this proves (\mathbf{a}) .

(b) : From (6.3.16), it is easy to see Γ_0^b is symmetric, and $\Gamma_0^b \ge 0$. Equation (6.3.17) follows easily by (6.3.15), (6.3.16) and (a).

Theorem 6.3.7. ([62]) The system (6.3.14) is approximately controllable on J iff any one of the following hold :

- (i) Γ_0^b is positive operator, that means $\langle \Gamma_0^b x, x \rangle > 0$, for all $0 \neq x \in \mathbb{X}$.
- (ii) $ker(\mathfrak{B}^b)^* = \{0\}.$
- (iii) $B^* \mathcal{S}^*(\frac{b-t}{\alpha}) x = 0 \text{ on } J \Longrightarrow x = 0.$

Theorem 6.3.8. ([132]) Let \mathbb{Z} be a reflexive separable complete norm space and \mathbb{Z}^* denotes the dual space of \mathbb{Z} , $\Gamma : \mathbb{Z}^* \to \mathbb{Z}$ is a symmetric map, then the statements given below are equivalent :

- (i) $\Gamma : \mathbb{Z}^* \to \mathbb{Z}$ is positive.
- (ii) $\lambda(\lambda I + \Gamma \mathfrak{J})^{-1}(v)$ converges to zero strongly as $\lambda \to 0^+$, $\forall v \in \mathbb{Z}$. Here \mathfrak{J} is the duality map from $\mathbb{Z} \to \mathbb{Z}^*$.

Lemma 6.3.9. The linear control system (6.3.14) is approximately controllable on J iff $\lambda R(\lambda, \Gamma_0^b) \to 0$ as $\lambda \to 0^+$ in strong operator topology, where $R(\lambda, \Gamma_0^b) = (\lambda I + \Gamma_0^b)^{-1}$.

Proof. The proof is straightforward from Theorem 6.3.7 and Theorem 6.3.8.

To investigate the approximate controllability of (6.1.2), we impose the following assumptions :

- (H1) A generates a C_0 -semigroup $\mathcal{S}(t)(t \ge 0)$ on X, and $\mathcal{S}(t)(t > 0)$ is compact.
- (H2) The function f is continuous on X for each $t \in J$, and is Lebesgue measurable on J for all $x \in X$.
- (H3) There exists a function $\phi \in L^{\frac{1}{\alpha_1}}(J, \mathbb{R}^+)$ where $\alpha_1 \in (0, \alpha)$ such that

$$||f(t,z)|| \leq \phi(t), \quad \forall z \in \mathbb{X}, \ t \in J.$$

(H4) The linear control system (6.3.14) is approximately controllable on J.

We will use the following notations :

$$B_r = \{ v \in C(J, \mathbb{X}) : \|v\| \leqslant r \}, \text{ for each finite constant } r > 0,$$

$$M_B = \|B\|, \ M_\phi = \|\phi\|_{L^{\frac{1}{\alpha_1}}(J, \mathbb{R}^+)}, \ q = \frac{1}{1 - \alpha_1}, \ N = \frac{1}{\alpha} \left(\frac{\alpha}{q\gamma}\right)^{\frac{1}{q}} M_\phi.$$

For an arbitrary function $v \in C(J, \mathbb{X})$, we consider the control function for the nonlinear system (6.1.2) as :

$$u(t) = u_{\lambda}(t, v) = B^* \mathcal{S}^*(\frac{b-t}{\alpha}) R(\lambda, \Gamma_0^b) p(v), \qquad (6.3.18)$$

where

$$p(v) = x^{b} - e^{\frac{-\gamma}{\alpha}b} \mathcal{S}(\frac{b}{\alpha}) x_{0} - \frac{1}{\alpha} e^{\frac{-\gamma}{\alpha}b} \int_{0}^{b} e^{\frac{\gamma}{\alpha}\varrho} \mathcal{S}(\frac{b-\varrho}{\alpha}) f(\varrho, v(\varrho)) d\varrho.$$
(6.3.19)

For any $\lambda > 0$, define $F_{\lambda} : C(J, \mathbb{X}) \to C(J, \mathbb{X})$ as following :

$$(F_{\lambda}v)(t) = e^{\frac{-\gamma}{\alpha}t} \mathcal{S}(\frac{t}{\alpha}) x_0 + \frac{1}{\alpha} e^{\frac{-\gamma}{\alpha}t} \int_0^t e^{\frac{\gamma}{\alpha}\varrho} \mathcal{S}(\frac{t-\varrho}{\alpha}) [f(\varrho, v(\varrho)) + Bu_{\lambda}(\varrho, v)] d\varrho. \quad (6.3.20)$$

Lemma 6.3.10. If the assumptions (H1)-(H3) hold, then for any $t \in J$ we have (i) $\frac{1}{\alpha}e^{\frac{-\gamma}{\alpha}t}\int_{0}^{t}e^{\frac{\gamma}{\alpha}\varrho}\|\mathcal{S}(\frac{t-\varrho}{\alpha})f(\varrho,v(\varrho))\|d\varrho \leq MN.$

(ii)
$$\|u_{\lambda}(t,v)\| \leq \frac{MM_B}{\lambda} \left[\|x^b\| + M(\|x_0\| + N) \right].$$

Proof. (i): By (H3) and Hölder inequality, we get

$$\begin{aligned} \frac{1}{\alpha} e^{\frac{-\gamma}{\alpha}t} \int_{0}^{t} e^{\frac{\gamma}{\alpha}\varrho} \|\mathcal{S}(\frac{t-\varrho}{\alpha})f(\varrho, v(\varrho))\| d\varrho &\leqslant \frac{M}{\alpha} e^{\frac{-\gamma}{\alpha}t} \int_{0}^{t} e^{\frac{\gamma}{\alpha}\varrho} \phi(\varrho) d\varrho \\ &\leqslant \frac{M}{\alpha} e^{\frac{-\gamma}{\alpha}t} \left(\int_{0}^{t} e^{\frac{q\gamma\varrho}{\alpha}} d\varrho\right)^{\frac{1}{q}} \|\phi\|_{L^{\frac{1}{\alpha_{1}}}(J,\mathbb{R}^{+})} \\ &\leqslant \frac{1}{\alpha} \left(\frac{\alpha}{q\gamma}\right)^{\frac{1}{q}} e^{\frac{-\gamma}{\alpha}t} ([e^{\frac{q\gamma t}{\alpha}} - 1])^{\frac{1}{q}} M M_{\phi} \\ &\leqslant \frac{1}{\alpha} \left(\frac{\alpha}{q\gamma}\right)^{\frac{1}{q}} e^{\frac{-\gamma}{\alpha}t} e^{\frac{\gamma}{\alpha}t} M M_{\phi} \\ &\leqslant \frac{1}{\alpha} \left(\frac{\alpha}{q\gamma}\right)^{\frac{1}{q}} M M_{\phi} = M N. \end{aligned}$$

(ii): Using (6.3.18), (6.3.19), and (i) we obtain

$$\begin{aligned} \|u_{\lambda}(t,v)\| &\leq \|B^{*}\mathcal{S}^{*}(\frac{b-t}{\alpha})R(\lambda,\Gamma_{0}^{b})p(v)\| \\ &\leq \frac{MM_{B}}{\lambda}\|p(v)\| \\ &\leq \frac{MM_{B}}{\lambda}\Big[\|x^{b}\|+M\|x_{0}\|+\frac{1}{\alpha}e^{\frac{-\gamma}{\alpha}b}\int_{0}^{b}e^{\frac{\gamma}{\alpha}\varrho}\|\mathcal{S}(\frac{b-\varrho}{\alpha})f(\varrho,v(\varrho))\|d\varrho\Big] \\ &\leq \frac{MM_{B}}{\lambda}\Big[\|x^{b}\|+M(\|x_{0}\|+N)\Big]. \end{aligned}$$

Theorem 6.3.11. If the hypotheses (H1)-(H3) hold, then the fractional semilinear control system (6.1.2) has a mild solution.

Proof. For convenience, we divide the proof into several steps :

Step I: For given $\lambda > 0$, we want to show that there exists a constant $R = R(\lambda) > 0$, satisfying $F_{\lambda}(B_R) \subset B_R$. Let $v \in B_r$, for any positive constant r. If $t \in J$, then by using Lemma 6.3.10, we have

$$\begin{split} \|(F_{\lambda}v)(t)\| &\leqslant e^{\frac{-\gamma}{\alpha}t} \|\mathcal{S}(\frac{t}{\alpha})x_{0}\| + \frac{1}{\alpha}e^{\frac{-\gamma}{\alpha}t} \int_{0}^{t} e^{\frac{\gamma}{\alpha}\varrho} \|\mathcal{S}(\frac{t-\varrho}{\alpha})[f(\varrho, v(\varrho)) + Bu_{\lambda}(\varrho, v)]\|d\varrho\\ &\leqslant M\|x_{0}\| + \frac{1}{\alpha}e^{\frac{-\gamma}{\alpha}t} \int_{0}^{t} e^{\frac{\gamma}{\alpha}\varrho} \|\mathcal{S}(\frac{t-\varrho}{\alpha})f(\varrho, v(\varrho))\|d\varrho\\ &+ \frac{1}{\alpha}e^{\frac{-\gamma}{\alpha}t} \int_{0}^{t} e^{\frac{\gamma}{\alpha}\varrho} \|\mathcal{S}(\frac{t-\varrho}{\alpha})Bu_{\lambda}(\varrho, v)\|d\varrho\\ &\leqslant M\|x_{0}\| + MN + \frac{1}{\gamma}MM_{B}e^{\frac{-\gamma}{\alpha}t}[e^{\frac{\gamma}{\alpha}t} - 1]\|u_{\lambda}\|\\ &\leqslant M(\|x_{0}\| + N) + \frac{M^{2}(M_{B})^{2}}{\lambda\gamma} \Big[\|x^{b}\| + M(\|x_{0}\| + N)\Big]. \end{split}$$

This implies that for large enough R > 0, $F_{\lambda}(B_R) \subset B_R$ holds.

Step II: For any $t \in J$ the set $\{(F_{\lambda}v)(t) : v \in B_R\}$ is relatively compact in X. If t = 0, clearly $\{(F_{\lambda}v)(0) : v \in B_R\} = \{x_0\}$, which is compact in X. Let $0 < t \leq b$ and $0 < \varepsilon < t$. For $v \in B_R$, define

$$\begin{aligned} (F_{\lambda}^{\varepsilon}v)(t) &= e^{\frac{-\gamma}{\alpha}t}\mathcal{S}(\frac{t}{\alpha})x_{0} + \frac{1}{\alpha}e^{\frac{-\gamma}{\alpha}t}\int_{0}^{t-\varepsilon}e^{\frac{\gamma}{\alpha}\varrho}\mathcal{S}(\frac{t-\varrho}{\alpha})[f(\varrho,v(\varrho)) + Bu_{\lambda}(\varrho,v)]d\varrho \\ &= e^{\frac{-\gamma}{\alpha}t}\mathcal{S}(\frac{t}{\alpha})x_{0} + \frac{1}{\alpha}e^{\frac{-\gamma}{\alpha}t}\mathcal{S}(\frac{\varepsilon}{\alpha})\int_{0}^{t-\varepsilon}e^{\frac{\gamma}{\alpha}\varrho}\mathcal{S}(\frac{t-\varrho-\varepsilon}{\alpha})[f(\varrho,v(\varrho)) \\ &+Bu_{\lambda}(\varrho,v)]d\varrho \\ &= e^{\frac{-\gamma}{\alpha}t}\mathcal{S}(\frac{t}{\alpha})x_{0} + \frac{1}{\alpha}e^{\frac{-\gamma}{\alpha}t}\mathcal{S}(\frac{\varepsilon}{\alpha})y(t,\varepsilon), \end{aligned}$$

since $y(t,\varepsilon)$ is bounded on B_R and $\mathcal{S}(t)$ is compact for (t > 0), we obtain that the set $\{(F_{\lambda}^{\varepsilon}v)(t) : v \in B_R\}$ is relatively compact in X. Also, observe that

$$\begin{aligned} \|(F_{\lambda}v)(t) - (F_{\lambda}^{\varepsilon}v)(t)\| &= \|\frac{1}{\alpha}e^{\frac{-\gamma}{\alpha}t}\int_{t-\varepsilon}^{t}e^{\frac{\gamma}{\alpha}\varrho}\mathcal{S}(\frac{t-\varrho}{\alpha})[f(\varrho,v(\varrho)) + Bu_{\lambda}(\varrho,v)]d\varrho\| \\ &\leqslant I_{1} + I_{2}, \end{aligned}$$
(6.3.21)

where

$$I_{1} = \|\frac{1}{\alpha}e^{\frac{-\gamma}{\alpha}t}\int_{t-\varepsilon}^{t}e^{\frac{\gamma}{\alpha}\varrho}\mathcal{S}(\frac{t-\varrho}{\alpha})f(\varrho,v(\varrho))d\varrho\|,$$

$$I_{2} = \|\frac{1}{\alpha}e^{\frac{-\gamma}{\alpha}t}\int_{t-\varepsilon}^{t}e^{\frac{\gamma}{\alpha}\varrho}\mathcal{S}(\frac{t-\varrho}{\alpha})Bu_{\lambda}(\varrho,v)d\varrho\|.$$

Now, proceeding in the same way as Lemma 6.3.10 yields

$$I_1 \leqslant MN [e^{\frac{\gamma q}{\alpha}t} - e^{\frac{\gamma q}{\alpha}(t-\varepsilon)}]^{\frac{1}{q}}, \qquad (6.3.22)$$

$$I_2 \leqslant \frac{M^2 (M_B)^2}{\lambda \gamma} \bigg[\|x^b\| + M(\|x_0\| + N) \bigg] [e^{\frac{\gamma}{\alpha}t} - e^{\frac{\gamma}{\alpha}(t-\varepsilon)}].$$
(6.3.23)

Therefore, by (6.3.21), (6.3.22) and (6.3.23) we conclude that

$$||(F_{\lambda}v)(t) - (F_{\lambda}^{\varepsilon}v)(t)|| \to 0 \text{ as } \varepsilon \to 0.$$

This implies that the set the set $\{(F_{\lambda}v)(t) : v \in B_R\}, t \in (0, b]$ is relatively compact in X.

Step III: The family of functions $\{F_{\lambda}v : v \in B_R\}$ is equicontinuous on J. For any $0 \leq t_1 < t_2 \leq b$, and $v \in B_R$

$$\begin{split} \|(F_{\lambda}v)(t_{2}) - (F_{\lambda}v)(t_{1})\| &\leqslant \left\| e^{\frac{-\gamma}{\alpha}t_{1}} [\mathcal{S}(\frac{t_{2}}{\alpha}) - \mathcal{S}(\frac{t_{1}}{\alpha})]x_{0} \right\| \\ &+ \left\| \frac{1}{\alpha} e^{\frac{-\gamma}{\alpha}t_{2}} \int_{t_{1}}^{t_{2}} e^{\frac{\gamma}{\alpha}\varrho} \mathcal{S}(\frac{t_{2}-\varrho}{\alpha})f(\varrho, v(\varrho))d\varrho \right\| \\ &+ \left\| \frac{1}{\alpha} e^{\frac{-\gamma}{\alpha}t_{2}} \int_{t_{1}}^{t_{2}} e^{\frac{\gamma}{\alpha}\varrho} \mathcal{S}(\frac{t_{2}-\varrho}{\alpha})Bu_{\lambda}(\varrho, v)d\varrho \right\| \\ &+ \left\| \frac{1}{\alpha} e^{\frac{-\gamma}{\alpha}t_{1}} \int_{0}^{t_{1}} e^{\frac{\gamma}{\alpha}\varrho} [\mathcal{S}(\frac{t_{2}-\varrho}{\alpha}) - \mathcal{S}(\frac{t_{1}-\varrho}{\alpha})]f(\varrho, v(\varrho))d\varrho \right\| \\ &+ \left\| \frac{1}{\alpha} e^{\frac{-\gamma}{\alpha}t_{1}} \int_{0}^{t_{1}} e^{\frac{\gamma}{\alpha}\varrho} [\mathcal{S}(\frac{t_{2}-\varrho}{\alpha}) - \mathcal{S}(\frac{t_{1}-\varrho}{\alpha})]Bu_{\lambda}(\varrho, v)d\varrho \right\| \\ &= J_{1} + J_{2} + J_{3} + J_{4} + J_{5}. \end{split}$$

Now, using Lemma 6.3.10 we get

$$J_{1} \leqslant \left\| \mathcal{S}(\frac{t_{2}}{\alpha}) - \mathcal{S}(\frac{t_{1}}{\alpha}) \right\| \|x_{0}\|,$$

$$J_{2} \leqslant \frac{MM_{\phi}}{\alpha} \left(\frac{\alpha}{\gamma q}\right)^{\frac{1}{q}} [e^{\frac{\gamma q}{\alpha}t_{2}} - e^{\frac{\gamma q}{\alpha}t_{1}}]^{\frac{1}{q}},$$

$$J_{3} \leqslant \frac{M^{2}(M_{B})^{2}}{\lambda \gamma} \left[\|x^{b}\| + M(\|x_{0}\| + N) \right] [e^{\frac{\gamma}{\alpha}t_{2}} - e^{\frac{\gamma}{\alpha}t_{1}}].$$

For $t_1 = 0$, it is easy to see that $J_4 = 0$. For $t_1 > 0$ and $\epsilon > 0$ (small enough), we obtain

$$J_{4} \leqslant \left\| \frac{1}{\alpha} e^{\frac{-\gamma}{\alpha}t_{1}} \int_{0}^{t_{1}-\epsilon} e^{\frac{\gamma}{\alpha}\varrho} [\mathcal{S}(\frac{t_{2}-\varrho}{\alpha}) - \mathcal{S}(\frac{t_{1}-\varrho}{\alpha})] f(\varrho, v(\varrho)) d\varrho \right\| \\ + \left\| \frac{1}{\alpha} e^{\frac{-\gamma}{\alpha}t_{1}} \int_{t_{1}-\epsilon}^{t_{1}} e^{\frac{\gamma}{\alpha}\varrho} [\mathcal{S}(\frac{t_{2}-\varrho}{\alpha}) - \mathcal{S}(\frac{t_{1}-\varrho}{\alpha})] f(\varrho, v(\varrho)) d\varrho \right\| \\ \leqslant \left\| \frac{1}{\alpha} e^{\frac{-\gamma}{\alpha}t_{1}} \int_{0}^{t_{1}-\epsilon} e^{\frac{\gamma}{\alpha}\varrho} \phi(\varrho) d\varrho \sup_{\varrho \in [0,t_{1}-\epsilon]} \left\| \mathcal{S}(\frac{t_{2}-\varrho}{\alpha}) - \mathcal{S}(\frac{t_{1}-\varrho}{\alpha}) \right\| \\ + \frac{2M}{\alpha} e^{\frac{-\gamma}{\alpha}t_{1}} \int_{t_{1}-\epsilon}^{t_{1}} e^{\frac{\gamma}{\alpha}\varrho} \phi(\varrho) d\varrho \\ \leqslant \left\| N e^{\frac{-\gamma}{\alpha}\epsilon} \sup_{\varrho \in [0,t_{1}-\epsilon]} \left\| \mathcal{S}(\frac{t_{2}-\varrho}{\alpha}) - \mathcal{S}(\frac{t_{1}-\varrho}{\alpha}) \right\| + 2MN [e^{\frac{\gamma q}{\alpha}t_{1}} - e^{\frac{\gamma q}{\alpha}(t_{1}-\epsilon)}]^{\frac{1}{q}}, \end{aligned}$$

similarly

$$J_{5} \leqslant \frac{M_{B}}{\gamma} \left[e^{\frac{-\gamma}{\alpha}\epsilon} - e^{\frac{-\gamma}{\alpha}t_{1}} \right] \|u_{\lambda}\| \sup_{\varrho \in [0,t_{1}-\epsilon]} \left\| \mathcal{S}(\frac{t_{2}-\varrho}{\alpha}) - \mathcal{S}(\frac{t_{1}-\varrho}{\alpha}) \right\| \\ + \frac{2MM_{B}}{\gamma} \left[e^{\frac{\gamma}{\alpha}t_{1}} - e^{\frac{\gamma}{\alpha}(t_{1}-\epsilon)} \right] \|u_{\lambda}\|.$$

Using (H1), it is clear that $J_i \to 0$ (i = 1, 2, 3, 4, 5) as $t_2 \to t_1$, $\epsilon \to 0$. As a result $||(F_{\lambda}v)(t_2) - (F_{\lambda}v)(t_1)|| \to 0$ independently of $v \in B_R$ as $t_2 \to t_1$, which means that $\{F_{\lambda}(B_R)\}$ is equicontinuous.

Thus, combining *StepII* and *StepIII*, we conclude F_{λ} is compact on B_R by Arzela-Ascoli theorem.

Step IV: F_{λ} is continuous in B_R . Let $\{v_n\}$ be a sequence which converges to v in B_R . Since, f is continuous in second variable, for each $\rho \in J$, we have

$$\lim_{n \to \infty} f(\varrho, v_n(\varrho)) = f(\varrho, v(\varrho)).$$
(6.3.24)

So, we can conclude that

$$\sup_{\varrho \in J} \|f(\varrho, v_n(\varrho)) - f(\varrho, v(\varrho))\| \to 0 \quad \text{as} \quad n \to \infty.$$
(6.3.25)

From (6.3.19), and (6.3.25) we obtain that

$$\begin{aligned} \|p(v_{n}) - p(v)\| &= \left\| \frac{1}{\alpha} e^{\frac{-\gamma}{\alpha}b} \int_{0}^{b} e^{\frac{\gamma}{\alpha}\varrho} \mathcal{S}(\frac{b-\varrho}{\alpha}) [f(\varrho, v_{n}(\varrho)) - f(\varrho, v(\varrho))] d\varrho \right\| \\ &\leqslant \left. \frac{M}{\alpha} e^{\frac{-\gamma}{\alpha}b} \left(\int_{0}^{b} e^{\frac{\gamma}{\alpha}\varrho} d\varrho \right) \sup_{\varrho \in J} \|f(\varrho, v_{n}(\varrho)) - f(\varrho, v(\varrho))\| \\ &\leqslant \left. \frac{M}{\gamma} [1 - e^{\frac{-\gamma}{\alpha}b}] \sup_{\varrho \in J} \|f(\varrho, v_{n}(\varrho)) - f(\varrho, v(\varrho))\| \\ &\leqslant \left. \frac{M}{\gamma} \sup_{\varrho \in J} \|f(\varrho, v_{n}(\varrho)) - f(\varrho, v(\varrho))\| \\ &\to 0 \quad \text{as} \quad n \to \infty, \end{aligned}$$
(6.3.26)

therefore, (6.3.18) and (6.3.26) imply that

$$\|u_{\lambda}(\varrho, v_n) - u_{\lambda}(\varrho, v)\| \to 0 \quad \text{as} \quad n \to \infty,$$
(6.3.27)

and (6.3.25), (6.3.27) yield

$$\begin{aligned} \|(F_{\lambda}v_{n})(t) - (F_{\lambda}v)(t)\| &\leqslant \frac{M}{\alpha}e^{\frac{-\gamma}{\alpha}t} \int_{0}^{t} e^{\frac{\gamma}{\alpha}\varrho} \|f(\varrho, v_{n}(\varrho)) - f(\varrho, v(\varrho))\| d\varrho \\ &+ \frac{MM_{B}}{\alpha}e^{\frac{-\gamma}{\alpha}t} \int_{0}^{t} e^{\frac{\gamma}{\alpha}\varrho} \|u_{\lambda}(\varrho, v_{n}(\varrho)) - u_{\lambda}(\varrho, v(\varrho))\| d\varrho \\ &\leqslant \frac{M}{\gamma} \sup_{\varrho \in J} \|f(\varrho, v_{n}(\varrho)) - f(\varrho, v(\varrho))\| \\ &+ \frac{MM_{B}}{\gamma} \sup_{\varrho \in J} \|u_{\lambda}(\varrho, v_{n}(\varrho)) - u_{\lambda}(\varrho, v(\varrho))\| \\ &\to 0 \quad \text{as} \quad n \to \infty, \end{aligned}$$

$$(6.3.28)$$

which means that F_{λ} is continuous in B_R .

Hence, by Schauder fixed point theorem there exists a fixed point of F_{λ} , thus a mild solution to (6.1.2).

Theorem 6.3.12. Suppose that the hypotheses (H1)-(H4) hold. Moreover, assume that f is uniformly bounded by a positive constant K. Then, the fractional semilinear system (6.1.2) is approximately controllable on J.

Proof. Let x_{λ} is a mild solution to the problem (6.1.2) with the control

$$u_{\lambda}(t, x_{\lambda}) = B^* \mathcal{S}^*(\frac{b-t}{\alpha}) R(\lambda, \Gamma_0^b) p(x_{\lambda}),$$

where

$$p(x_{\lambda}) = x^{b} - e^{\frac{-\gamma}{\alpha}b} \mathcal{S}(\frac{b}{\alpha}) x_{0} - \frac{1}{\alpha} e^{\frac{-\gamma}{\alpha}b} \int_{0}^{b} e^{\frac{\gamma}{\alpha}\varrho} \mathcal{S}(\frac{b-\varrho}{\alpha}) f(\varrho, x_{\lambda}(\varrho)) d\varrho,$$

and satisfies

$$\begin{aligned} x_{\lambda}(b) &= e^{\frac{-\gamma}{\alpha}b} \mathcal{S}(\frac{b}{\alpha}) x_{0} + \frac{1}{\alpha} e^{\frac{-\gamma}{\alpha}b} \int_{0}^{b} e^{\frac{\gamma}{\alpha}\varrho} \mathcal{S}(\frac{b-\varrho}{\alpha}) [f(\varrho, x_{\lambda}(\varrho)) + Bu_{\lambda}(\varrho, x_{\lambda})] d\varrho \\ &= x^{b} - p(x_{\lambda}) + \left(\frac{1}{\alpha} e^{\frac{-\gamma}{\alpha}b} \int_{0}^{b} e^{\frac{\gamma}{\alpha}\varrho} \mathcal{S}(\frac{b-\varrho}{\alpha}) BB^{*} \mathcal{S}^{*}(\frac{b-\varrho}{\alpha}) d\varrho \right) R(\lambda, \Gamma_{0}^{b}) p(x_{\lambda}) \\ &= x^{b} - p(x_{\lambda}) + \Gamma_{0}^{b} R(\lambda, \Gamma_{0}^{b}) p(x_{\lambda}) \\ &= x^{b} - \lambda R(\lambda, \Gamma_{0}^{b}) p(x_{\lambda}). \end{aligned}$$
(6.3.29)

Since, f is uniformly bounded, we have

$$\int_0^b \|f(\varrho, x_\lambda(\varrho))\|^2 d\varrho \leqslant K^2 b.$$

Hence, the sequence $f(\cdot, x_{\lambda}(\cdot))$ is bounded in $L^{2}(J, \mathbb{X})$. So, we have a subsequence of $\{f(\cdot, x_{\lambda}(\cdot)) : \lambda > 0\}$, still denoted by itself, converges weakly to some $f(\cdot) \in L^{2}(J, \mathbb{X})$. Define

$$\omega = x^b - e^{\frac{-\gamma}{\alpha}b} \mathcal{S}(\frac{b}{\alpha}) x_0 - \frac{1}{\alpha} e^{\frac{-\gamma}{\alpha}b} \int_0^b e^{\frac{\gamma}{\alpha}\varrho} \mathcal{S}(\frac{b-\varrho}{\alpha}) f(\varrho) d\varrho.$$

It follows that

$$\begin{aligned} \|p(x_{\lambda}) - \omega\| &= \left\| \frac{1}{\alpha} e^{\frac{-\gamma}{\alpha}b} \int_{0}^{b} e^{\frac{\gamma}{\alpha}\varrho} \mathcal{S}(\frac{b-\varrho}{\alpha}) [f(\varrho, x_{\lambda}(\varrho)) - f(\varrho)] d\varrho \right\| \\ &\leqslant \sup_{t \in J} \left\| \frac{1}{\alpha} e^{\frac{-\gamma}{\alpha}t} \int_{0}^{t} e^{\frac{\gamma}{\alpha}\varrho} \mathcal{S}(\frac{t-\varrho}{\alpha}) [f(\varrho, x_{\lambda}(\varrho)) - f(\varrho)] d\varrho \right\|. \end{aligned}$$

By the proof of Theorem 6.3.11, it is easy to observe that the operator

$$l(\cdot) \to \frac{1}{\alpha} e^{\frac{-\gamma}{\alpha}(\cdot)} \int_0^{\cdot} e^{\frac{\gamma}{\alpha}\varrho} \mathcal{S}(\frac{\cdot-\varrho}{\alpha}) l(\varrho) d\varrho : L^2(J, \mathbb{X}) \to C(J, \mathbb{X}),$$

is compact, consequently

$$||p(x_{\lambda}) - \omega|| \to 0 \quad \text{as} \quad \lambda \to 0^+.$$
 (6.3.30)

Then, from (6.3.29), (6.3.30), and Lemma 6.3.9, we obtain

$$\begin{aligned} \|x_{\lambda}(b) - x^{b}\| &\leq \|\lambda R(\lambda, \Gamma_{0}^{b}) p(x_{\lambda})\| \\ &\leq \|\lambda R(\lambda, \Gamma_{0}^{b}) \omega\| + \|\lambda R(\lambda, \Gamma_{0}^{b})\| \|p(x_{\lambda}) - \omega\| \\ &\leq \|\lambda R(\lambda, \Gamma_{0}^{b}) \omega\| + \|p(x_{\lambda}) - \omega\| \to 0 \text{ as } \lambda \to 0^{+}. \end{aligned}$$

It proves the approximate controllability of (6.1.2).

84

6.4 Examples

Example(1): Consider the deformable fractional partial differential equation as given below

$$\begin{cases} D^{\frac{1}{2}}x(t,v) = \frac{\partial^2}{\partial v^2}x(t,v) + \frac{1}{4}\sin t \frac{|x(t,v)|}{1+|x(t,v)|}, & v \in (0,1), 0 < t < 1; \\ x(t,1) = x(t,0) = 0, & 0 \leqslant t \leqslant 1; \\ x(0,v) = x_0(v), & v \in [0,1], \end{cases}$$
(6.4.1)

where $\mathbb{X} = L^2[0, 1], x_0(v) \in \mathbb{X}$. Define Aw = w'' with

 $D(A) = \{ w \in \mathbb{X} : w, w' \text{ are absolutely continuous and } w'' \in \mathbb{X}, w(0) = w(1) = 0 \}.$

Then

$$Aw = \sum_{n=1}^{\infty} -n^2 < w, e_n > e_n, \ w \in D(A),$$
(6.4.2)

where $e_n(v) = \sqrt{\frac{2}{\pi}} \sin(nv), 0 \leq v \leq 1, n = 1, 2, \dots$ It is well known that A generates a C_0 -semigroup $\mathcal{S}(t)(t \geq 0)$, on X and is expressed as

$$\mathcal{S}(t)w = \sum_{n=1}^{\infty} e^{-n^2 t} < w, e_n > e_n, \ w \in \mathbb{X},$$
(6.4.3)

with $\|\mathcal{S}(t)\| \leq 1$, for any $t \geq 0$. Put $x(t) = x(t, \cdot)$, that means $x(t)(v) = x(t, v), 0 \leq t, v \leq 1$. Further

$$f(t, x(t)) = \frac{1}{4} \sin t \frac{|x(t, \cdot)|}{1 + |x(t, \cdot)|},$$

then, the system (6.4.1) can be rewritten into the abstract form of (6.1.1).

$$||f(t,x_1) - f(t,x_2)|| \leq \frac{1}{4} \frac{||x_1 - x_2||}{(1 + ||x_1||)(1 + ||x_2||)} \\ \leq \frac{1}{4} ||x_1 - x_2||.$$

Therefore $L = \frac{1}{4}$, also we have $\alpha = \gamma = \frac{1}{2}$ and M = 1. So $\frac{ML}{\gamma} = \frac{1}{2} < 1$. Hence the required assumptions for Theorem 6.3.5 are fulfilled, and we have a unique mild solution for the system (6.4.1). **Example(2):** Let a control system governed by a deformable fractional partial differential equation as

$$\begin{cases} D^{\frac{1}{2}}x(t,z) = \frac{\partial^2}{\partial z^2}x(t,z) + u(t,z) + \frac{1}{8}\frac{e^-t}{1+e^t}\frac{|x(t,z)|}{1+|x(t,z)|}, & z \in (0,1), t \in (0,b]; \\ x(t,0) = x(t,1) = 0, & t \in [0,b]; \\ x(0,z) = x_0(z), & z \in [0,1], \end{cases}$$
(6.4.4)

where $\mathbb{X} = U = L^2[0,1], x_0(z) \in \mathbb{X}, J = [0,b]$. Define Aw = w'' with

 $D(A) = \{ w \in \mathbb{X} : w, w' \text{ are absolutely continuous and } w'' \in \mathbb{X}, w(0) = w(1) = 0 \}.$

Then, A generates a compact semigroup $\mathcal{S}(t)(t > 0)$ given by expression (6.4.3), clearly assumption (H1) holds. Let $B: U \to \mathbb{X}$ is defined as $Bu(t) = u(t, \cdot)$, which is linear bounded operator. It is easy to observe that the assumptions (H2) and (H3) hold with $\phi(t) = \frac{e^{-t}}{1+e^{t}}$ and $K = \frac{1}{8}$.

By Theorem 6.3.7, the linear system corresponding to (6.4.4) is approximately controllable on J iff

$$B^* \mathcal{S}^* (\frac{b-t}{\alpha}) x = 0, \ t \in J \Longrightarrow x = 0.$$
(6.4.5)

Using (6.4.3), observe that

$$B^*\mathcal{S}^*(\frac{b-t}{\alpha})x = \sum_{n=1}^{\infty} e^{-n^2(\frac{b-t}{\alpha})} < x, e_n > e_n, \ x \in \mathbb{X}, \ t \in J.$$

Therefore the condition (6.4.5) hold, and hence the assumption (H4). Thus by Theorem 6.3.12, we have approximate controllability of the system (6.4.4) on J.

Chapter 7

Monotone Iterative Technique for Non-Autonomous Nonlocal Differential Equations

0

7.1 Introduction

Monotone iterative technique is an effective method to find the existence and uniqueness of mild solutions. Using this method, we get monotone sequences of approximate solutions that converge to maximal and minimal mild solutions. Du [121], first used this technique to find extremal mild solutions for a differential equation. Recently, in [49; 52; 56; 109] MIT has been used to established the existence and uniqueness results for various differential systems. The technique has been used for autonomous systems till now. Motivated by this fact, in this chapter, we will study monotone iterative method to find the existence and uniqueness of extremal mild solutions for the following non-autonomous system in an ordered Banach space X :

$$x'(t) + \mathbb{A}(t)x(t) = \mathcal{F}(t, x(t)), \quad 0 < t \le b, \tag{7.1.1}$$

$$x(0) = \sum_{i=1}^{\kappa} \xi_i x(t_i) + x_0, \qquad (7.1.2)$$

The contents of this chapter are accepted in **Demonstratio Mathematica** as Arshi Meraj and Dwijendra Narain Pandey: "Monotone iterative technique for non-autonomous semilinear differential equations with nonlocal condition", DOI: https://doi.org/10.1515/dema-2019-0005.

where $\mathbb{A}(t) : D(\mathbb{A}(t)) \subset \mathbb{X} \to \mathbb{X}$ is a linear operator, J = [0, b], f is a given function from $J \times \mathbb{X}$ to \mathbb{X} satisfying certain assumptions, $t_i \in (0, b)$, for $i < j, t_i < t_j, 0 \neq \xi_i \in \mathbb{R}, i = 1, 2, ..., k \in \mathbb{N}, x_0 \in \mathbb{X}$.

We organize the chapter as following: in section 7.2, we will recall some basic definitions, notations and theorems. In section 7.3, we will show the existence of extremal mild solutions for the system (7.1.1)-(7.1.2), and also we will show the uniqueness of extremal mild solutions with the help of evolution system and Kuratowski measure of noncompactness. Section 7.4 contains an example to illustrate our results.

7.2 Preliminaries

Let $(\mathbb{X}, \|\cdot\|, \leq)$ be a partially ordered complete norm space, $\mathcal{P} = \{x \in \mathbb{X} | x \geq 0\}$ (0 is the zero of \mathbb{X}) is a positive cone of \mathbb{X} . The cone \mathcal{P} is called normal if there is a real number $\mathcal{N} > 0$ such that $0 \leq x_1 \leq x_2 \Rightarrow \|x_1\| \leq \mathcal{N}\|x_2\|$, for all $x_1, x_2 \in \mathbb{X}$, the smallest value of such \mathcal{N} is called normal constant. For $x_1, x_2 \in C(J, \mathbb{X}), x_1 \leq x_2 \Leftrightarrow$ $x_1(t) \leq x_2(t), t \in J$, we denote the intervals $[\nu, \omega] = \{x \in C(J, \mathbb{X}) : \nu \leq x \leq \omega\}$ in $C(J, \mathbb{X})$, and $[\nu(t), \omega(t)] = \{x \in \mathbb{X} : \nu(t) \leq x \leq \omega(t)\}(t \in J)$ in \mathbb{X} . Let us denote $C^1(J, \mathbb{X}) = \{x \in C(J, \mathbb{X}) : x' \text{ exists on } J, x' \in C(J, \mathbb{X}), x(t) \in D(A) \ (t \geq 0)\}$. For more details related to ordered spaces and cone one may see **[11] [66] [63]**.

First, we recall the definition and some basic properties of evolution system. For more details, we refer [82] and [156].

Definition 7.2.1. [156] Let X be a Banach space. A two parameter family of bounded linear operators $\mathbb{U}(t,s), 0 \leq s \leq t \leq b$ on X is called an evolution system, if the following conditions are satisfied :

- 1. $\mathbb{U}(s,s) = I$, where I is the identity operator.
- 2. $\mathbb{U}(\tau, \rho)\mathbb{U}(\rho, s) = \mathbb{U}(\tau, s)$ for $0 \leq s \leq \rho \leq \tau \leq b$.
- 3. $(t,s) \to \mathbb{U}(t,s)$ is strongly continuous for $0 \leq s \leq t \leq b$.

For the family of linear operators $\{\mathbb{A}(t) : t \in J\}$ on \mathbb{X} , we impose the following assumptions :

- (A1) The domain of $\mathbb{A}(t)$ is dense and independent of t, and $\mathbb{A}(t)$ is a closed operator.
- (A2) The resolvent of $\mathbb{A}(t)$ exists for $Re(\vartheta) \leq 0, t \in J$, and $|| R(\vartheta, \mathbb{A}(t)) || \leq \frac{\varsigma}{|\vartheta|+1}$, for some constant $\varsigma > 0$.
- (A3) There are positive constants K, and $\rho \in (0, 1]$ such that

$$\|[\mathbb{A}(\tau_1) - \mathbb{A}(\tau_2)]\mathbb{A}^{-1}(\tau_3)\| \leq K |\tau_1 - \tau_2|^{\rho}$$
, for any $\tau_1, \tau_2, \tau_3 \in J$.

The above assumptions imply that, $-\mathbb{A}(t)$ generates a unique evolution system $\mathbb{U}(t,s)(0 \leq s \leq t \leq b)$ of linear bounded operators on \mathbb{X} , which satisfies the following properties (see [82]; [156]) :

- (i) $\mathbb{U}(t_1, t_2) \in \mathbb{B}(\mathbb{X})$ the space of bounded linear operators on \mathbb{X} , and is continuous strongly for $0 \leq t_2 \leq t_1 \leq b$.
- (ii) $\mathbb{U}(t_1, t_2) x \in D(\mathbb{A}), x \in \mathbb{X}, 0 \leq t_2 \leq t_1 \leq b.$
- (iii) $\mathbb{U}(t_1, t_2)U(t_2, t_3) = \mathbb{U}(t_1, t_3), \ 0 \le t_3 \le t_2 \le t_1 \le b.$
- (iv) $\mathbb{U}(\tau, \tau) = I$ the identity operator on \mathbb{X} , for each $\tau \in J$.
- (v) There is a positive constant \mathcal{M} such that $\|\mathbb{U}(t_1, t_2)\| \leq \mathcal{M}, 0 \leq t_2 \leq t_1 \leq b$.
- (vi) For each fixed s, $\{\mathbb{U}(\tau, s), s \leq \tau\}$ is continuous in τ in uniform operator topology.
- (vii) The derivative $\frac{\partial \mathbb{U}(\tau,s)}{\partial \tau}$ exists in strong operator topology for $0 \leq s < \tau \leq b$, and strongly continuous in τ , where $0 \leq s < \tau \leq b$.
- (viii) $\frac{\partial \mathbb{U}(\tau,s)}{\partial \tau} + \mathbb{A}(\tau)\mathbb{U}(\tau,s) = 0, \ 0 \leq s < \tau \leq b.$

Lemma 7.2.1. ([82]: Lemma 14.1) Let assumptions (A1)-(A3) hold. If $0 \leq \gamma \leq 1, 0 \leq \varpi \leq \delta < 1 + \alpha, 0 < \delta - \gamma \leq 1$, then for any $0 \leq \tau < t < t + \Delta t \leq t_0, 0 \leq \zeta \leq t_0$,

 $\|\mathbb{A}^{\gamma}(\zeta)[\mathbb{U}(t+\triangle t,\tau)-\mathbb{U}(t,\tau)]\mathbb{A}^{-\varpi}(\tau)\| \leqslant C(\varpi,\gamma,\delta)(\triangle t)^{\delta-\gamma}|t-\tau|^{\varpi-\delta}.$

Definition 7.2.2. An evolution system $\mathbb{U}(t,s)$ is called positive if $\mathbb{U}(t,s)y \ge 0$, for $y \in \mathcal{P}$ and $0 \le s \le t \le b$.

Definition 7.2.3. $\omega_0 \in C^1(J, \mathbb{X})$ is called lower solution for the system (7.1.1)-(7.1.2), if

$$\omega_0'(t) + \mathbb{A}(t)\omega_0(t) \leqslant \mathcal{F}(t,\omega_0(t)), \quad 0 < t \leqslant b,$$

$$\omega_0(0) \leqslant \sum_{i=1}^k \xi_i \omega_0(t_i) + x_0. \tag{7.2.1}$$

If the inequalities of (7.2.1) are opposite, solution is known as upper solution.

Theorem 7.2.2. ([82]; [156]) Suppose that the assumptions (A1)-(A3) hold and f satisfies uniform Hölder continuity on J with exponent $\gamma \in (0, 1]$, then the unique solution of the following linear Cauchy problem

$$\begin{aligned} x'(t) + \mathbb{A}(t)x(t) &= \mathcal{F}(t), \quad 0 < t \le b, \\ x(0) &= x_0 \in \mathbb{X}, \end{aligned}$$
(7.2.2)

is

$$x(t) = \mathbb{U}(t,0)x_0 + \int_0^t \mathbb{U}(t,\varrho)\mathcal{F}(\varrho)d\varrho.$$
(7.2.3)

7.3 Main Results

Let us introduce our basic assumptions :

(H0) The evolution system $\mathbb{U}(t,s)$ is positive.

(H1)
$$\xi_i > 0, i = 1, 2, ..., k$$
 and $\mathcal{M} \sum_{i=1}^k \xi_i < 1$.

Assumption (H1) implies that $\|\sum_{i=1}^{k} \xi_i \mathbb{U}(t_i, 0)\| \leq \mathcal{M} \sum_{i=1}^{k} \xi_i < 1$, by operator spectrum theorem, we know that

$$\mathcal{B} := \left(I - \sum_{i=1}^{k} \xi_i \mathbb{U}(t_i, 0)\right)^{-1}$$
(7.3.1)

exists, bounded and can be expressed as

$$\mathcal{B} = \sum_{j=0}^{\infty} \left(\sum_{i=1}^{k} \xi_i \mathbb{U}(t_i, 0) \right)^j.$$
(7.3.2)

Since $\mathbb{U}(t,s)$ is positive, so is \mathcal{B} and

$$\|\mathcal{B}\| \leqslant \sum_{j=0}^{\infty} \|\sum_{i=1}^{k} \xi_{i} \mathbb{U}(t_{i}, 0)\|^{j} \leqslant \sum_{j=0}^{\infty} \left(\mathcal{M}\sum_{i=1}^{k} \xi_{i}\right)^{j} = \frac{1}{1 - \mathcal{M}\sum_{i=1}^{k} \xi_{i}}.$$
 (7.3.3)

Definition 7.3.1. A function $x \in C(J, \mathbb{X})$ is called mild solution of the system (7.1.1) -(7.1.2), if x satisfies the following integral equation :

$$x(t) = \mathbb{U}(t,0)\mathcal{B}x_0 + \sum_{i=1}^k \xi_i \mathbb{U}(t,0)\mathcal{B}\int_0^{t_i} \mathbb{U}(t_i,\varrho)\mathcal{F}(\varrho, x(\varrho))d\varrho + \int_0^t \mathbb{U}(t,\varrho)\mathcal{F}(\varrho, x(\varrho))d\varrho.$$
(7.3.4)

Lemma 7.3.1. If the assumptions (A1)-(A3) and (H1) hold and $\mathcal{F} \in C(J \times \mathbb{X}, \mathbb{X})$, then a mild solution of (7.1.1) -(7.1.2) is given as

$$x(t) = \mathbb{U}(t,0)\mathcal{B}x_0 + \sum_{i=1}^k \xi_i \mathbb{U}(t,0)\mathcal{B}\int_0^{t_i} \mathbb{U}(t_i,\varrho)\mathcal{F}(\varrho, x(\varrho))d\varrho + \int_0^t \mathbb{U}(t,\varrho)\mathcal{F}(\varrho, x(\varrho))d\varrho.$$
(7.3.5)

Proof. By Theorem 7.2.2, the mild solution of problem (7.1.1)-(7.1.2) can be expressed as

$$x(t) = \mathbb{U}(t,0)x(0) + \int_0^t \mathbb{U}(t,\varrho)\mathcal{F}(\varrho,x(\varrho))d\varrho.$$
(7.3.6)

So,

$$x(t_i) = \mathbb{U}(t_i, 0)x(0) + \int_0^{t_i} \mathbb{U}(t_i, \varrho)\mathcal{F}(\varrho, x(\varrho))d\varrho, \quad i = 1, 2, 3, \dots, k.$$
(7.3.7)

By using (7.3.7) in (7.1.2), we get

$$x(0) = \sum_{i=1}^{k} \xi_i [\mathbb{U}(t_i, 0) x(0) + \int_0^{t_i} \mathbb{U}(t_i, \varrho) \mathcal{F}(\varrho, x(\varrho))] d\varrho + x_0$$

Now, assumption (H1) and (7.3.1) imply that

$$x(0) = \mathcal{B}x_0 + \sum_{i=1}^k \xi_i \mathcal{B} \int_0^{t_i} \mathbb{U}(t_i, \varrho) \mathcal{F}(\varrho, x(\varrho)) d\varrho.$$
(7.3.8)
If (7.3.8) imply that (7.3.5) is satisfied.

Hence (7.3.6) and (7.3.8) imply that (7.3.5) is satisfied.

Let us define $\mathcal{Q}: C(J, \mathbb{X}) \to C(J, \mathbb{X})$ as :

$$Qx(t) = \mathbb{U}(t,0)\mathcal{B}x_0 + \sum_{i=1}^k \xi_i \mathbb{U}(t,0)\mathcal{B}\int_0^{t_i} \mathbb{U}(t_i,\varrho)\mathcal{F}(\varrho, x(\varrho))d\varrho + \int_0^t \mathbb{U}(t,\varrho)\mathcal{F}(\varrho, x(\varrho))d\varrho.$$
(7.3.9)

To prove that the system (7.1.1)-(7.1.2) has a mild solution, we need to show the operator Q has a fixed point.

Theorem 7.3.2. Suppose X is a partially ordered complete norm space with normal positive cone \mathcal{P} , the assumptions (H0), (H1) and (A1)-(A3) hold, \mathcal{F} is continuous from $J \times \mathbb{X} \to \mathbb{X}$, $x_0 \in \mathbb{X}$, and $\omega_0, \nu_0 \in C^1(J, \mathbb{X})$ are lower and upper solutions respectively for (7.1.1)-(7.1.2). Moreover, assume the following :

(H2) For $t \in J$, and $y_1, y_2 \in \mathbb{X}$ with $\omega_0(t) \leq y_1 \leq y_2 \leq \nu_0(t)$;

$$\mathcal{F}(t, y_1) \leqslant \mathcal{F}(t, y_2).$$

(H3) There is a constant $\mathcal{L} > 0$ such that for $t \in J$, and monotone sequence $\{x_n\}$ in $[\omega_0(t), \nu_0(t)];$

$$\beta(\{\mathcal{F}(t, x_n)\}) \leqslant \mathcal{L}\beta(\{x_n\}).$$

(H4) Suppose $\omega_n = \mathcal{Q}\omega_{n-1}$ and $\nu_n = \mathcal{Q}\nu_{n-1}$, $n \in \mathbb{N}$ with $\{\omega_n(0)\}\ and \{\nu_n(0)\}\ are$ convergent.

Then, the system (7.1.1)-(7.1.2) has extremal mild solutions in the interval $[\omega_0, \nu_0]$, provided that $K := 2\mathcal{MLb} < 1$.

Proof. Let us denote $D = [\omega_0, \nu_0]$. For any $x \in D$, (H2) implies

$$\mathcal{F}(t,\omega_0(t)) \leqslant \mathcal{F}(t,x(t)) \leqslant \mathcal{F}(t,\nu_0(t)).$$

Therefore, from the normality of \mathcal{P} we get a constant c > 0, such that

$$\|\mathcal{F}(t, x(t))\| \leqslant c, \ x \in D.$$
(7.3.10)

First, we will prove that the map $Q: D \to C(J, \mathbb{X})$ is continuous. Let $\{x_n\}$ be a sequence in D such that $x_n \to x \in D$. Since \mathcal{F} is continuous, so $\mathcal{F}(\varrho, x_n(\varrho)) \to \mathcal{F}(\varrho, x(\varrho))$ for $\varrho \in J$, and from (7.3.10) we get that $\|\mathcal{F}(\varrho, x_n(\varrho)) - \mathcal{F}(\varrho, x(\varrho))\| \leq 2c$.

So, by Lebesgue dominated convergence theorem, we estimate

$$\begin{aligned} \|\mathcal{Q}x_{n}(t) - \mathcal{Q}x(t)\| &\leq \sum_{i=1}^{k} \xi_{i} \|\mathbb{U}(t,0)\| \|\mathcal{B}\| \int_{0}^{t_{i}} \|\mathbb{U}(t_{i},\varrho)\| \|\mathcal{F}(\varrho,x_{n}(\varrho)) - \mathcal{F}(\varrho,x(\varrho))\| d\varrho \\ &+ \int_{0}^{t} \|\mathbb{U}(t,\varrho)\| \|\mathcal{F}(\varrho,x_{n}(\varrho)) - \mathcal{F}(\varrho,x(\varrho))\| d\varrho \\ &\leq \mathcal{M}^{2} \|\mathcal{B}\| \sum_{i=1}^{k} \xi_{i} \int_{0}^{t_{i}} \|\mathcal{F}(\varrho,x_{n}(\varrho)) - \mathcal{F}(\varrho,x(\varrho))\| d\varrho \\ &+ \mathcal{M} \int_{0}^{t} \|\mathcal{F}(\varrho,x_{n}(\varrho)) - \mathcal{F}(\varrho,x(\varrho))\| d\varrho \\ &\to 0 \quad \text{as} \quad n \to \infty. \end{aligned}$$

Thus \mathcal{Q} is continuous map on D.

Next, we will prove $Q: D \to D$ is monotone increasing. Let $x_1, x_2 \in D$, $x_1 \leq x_2$. Since $\mathbb{U}(t,s)$ is positive evolution system and the hypothesis (H2) holds, it is easy to see $Qx_1 \leq Qx_2$. Suppose $\omega'_0(\eta) + \mathbb{A}(\eta)\omega_0(\eta) = h(\eta)$, Definition 7.2.3, implies $h(\eta) \leq \mathcal{F}(\eta, \omega_0(\eta))$, for $\eta \in J$. From Lemma 7.3.1 and the positivity of evolution system, we have

$$\begin{split} \omega_{0}(\eta) &= \mathbb{U}(\eta, 0)\mathcal{B}x_{0} + \sum_{i=1}^{k} \xi_{i}\mathbb{U}(\eta, 0)\mathcal{B}\int_{0}^{t_{i}} \mathbb{U}(t_{i}, \varrho)h(\varrho)d\varrho \\ &+ \int_{0}^{\eta} \mathbb{U}(\eta, \varrho)h(\varrho)d\varrho \\ &\leqslant \mathbb{U}(\eta, 0)\mathcal{B}x_{0} + \sum_{i=1}^{k} \xi_{i}\mathbb{U}(\eta, 0)\mathcal{B}\int_{0}^{t_{i}} \mathbb{U}(t_{i}, \varrho)\mathcal{F}(\varrho, \omega_{0}(\varrho))d\varrho \\ &+ \int_{0}^{\eta} \mathbb{U}(\eta, \varrho)\mathcal{F}(\varrho, \omega_{0}(\varrho))d\varrho \\ &= \mathcal{Q}\omega_{0}(\eta). \end{split}$$

Hence, $\omega_0 \leq \mathcal{Q}\omega_0$. In the same way, we get $\mathcal{Q}\nu_0 \leq \nu_0$. For $u \in D$, we have $\omega_0 \leq \mathcal{Q}\omega_0 \leq \mathcal{Q}u \leq \mathcal{Q}\nu_0 \leq \nu_0$. Therefore, $\mathcal{Q}: D \to D$ is monotone increasing.

Now we will show $\mathcal{Q}(D)$ is equicontinuous on J. For $x \in D$ and $\eta_1, \eta_2 \in J$ with

 $\eta_1 < \eta_2$ we have

$$\begin{split} \|\mathcal{Q}x(\eta_{2}) - \mathcal{Q}x(\eta_{1})\| &\leqslant \|\mathbb{U}(\eta_{2}, 0) - \mathbb{U}(\eta_{1}, 0)\| \|\mathcal{B}\| \|x_{0}\| \\ &+ \|\mathcal{B}\| \left(\sum_{i=1}^{k} \xi_{i} \int_{0}^{t_{i}} \|\mathbb{U}(t_{i}, \varrho)\| \|\mathcal{F}(\varrho, x(\varrho))\| d\varrho \right) \\ &\|\mathbb{U}(\eta_{2}, 0) - \mathbb{U}(\eta_{1}, 0)\| + \int_{0}^{\eta_{1}} \|\mathbb{U}(\eta_{2}, \varrho) - \mathbb{U}(\eta_{1}, \varrho)\| \\ &\|\mathcal{F}(\varrho, x(\varrho))\| d\varrho + \int_{\eta_{1}}^{\eta_{2}} \|\mathbb{U}(\eta_{2}, \varrho)\| \|\mathcal{F}(\varrho, x(\varrho))\| d\varrho \\ &\leqslant \|\mathbb{U}(\eta_{2}, 0) - \mathbb{U}(\eta_{1}, 0)\| \|\mathcal{B}\| \|x_{0}\| \\ &+ \mathcal{M}\|\mathcal{B}\| bc \left(\sum_{i=1}^{k} \xi_{i}\right) \|\mathbb{U}(\eta_{2}, 0) - \mathbb{U}(\eta_{1}, 0)\| \\ &+ \int_{0}^{\eta_{1}} \|\mathbb{U}(\eta_{2}, \varrho) - \mathbb{U}(\eta_{1}, \varrho)\| \|\mathcal{F}(\varrho, x(\varrho))\| d\varrho + \mathcal{M}c(\eta_{2} - \eta_{1}) \\ &\leqslant I_{1} + I_{2} + I_{3} + I_{4}. \end{split}$$

For $\eta_1 = 0$, it is easy to see that $I_3 = 0$. For $\eta_1 > 0$ and $\epsilon > 0$ small enough, we obtain

$$I_{3} \leqslant \int_{0}^{\eta_{1}-\epsilon} \|\mathbb{U}(\eta_{2},\varrho) - \mathbb{U}(\eta_{1},\varrho)\| \|\mathcal{F}(\varrho,x(\varrho))\| d\varrho$$

+ $\int_{\eta_{1}-\epsilon}^{\eta_{1}} \|\mathbb{U}(\eta_{2},\varrho) - \mathbb{U}(\eta_{1},\varrho)\| \|\mathcal{F}(\varrho,x(\varrho))\| d\varrho$
$$\leqslant c(\eta_{1}-\epsilon) \sup_{\varrho \in [0,\eta_{1}-\epsilon]} \|\mathbb{U}(\eta_{2},\varrho) - \mathbb{U}(\eta_{1},\varrho)\| + 2\mathcal{M}c\epsilon.$$

 $\rightarrow 0 \text{ as } \eta_{2} \rightarrow \eta_{1}, \epsilon \rightarrow 0,$

using the continuity of $\{U(\eta, \varrho) : \eta > \varrho\}$ in uniform operator topology. By using Lemma 7.2.1, it is easy to see that $I_1 \to 0$, $I_2 \to 0$ as $\eta_2 \to \eta_1$. Also it is clear from the expression of I_4 that $I_4 \to 0$ as $\eta_2 \to \eta_1$. As a result $||Qx(\eta_2) - Qx(\eta_1)|| \to 0$ as $\eta_2 \to \eta_1$, independently of $x \in D$. Hence Q(D) is equicontinuous on J.

Now we define the sequences

$$\omega_n = \mathcal{Q}\omega_{n-1} \quad \text{and} \quad \nu_n = \mathcal{Q}\nu_{n-1}, \quad n \in \mathbb{N},$$
(7.3.11)

monotonicity of \mathcal{Q} implies

$$\omega_0 \leqslant \omega_1 \leqslant \cdots \leqslant \omega_n \leqslant \cdots \leqslant \nu_n \leqslant \cdots \leqslant \nu_1 \leqslant \nu_0. \tag{7.3.12}$$

Let $S = \{\omega_n\}$ and $S_0 = \{\omega_{n-1}\}$. Then $S_0 = S \cup \{\omega_0\}$ and $\beta(S_0(t)) = \beta(S(t)), t \in J$. Since the sequence $\{\omega_n(0)\}$ is convergent, so $\beta(\{\omega_n(0)\}) = 0$. From Lemma 2.4.5, (H3), (7.3.9) and (7.3.11), we get

$$\begin{aligned}
\beta(S(t)) &= \beta(\mathcal{Q}(S_0(t))) \\
&= \beta\left(\left\{\mathbb{U}(t,0)\mathcal{B}x_0 + \sum_{i=1}^k \xi_i\mathbb{U}(t,0)\mathcal{B}\int_0^{t_i}\mathbb{U}(t_i,\varrho)\mathcal{F}(\varrho,\omega_{n-1}(\varrho))d\varrho\right.\right. \\
&+ \int_0^t\mathbb{U}(t,\varrho)\mathcal{F}(\varrho,\omega_{n-1}(\varrho))d\varrho\right\}\right) \\
&\leqslant \mathcal{M}\beta(\{\omega_n(0)\}) + 2\mathcal{M}\int_0^t\beta(\mathcal{F}(\varrho,\omega_{n-1}(\varrho)))d\varrho \\
&\leqslant 2\mathcal{M}\mathcal{L}\int_0^t\beta(\{\omega_{n-1}(\varrho)\})d\varrho \\
&\leqslant 2\mathcal{M}\mathcal{L}\int_0^t\beta(S_0(\varrho))d\varrho = 2\mathcal{M}\mathcal{L}\int_0^t\beta(S(\varrho))d\varrho.
\end{aligned}$$
(7.3.13)

Since $\{\mathcal{Q}\omega_{n-1}\}$ i.e. $\{\omega_n\}$ is equicontinuous, by Lemma 2.4.4 and (7.3.13), we obtain

$$\begin{array}{lll} \beta(S) &=& \sup_{t \in J} \beta(S(t)) \\ &\leqslant& 2\mathcal{ML}b \sup_{t \in J} \beta(S(t)) = 2\mathcal{ML}b\beta(S) = K\beta(S) \end{array}$$

Since K < 1, therefore $\beta(S) = 0$. Hence the set S is relatively compact in D, so there is a convergent subsequence of $\{\omega_n\}$ in D. From (7.3.12), it is easy to see that $\{\omega_n\}$ itself is a convergent sequence. Let $\omega_n \to \omega^*$ as $n \to \infty$. By (7.3.9) and (7.3.11)

$$\begin{aligned}
\omega_n(t) &= \mathcal{Q}\omega_{n-1}(t) \\
&= \mathbb{U}(t,0)\mathcal{B}x_0 + \sum_{i=1}^k \xi_i \mathbb{U}(t,0)\mathcal{B}\int_0^{t_i} \mathbb{U}(t_i,\varrho)\mathcal{F}(\varrho,\omega_{n-1}(\varrho))d\varrho \\
&+ \int_0^t \mathbb{U}(t,\varrho)\mathcal{F}(\varrho,\omega_{n-1}(\varrho))d\varrho.
\end{aligned}$$

Let $n \to \infty$ and using Lebesgue dominated convergence theorem, we get $\omega^* = \mathcal{Q}\omega^*$ and $\omega^* \in C(J, \mathbb{X})$. Hence ω^* is a mild solution for (7.1.1)-(7.1.2). In the same way there is $\nu^* \in C(J, \mathbb{X})$ with $\nu_n \to \nu^*$ as $n \to \infty$, and $\nu^* = \mathcal{Q}\nu^*$. If $x \in D$ and $x = \mathcal{Q}x$, then $\omega_1 = \mathcal{Q}\omega_0 \leq \mathcal{Q}x = x \leq \mathcal{Q}\nu_0 = \nu_1$. From the process of induction $\omega_n \leq x \leq \nu_n$, and $\omega_0 \leq \omega^* \leq x \leq \nu^* \leq \nu_0$ as $n \to \infty$. That means ω^* is the minimal and ν^* is the maximal mild solution for (7.1.1)-(7.1.2) in $[\omega_0, \nu_0]$.

Corollary 7.3.3. Suppose X is a partially ordered and weakly sequentially complete Banach space with normal positive cone \mathcal{P} , the assumptions (H0), (H1), (H2) and (A1)-(A3) hold, \mathcal{F} is continuous from $J \times X \to X$, $x_0 \in X$, and ω_0 , $\nu_0 \in C^1(J, X)$ are lower and upper solutions respectively for (7.1.1)-(7.1.2), then the system (7.1.1)-(7.1.2) has extremal mild solutions in the interval $[\omega_0, \nu_0]$. Proof. Since the assumptions (H1), (H2) hold, by the proof of Theorem 7.3.2 we have the sequences $\{\omega_n(t)\}$ and $\{\nu_n(t)\}$ defined by (7.3.11) satisfying (7.3.12). So, $\{\omega_n(t)\}$ and $\{\nu_n(t)\}$ are monotone and bounded, hence are precompact on X for X is weakly sequentially complete Banach space. Therefore, the sequences are convergent uniformly. Let $\omega^*(t) = \lim_{n \to \infty} \omega_n(t)$ and $\nu^*(t) = \lim_{n \to \infty} \nu_n(t)$, for $t \in J$. By Lebesgue dominated convergence theorem, we get $\omega^* = \mathcal{Q}\omega^*$ and $\nu^* = \mathcal{Q}\nu^*$ with ω^* , $\nu^* \in C(J, \mathbb{X})$. Hence ω^* and ν^* are a mild solutions for (7.1.1)-(7.1.2). If $x \in D$ and $x = \mathcal{Q}x$, then $\omega_1 = \mathcal{Q}\omega_0 \leq \mathcal{Q}x = x \leq \mathcal{Q}\nu_0 = \nu_1$. From the process of induction $\omega_n \leq x \leq \nu_n$, and $\omega_0 \leq \omega^* \leq x \leq \nu^* \leq \nu_0$ as $n \to \infty$. That means ω^* is the minimal and ν^* is the maximal mild solution for (7.1.1)-(7.1.2).

Theorem 7.3.4. Suppose X is a partially ordered complete norm space, with normal positive cone \mathcal{P} and normal constant \mathcal{N} , the assumptions (H0), (H1), (H2), (H4) and (A1)-(A3) hold, $\mathcal{F} \in C(J \times X, X), x_0 \in X$, and $\omega_0, \nu_0 \in C^1(J, X)$ are lower and upper solutions respectively for (7.1.1)-(7.1.2). Moreover, assume the following :

(H5) There is a constant $\mathcal{L}_1 > 0$ such that, for $t \in J$, and $z_1, z_2 \in \mathbb{X}$ with $\omega_0(t) \leq z_1 \leq z_2 \leq \nu_0(t)$,

$$\mathcal{F}(t, z_2) - \mathcal{F}(t, z_1) \leqslant \mathcal{L}_1(z_2 - z_1)$$

Then, the system (7.1.1)-(7.1.2) has a unique mild solution in $[\omega_0, \nu_0]$, provided that

$$K_1 := \frac{2 - \mathcal{M} \sum_{i=1}^k \xi_i}{1 - \mathcal{M} \sum_{i=1}^k \mathcal{NL}_1 \mathcal{M} b < 1.}$$

Proof. Let $\{x_n\} \subset \mathbb{X}$ be an increasing monotone sequence, and $n, m \in \mathbb{N}$ with n > m. (H2) and (H5) imply

$$0 \leqslant \mathcal{F}(t, x_n) - \mathcal{F}(t, x_m) \leqslant \mathcal{L}_1(x_n - x_m).$$

From the normality of \mathcal{P}

$$\|\mathcal{F}(t,x_n) - \mathcal{F}(t,x_m)\| \leq \mathcal{NL}_1 \|(x_n - x_m)\|.$$
(7.3.14)

So by Lemma 2.4.3, we get

$$\beta(\{\mathcal{F}(t, x_n)\}) \leqslant \mathcal{NL}_1\beta(\{x_n\}).$$

Hence the assumption (H3) holds, and Theorem 7.3.2 is applicable. Therefore, (7.1.1)-(7.1.2) has minimal mild solution ω^* and maximal mild solutions ν^* in $[\omega_0, \nu_0]$.

By (7.3.9), (H5), and the positivity of the operator $\mathbb{U}(t,s)$, we get

$$0 \leq \nu^{*}(t) - \omega^{*}(t) = \mathcal{Q}\nu^{*}(t) - \mathcal{Q}\omega^{*}(t)$$

$$= \sum_{i=1}^{k} \xi_{i}\mathbb{U}(t,0)\mathcal{B}\int_{0}^{t_{i}}\mathbb{U}(t_{i},\varrho)[\mathcal{F}(\varrho,\nu^{*}(\varrho)) - \mathcal{F}(\varrho,\omega^{*}(\varrho))]d\varrho$$

$$+ \int_{0}^{t}\mathbb{U}(t,\varrho)[\mathcal{F}(\varrho,\nu^{*}(\varrho)) - \mathcal{F}(\varrho,\omega^{*}(\varrho))]d\varrho$$

$$\leq \mathcal{L}_{1}\bigg[\sum_{i=1}^{k} \xi_{i}\mathbb{U}(t,0)\mathcal{B}\int_{0}^{t_{i}}\mathbb{U}(t_{i},\varrho)(\nu^{*}(\varrho) - \omega^{*}(\varrho))d\varrho$$

$$+ \int_{0}^{t}\mathbb{U}(t,\varrho)(\nu^{*}(\varrho) - \omega^{*}(\varrho))d\varrho\bigg].$$

Now using (H1), (7.3.3), and the normality of positive cone, we get

$$\begin{aligned} \|\nu^* - \omega^*\| &\leq \mathcal{NL}_1 \bigg[\bigg(\sum_{i=1}^k \xi_i \bigg) \mathcal{M}^2 \|\mathcal{B}\| \int_0^b \|\nu^*(\varrho) - \omega^*(\varrho)\| d\varrho \\ &+ \mathcal{M} \int_0^b \|\nu^*(\varrho) - \omega^*(\varrho)\| d\varrho \bigg] \\ &\leqslant \mathcal{NL}_1 \mathcal{M} b(\|\mathcal{B}\| + 1) \|\nu^* - \omega^*\| \\ &\leqslant \frac{2 - \mathcal{M} \sum_{i=1}^k \xi_i}{1 - \mathcal{M} \sum_{i=1}^k \xi_i} \mathcal{NL}_1 \mathcal{M} b \|\nu^* - \omega^*\| = K_1 \|\nu^* - \omega^*\|. \end{aligned}$$

Since $K_1 < 1$, so $\|\nu^* - \omega^*\| = 0$, i.e. $\nu^*(t) = \omega^*(t)$ for all $t \in J$. Hence $\nu^* = \omega^*$ is the unique mild solution for (7.1.1)-(7.1.2) in $[\omega_0, \nu_0]$.

7.4 Example

Now we consider an example to show how our abstract results can be applied to a concrete problem. Consider the following partial differential equation :

$$\begin{cases} x'(t,z) + a(t,z)\frac{\partial^2}{\partial z^2}x(t,z) = \frac{1}{4}t^2x(t,z), & z \in [0,\pi], \quad t \in J = [0,b]; \\ x(t,0) = 0, \ x(t,\pi) = 0 & t \in J; \\ x(0,z) = \sum_{i=1}^k \xi_i x(t_i,z) + x_0(z), & z \in [0,\pi], \end{cases}$$
(7.4.1)

where $\mathbb{X} = L^2([0, b] \times [0, \pi], \mathbb{R}), x_0(z) \in \mathbb{X}, a(t, z)$ is continuous function and satisfies uniformly Hölder continuity in t, and $0 < t_1 < t_2 < \ldots < t_i < b, \xi_i \neq 0$ (i = $1, 2, \ldots k$) are real numbers. Define

$$\mathbb{A}(t)x(t,z) = a(t,z)\frac{\partial^2}{\partial z^2}x(t,z), \qquad (7.4.2)$$

with domain

$$\mathcal{D}(\mathbb{A}) = \{ w \in \mathbb{X} : w, \ \frac{\partial w}{\partial z} \text{ are absolutely continuous, } \ \frac{\partial^2 w}{\partial z^2} \in \mathbb{X}, \ w(0) = w(\pi) = 0 \}.$$

Then, $-\mathbb{A}(t)$ generates a positive evolution system of bounded linear operators $\mathbb{U}(t,s)$ on \mathbb{X} with $\|\mathbb{U}(t,s)\| \leq \mathcal{M}$ and satisfies the conditions (A1)-(A3) (see [156]). Put $x(t)(z) = x(t,z), t \in [0,b], z \in [0,\pi]$ and $f(t,x(t))(z) = \frac{1}{4}t^2x(t,z)$, then the system (7.4.1) can be rewritten into the abstract form of (7.1.1)-(7.1.2). It is easy to check that assumption (H2) holds. Now, assume the following :

1. $x_0(z) \ge 0$ for $z \in [0, \pi]$, and there exists a function $v(t, z) \ge 0$ such that

$$v'(t,z) + \mathbb{A}(t)v(t,z) \geq \frac{1}{4}t^2v(t,z), \quad t \in J, \ z \in [0,\pi],$$
$$v(t,0) = v(t,\pi) = 0, \quad t \in J,$$
$$v(0,z) \geq \sum_{i=1}^k \xi_i v(t_i,z) + x_0(z), \quad z \in [0,\pi]$$

2.
$$\xi_i > 0, i = 1, 2, \dots, k \text{ and } \mathcal{M} \sum_{i=1}^k \xi_i < 1.$$

From the above assumptions it is clear that (H1) holds, and $\omega_0 = 0$ and $\nu_0 = v(t, z)$ are the lower and upper solutions for the system (7.4.1). So, by Corollary 7.3.3, we conclude that the minimal and maximal mild solutions for (7.4.1) exist between the lower solution 0 and upper solution v.

Chapter 8

Monotone Iterative Technique for Non-Autonomous Nonlocal Integro-Differential Equations

0

8.1 Introduction

In this chapter, we extend the result of chapter 7 for the following non-autonomous integro-differential system involving nonlocal condition in an ordered Banach space X:

$$x'(t) + \mathbb{A}(t)x(t) = \mathcal{F}\left(t, x(t), \int_0^t k(t, s)x(s)ds\right), \quad t \in (0, b],$$

$$x(0) = x_0 + \mathcal{G}(x), \quad (8.1.1)$$

where $\mathbb{A}(t)$ is an X-valued linear operator defined on $D(\mathbb{A}(t)) \subset \mathbb{X}, x_0 \in \mathbb{X}, \mathcal{F}$ is X-valued function defined over $J \times \mathbb{X} \times \mathbb{X}, \mathcal{G}$ is X-valued function defined over $C(J, \mathbb{X})$ with J = [0, b], and k is continuous function from \mathbb{D} to \mathbb{R}^+ where $\mathbb{D} := \{(\tau, s) : 0 \leq s \leq \tau \leq b\}$.

We organize the chapter as following. In section 8.2, we will recall some basic theory. In section 8.3, we will establish the existence of extremal mild solutions for

The contents of this chapter are accepted in **Filomat** as Arshi Meraj, Dwijendra Narain Pandey: "Existence and uniqueness of extremal mild solutions for non-autonomous nonlocal integro-differential equations via monotone iterative technique".

the system (8.1.1), and also we will show the uniqueness of extremal mild solutions with the help of evolution system and Kuratowski measure of noncompactness by MIT. In last section, we will discuss an example to illustrate our results.

8.2 Preliminaries

Let $(\mathbb{X}, \|\cdot\|, \leqslant)$ is a partially ordered complete norm space, $\mathcal{P} = \{x \in \mathbb{X} : x \ge 0\}$ (0 is the zero element of \mathbb{X}) is a positive cone of \mathbb{X} . The cone \mathcal{P} is known as normal if there is a real number $\mathcal{N} > 0$ such that $0 \leqslant x_1 \leqslant x_2 \Rightarrow \|x_1\| \leqslant \mathcal{N}\|x_2\|$, for all $x_1, x_2 \in \mathbb{X}$, the smallest value of such \mathcal{N} is called normal constant. For $x_1, x_2 \in$ $C(J, \mathbb{X}), x_1 \leqslant x_2 \Leftrightarrow x_1(t) \leqslant x_2(t), \forall t \in J$. For $\nu, \omega \in C(J, \mathbb{X})$ with $\nu \leqslant \omega$, we will use the notation $[\nu, \omega] := \{x \in C(J, \mathbb{X}) : \nu \leqslant x \leqslant \omega\}$ for an interval in $C(J, \mathbb{X})$, and $[\nu(t), \omega(t)] := \{x \in \mathbb{X} : \nu(t) \leqslant x \leqslant \omega(t)\}(t \in J)$ for an interval in \mathbb{X} . Let us denote $C^1(J, \mathbb{X}) = \{x \in C(J, \mathbb{X}) : x' \text{ exists on } J, x' \in C(J, \mathbb{X}), x(t) \in D(\mathbb{A}) \ (t \ge 0)\}$. For our convenience, we denote $\mathcal{K}x(t) := \int_0^t k(t, s)x(s)ds$, and $K^* := \sup_{(t,s)\in\mathbb{D}} k(t, s)$.

First, we recall the definition and some basic properties of evolution system. For more details, we refer [82] and [156].

Definition 8.2.1. ([150]) Let X be a Banach space. A two parameter family of bounded linear operators $\mathbb{U}(t_1, t_2), 0 \leq t_2 \leq t_1 \leq b$ on X is known as evolution system, if :

- 1. $\mathbb{U}(s,s) = I$, where I is the identity operator.
- 2. $\mathbb{U}(t_1, t_2)\mathbb{U}(t_2, t_3) = \mathbb{U}(t_1, t_3)$ for $0 \leq t_3 \leq t_2 \leq t_1 \leq b$.
- 3. $(t_1, t_2) \to \mathbb{U}(t_1, t_2)$ is strongly continuous for $0 \leq t_2 \leq t_1 \leq b$.

For the family of linear operators $\{\mathbb{A}(t) : t \in J\}$ on \mathbb{X} , we impose the restrictions given below :

- (A1) The operator A(t) is closed, the domain of A(t) is independent of t, and dense in X.
- (A2) The resolvent of $\mathbb{A}(t)$ exists for $Re(\vartheta) \leq 0, t \in J$, and $||R(\vartheta, \mathbb{A}(t))|| \leq \frac{\varsigma}{|\vartheta|+1}$, for some positive constant ς .

(A3) There exist positive constants K, and $\rho \in (0, 1]$ such that $\|[\mathbb{A}(\tau_1) - \mathbb{A}(\tau_2)]\mathbb{A}^{-1}(\tau_3)\| \leq 1$

$$K|\tau_1 - \tau_2|^{\rho}$$
 for any $\tau_1, \tau_2, \tau_3 \in J$.

Theorem 8.2.1. ([156]) Suppose that the assumptions (A1)-(A3) hold, then $-\mathbb{A}(t)$ generates a unique evolution system { $\mathbb{U}(t_1, t_2) : 0 \leq t_2 \leq t_1 \leq b$ }, which satisfies the following properties :

- (i) There exists a positive constant \mathcal{M} such that $\|\mathbb{U}(t_1, t_2)\| \leq \mathcal{M}, 0 \leq t_2 \leq t_1 \leq b$.
- (ii) For $0 \leq t_2 < t_1 \leq b$, the derivative $\frac{\partial \mathbb{U}(t_1,t_2)}{\partial t_1}$ exists in strong operator topology, is strongly continuous, and belongs to $\mathbb{B}(\mathbb{X})$ (set of all bounded linear operators on \mathbb{X}). Moreover,

$$\frac{\partial \mathbb{U}(t_1, t_2)}{\partial t_1} + \mathbb{A}(t_1)\mathbb{U}(t_1, t_2) = 0, \ 0 \leqslant t_2 < t_1 \leqslant b.$$

Proposition 8.2.2. ([189]) The family of operators $\{\mathbb{U}(t_1, t_2), t_2 < t_1\}$ is continuous in t_1 uniformly for t_2 with respect to operator norm.

Theorem 8.2.3. ([156]) Suppose that the assumptions (A1)-(A3) hold and \mathcal{F} satisfies uniform Hölder continuity on J with exponent $\alpha \in (0,1]$, then the unique solution of the following linear Cauchy problem

$$x'(t) + \mathbb{A}(t)x(t) = \mathcal{F}(t), \quad 0 < t \le b,$$

 $x(0) = x_0 \in \mathbb{X},$ (8.2.1)

is given as

$$x(t) = \mathbb{U}(t,0)x_0 + \int_0^t \mathbb{U}(t,\eta)\mathcal{F}(\eta)d\eta.$$
(8.2.2)

Definition 8.2.2. A mild solution of (8.1.1) is a function $x \in C(J, \mathbb{X})$ satisfying the following integral equation

$$x(\varrho) = \mathbb{U}(\varrho, 0)(x_0 + \mathcal{G}(x)) + \int_0^{\varrho} \mathbb{U}(\varrho, \eta) \mathcal{F}(\eta, x(\eta), \mathcal{K}x(\eta)) d\eta, \ \varrho \in J.$$

Definition 8.2.3. An evolution system $\mathbb{U}(t,s)$ is called positive if $\mathbb{U}(t,s)y \ge 0$, for all $y \in \mathcal{P}$ and $0 \le s \le t \le b$.

Definition 8.2.4. $\omega_0 \in C^1(J, \mathbb{X})$ is called lower solution for the system (8.1.1), if

$$\omega_0'(t) + \mathbb{A}(t)\omega_0(t) \leqslant \mathcal{F}\left(t,\omega_0(t),\int_0^t k(t,s)\omega_0(s)ds\right), \quad t \in (0,b], \\
\omega_0(0) \leqslant x_0 + \mathcal{G}(\omega_0).$$
(8.2.3)

If the inequalities of (8.2.3) are opposite, solution is known as upper solution.

8.3 Main Results

First, we will show the existence of extremal mild solutions for (8.1.1), then the uniqueness will be discussed.

Let us define $\mathcal{Q}: C(J, \mathbb{X}) \to C(J, \mathbb{X})$ in the following way :

$$\mathcal{Q}x(\varrho) = \mathbb{U}(\varrho, 0)(x_0 + \mathcal{G}(x)) + \int_0^{\varrho} \mathbb{U}(\varrho, \eta) \mathcal{F}(\eta, x(\eta), \mathcal{K}x(\eta)) d\eta. \quad (8.3.1)$$

To prove that the system (8.1.1) has a mild solution, we need to show the operator Q has a fixed point.

Theorem 8.3.1. Suppose X is a partially ordered complete norm space with normal positive cone \mathcal{P} , the assumptions (A1)-(A3) hold, the evolution system $\mathbb{U}(t,s)$ is positive, \mathcal{F} is continuous from $J \times \mathbb{X} \times \mathbb{X} \to \mathbb{X}$, $x_0 \in \mathbb{X}$, and ω_0 , $\nu_0 \in C^1(J, \mathbb{X})$ with $\omega_0 \leq \nu_0$ are lower and upper solutions respectively for (8.1.1). Moreover, assume the following :

(H1) For $t \in J$, we have

$$\mathcal{F}(t, y_1, x_1) \leqslant \mathcal{F}(t, y_2, x_2),$$

where $y_1, y_2 \in \mathbb{X}$ with $\omega_0(t) \leq y_1 \leq y_2 \leq \nu_0(t)$, and $\mathcal{K}\omega_0(t) \leq x_1 \leq x_2 \leq \mathcal{K}\nu_0(t)$.

(H2) There exists a constant $\mathcal{L} > 0$ such that for all $t \in J$,

$$\beta(\{\mathcal{F}(t, y_n, x_n)\}) \leqslant \mathcal{L}(\beta(\{y_n\}) + \beta(\{x_n\})),$$

where $\{y_n\} \subset [\omega_0(t), \nu_0(t)]$ and $\{x_n\} \subset [\mathcal{K}\omega_0(t), \mathcal{K}\nu_0(t)]$ are monotone increasing or decreasing sequences.

(H3) $\mathcal{G}: C(J, \mathbb{X}) \to \mathbb{X}$ is a continuous increasing compact function.

Then, the system (8.1.1) has extremal mild solutions in the interval $[\omega_0, \nu_0]$, provided that

$$\Lambda_1 := 2\mathcal{ML}b(1+bK^*) < 1.$$

Proof. Let us denote $I = [\omega_0, \nu_0]$. For any $x \in I$, (H1) implies

$$\mathcal{F}(\varrho,\omega_0(\varrho),\mathcal{K}\omega_0(\varrho)) \leqslant \mathcal{F}(\varrho,x(\varrho),\mathcal{K}x(\varrho)) \leqslant \mathcal{F}(\varrho,\nu_0(\varrho),\mathcal{K}\nu_0(\varrho)).$$

Therefore, from the normality of \mathcal{P} we get a constant c > 0, such that

$$\|\mathcal{F}(\varrho, x(\varrho), \mathcal{K}x(\varrho))\| \leqslant c, \ x \in I.$$
(8.3.2)

First, we will prove that the map $Q: I \to C(J, \mathbb{X})$ is continuous. Let $\{x_n\}$ be a sequence in I such that $x_n \to x \in I$. Since \mathcal{G} , \mathcal{F} are continuous, so $\mathcal{G}x_n \to \mathcal{G}x$, and $\mathcal{F}(\varrho, x_n(\varrho), \mathcal{K}x_n(\varrho)) \to \mathcal{F}(\varrho, x(\varrho), \mathcal{K}x(\varrho))$ for $\varrho \in J$, and from (8.3.2) we get that $\|\mathcal{F}(\varrho, x_n(\varrho), \mathcal{K}x_n(\varrho)) - \mathcal{F}(\varrho, x(\varrho), \mathcal{K}x(\varrho))\| \leq 2c$. So, by Lebesgue dominated convergence theorem, we estimate

$$\begin{aligned} \|\mathcal{Q}x_n(t) - \mathcal{Q}x(t)\| &\leq \mathcal{M} \|\mathcal{G}x_n - \mathcal{G}x\| \\ &+ \mathcal{M} \int_0^t \|\mathcal{F}(\varrho, x_n(\varrho), \mathcal{K}x_n(\varrho)) - \mathcal{F}(\varrho, x(\varrho), \mathcal{K}x(\varrho))\| d\varrho \\ &\to 0 \quad \text{as} \quad n \to \infty. \end{aligned}$$

Thus \mathcal{Q} is continuous map on I.

Next, we will prove $Q: I \to I$ is monotone increasing. Let $x_1, x_2 \in I$, $x_1 \leq x_2$. Using the positivity of $\mathbb{U}(t,s)$, the hypotheses (H1) and (H3), it is easy to see that $Qx_1 \leq Qx_2$. Suppose $\omega'_0(\eta) + \mathbb{A}(\eta)\omega_0(\eta) = h(\eta)$, Definition 8.2.4 implies $h(\eta) \leq \mathcal{F}(\eta, \omega_0(\eta), \mathcal{K}\omega_0(\eta))$ for $\eta \in J$, and $\omega_0(0) \leq x_0 + \mathcal{G}(\omega_0)$. Therefore, for any $t \in J$, Theorem 8.2.3 yields

$$\begin{aligned} \omega_0(t) &= \mathbb{U}(t,0)\omega_0(0) + \int_0^t \mathbb{U}(t,\eta)h(\eta)d\eta \\ &\leqslant \mathbb{U}(t,0)(x_0 + \mathcal{G}(\omega_0)) + \int_0^t \mathbb{U}(t,\eta)\mathcal{F}(\eta,\omega_0(\eta),\mathcal{K}\omega_0(\eta))d\eta \\ &= \mathcal{Q}\omega_0(t). \end{aligned}$$

Hence, $\omega_0 \leq \mathcal{Q}\omega_0$. In the same way, we get $\mathcal{Q}\nu_0 \leq \nu_0$. Let $u \in I$, so we have $\omega_0 \leq \mathcal{Q}\omega_0 \leq \mathcal{Q}u \leq \mathcal{Q}\nu_0 \leq \nu_0$, that means $\mathcal{Q}u \in I$. Therefore, $\mathcal{Q}: I \to I$ is monotone increasing.

Now, we will show $\mathcal{Q}(I)$ is equicontinuous on J. For $x \in I$ and $\eta_1, \eta_2 \in J$ with $\eta_1 < \eta_2$, we have

$$\begin{aligned} \|\mathcal{Q}x(\eta_2) - \mathcal{Q}x(\eta_1)\| &\leqslant \|\mathbb{U}(\eta_2, 0) - \mathbb{U}(\eta_1, 0)\| \|x_0 + \mathcal{G}x\| \\ &+ \int_0^{\eta_1} \|\mathbb{U}(\eta_2, \varrho) - \mathbb{U}(\eta_1, \varrho)\| \|\mathcal{F}(\varrho, x(\varrho), \mathcal{K}x(\varrho))\| d\varrho \\ &+ \int_{\eta_1}^{\eta_2} \|\mathbb{U}(\eta_2, \varrho)\| \|\mathcal{F}(\varrho, x(\varrho), \mathcal{K}x(\varrho))\| d\varrho \\ &\leqslant I_1 + I_2 + I_3. \end{aligned}$$

For $\eta_1 = 0$, it is easy to see that $I_2 = 0$. For $\eta_1 > 0$ and $\epsilon > 0$ small enough, we obtain

$$I_{2} \leqslant \int_{0}^{\eta_{1}-\epsilon} \|\mathbb{U}(\eta_{2},\varrho) - \mathbb{U}(\eta_{1},\varrho)\| \|\mathcal{F}(\varrho, x(\varrho), \mathcal{K}x(\varrho))\| d\varrho$$

+
$$\int_{\eta_{1}-\epsilon}^{\eta_{1}} \|\mathbb{U}(\eta_{2},\varrho) - \mathbb{U}(\eta_{1},\varrho)\| \|\mathcal{F}(\varrho, x(\varrho), \mathcal{K}x(\varrho))\| d\varrho$$

$$\leqslant c(\eta_{1}-\epsilon) \sup_{\varrho \in [0,\eta_{1}-\epsilon]} \|\mathbb{U}(\eta_{2},\varrho) - \mathbb{U}(\eta_{1},\varrho)\| + 2\mathcal{M}c\epsilon.$$

$$\to 0 \quad \text{as} \quad \eta_{2} \to \eta_{1}, \epsilon \to 0,$$

by using the continuity of $\{\mathbb{U}(\eta, \varrho) : \varrho < \eta\}$ in η in uniform operator topology. Also It is clear from the expression of I_1, I_3 that $I_1 \to 0$, $I_3 \to 0$ as $\eta_2 \to \eta_1$. As a result $\|Qx(\eta_2) - Qx(\eta_1)\| \to 0$ as $\eta_2 \to \eta_1$, independently of $x \in I$. Hence Q(I) is equicontinuous on J.

Now we define the sequences

$$\omega_n = \mathcal{Q}\omega_{n-1} \quad \text{and} \quad \nu_n = \mathcal{Q}\nu_{n-1}, \quad n \in \mathbb{N},$$
(8.3.3)

monotonicity of \mathcal{Q} implies

$$\omega_0 \leqslant \omega_1 \leqslant \cdots \leqslant \nu_n \leqslant \cdots \leqslant \nu_1 \leqslant \nu_0. \tag{8.3.4}$$

Let $S = \{\omega_n\}$ and $S_0 = \{\omega_{n-1}\}$. Then $S_0 = S \cup \{\omega_0\}$ and $\beta(S_0(t)) = \beta(S(t))$, $t \in J$. Observe that $\beta(\mathbb{U}(t,0)(x_0)) = 0 = \beta(\mathbb{U}(t,0)\mathcal{G}(\omega_{n-1}))$ for $\{x_0\}$ is compact set, \mathcal{G} is compact map and $\mathbb{U}(t,0)$ is bounded. Also, with the help of Lemma 2.4.4 and Lemma 2.4.5, we observe that

$$\beta\left(\{\mathcal{K}\omega_{n-1}(\eta)\}\right) = \beta\left(\int_{0}^{\eta} k(\eta, s)\omega_{n-1}(s)ds\right)$$
$$\leqslant K^{*}\beta\left(\int_{0}^{\eta} \omega_{n-1}(s)ds\right)$$
$$\leqslant 2K^{*}\int_{0}^{\eta}\beta(\omega_{n-1}(s))ds$$
$$\leqslant 2K^{*}\eta \sup_{s\in[0,\eta]}\beta(S_{0}(s)).$$

Now, from Lemma 2.4.5, (H2), (H3), (8.3.1) and (8.3.3), we get

$$\beta(S(t)) = \beta(\mathcal{Q}(S_0(t)))
= \beta\left(\mathbb{U}(t,0)(x_0 + \mathcal{G}(\omega_{n-1})) + \int_0^t \mathbb{U}(t,\eta)\mathcal{F}(\eta,\omega_{n-1}(\eta),\mathcal{K}\omega_{n-1}(\eta))d\eta\right)
\leqslant \beta(\mathbb{U}(t,0)x_0) + \beta(\mathbb{U}(t,0)\mathcal{G}(\omega_{n-1})) + 2\mathcal{M}\int_0^t \beta\left(\mathcal{F}(\eta,\omega_{n-1}(\eta),\mathcal{K}\omega_{n-1}(\eta))d\eta\right)
\leqslant 2\mathcal{M}\mathcal{L}\int_0^t \left[\beta(\{\omega_{n-1}(\eta)\}) + \beta(\{\mathcal{K}\omega_{n-1}(\eta)\})\right]d\eta
\leqslant 2\mathcal{M}\mathcal{L}b(1 + bK^*) \sup_{t \in J} \beta(S(t)).$$
(8.3.5)

Since $\{\mathcal{Q}\omega_{n-1}\}$ i.e. $\{\omega_n\}$ is equicontinuous, by Lemma 2.4.4 and (8.3.5), we obtain

$$\beta(S) = \sup_{t \in J} \beta(S(t))$$

$$\leqslant 2\mathcal{M}\mathcal{L}b(1+bK^*) \sup_{t \in J} \beta(S(t)) = 2\mathcal{M}\mathcal{L}b(1+bK^*)\beta(S) = \Lambda_1\beta(S).$$

Since $\Lambda_1 < 1$, therefore $\beta(S) = 0$. Hence the set S is relatively compact in I, so there exists a convergent subsequence of $\{\omega_n\}$ in I. From (8.3.4), it is easy to see

that $\{\omega_n\}$ itself is a convergent sequence, let $\omega_n \to \omega^*$ as $n \to \infty$. By (8.3.1) and (8.3.3)

$$\omega_n(t) = \mathcal{Q}\omega_{n-1}(t)
= \mathbb{U}(t,0)(x_0 + \mathcal{G}(\omega_{n-1}))
+ \int_0^t \mathbb{U}(t,\eta)\mathcal{F}(\eta,\omega_{n-1}(\eta),\mathcal{K}\omega_{n-1}(\eta))d\eta.$$
(8.3.6)

In (8.3.6), let $n \to \infty$ and use Lebesgue dominated convergence theorem, we get

$$\omega^*(t) = \mathbb{U}(t,0)(x_0 + \mathcal{G}(\omega^*)) + \int_0^t \mathbb{U}(t,\eta)\mathcal{F}(\eta,\omega^*(\eta),\mathcal{K}\omega^*(\eta))d\eta$$

So, $\omega^* = \mathcal{Q}\omega^*$ and $\omega^* \in C(J, \mathbb{X})$. Hence ω^* is a mild solution for (8.1.1). In the same way there exists $\nu^* \in C(J, \mathbb{X})$ with $\nu_n \to \nu^*$ as $n \to \infty$, and $\nu^* = \mathcal{Q}\nu^*$. Now we show ω^* , ν^* are extremal mild solutions. Let $x \in I$ and $x = \mathcal{Q}x$, then $\omega_1 = \mathcal{Q}\omega_0 \leq \mathcal{Q}x = x \leq \mathcal{Q}\nu_0 = \nu_1$. From the process of induction $\omega_n \leq x \leq \nu_n$, and $\omega_0 \leq \omega^* \leq x \leq \nu^* \leq \nu_0$ as $n \to \infty$. That means ω^* is the minimal and ν^* is the maximal mild solution for (8.1.1) in $[\omega_0, \nu_0]$.

Theorem 8.3.2. Suppose X is a partially ordered complete norm space, with normal positive cone \mathcal{P} and normal constant \mathcal{N} , the assumptions (H1), (H3), (A1)-(A3) hold, the evolution system $\mathbb{U}(t,s)(0 \leq s \leq t \leq b)$ is positive, \mathcal{F} is continuous from $J \times \mathbb{X} \times \mathbb{X}$ to X, $x_0 \in \mathbb{X}$, and $\omega_0, \nu_0 \in C^1(J, \mathbb{X})$ with $\omega_0 \leq \nu_0$ are lower and upper solutions respectively for (8.1.1). Moreover, assume the following :

(H4) There is a constant $\mathcal{L}_1 > 0$ such that for $t \in J$

$$\mathcal{F}(t, y_2, x_2) - \mathcal{F}(t, y_1, x_1) \leqslant \mathcal{L}_1[(y_2 - y_1) + (x_2 - x_1)],$$

where $y_1, y_2 \in \mathbb{X}$ with $\omega_0(t) \leq y_1 \leq y_2 \leq \nu_0(t)$, and $\mathcal{K}\omega_0(t) \leq x_1 \leq x_2 \leq \mathcal{K}\nu_0(t)$.

(H5) There exists a constant $\mathcal{L}_2 > 0$ such that

$$\mathcal{G}(y) - \mathcal{G}(x) \leq \mathcal{L}_2(y - x), \text{ for } x, y \in I \text{ with } x \leq y.$$

Then, the system (8.1.1) has a unique mild solution in $[\omega_0, \nu_0]$, provided that

$$\Lambda_2 := \mathcal{N}\mathcal{M}\bigg[\mathcal{L}_2 + \mathcal{L}_1 b(1 + bK^*)\bigg] < 1.$$

Proof. Let $\{y_n\} \subset [\omega_0(t), \nu_0(t)]$ and $\{x_n\} \subset [\mathcal{K}\omega_0(t), \mathcal{K}\nu_0(t)]$ be increasing monotone sequences. For $t \in J$ and $n, m \in \mathbb{N}$ with n > m, the assumptions (H1) and (H4) imply

$$0 \leq \mathcal{F}(t, y_n, x_n) - \mathcal{F}(t, y_m, x_m) \leq \mathcal{L}_1[(y_n - y_m) + (x_n - x_m)].$$

Since the positive cone is normal, therefore

$$\|\mathcal{F}(t, y_n, x_n) - \mathcal{F}(t, y_m, x_m)\| \leq \mathcal{NL}_1 \|(y_n - y_m) + (x_n - x_m)\|.$$
 (8.3.7)

So by Lemma 2.4.3, we get

$$\mu(\{\mathcal{F}(t, y_n, x_n)\}) \leqslant \mathcal{NL}_1(\mu(\{y_n\}) + \mu(\{x_n\})).$$

Hence the assumption (H2) holds, and Theorem 8.3.1 is applicable. Therefore (8.1.1) has minimal mild solution ω^* and maximal mild solutions ν^* in $[\omega_0, \nu_0]$. From (8.3.1), (H4), (H5), and the positivity of the operator $\mathbb{U}(t, s)$, we get

$$0 \leqslant \nu^{*}(t) - \omega^{*}(t) = \mathcal{Q}\nu^{*}(t) - \mathcal{Q}\omega^{*}(t)$$

$$= \mathbb{U}(t,0)(\mathcal{G}(\nu^{*}) - \mathcal{G}(\omega^{*})) + \int_{0}^{t} \mathbb{U}(t,\eta)[\mathcal{F}(\eta,\nu^{*}(\eta),\mathcal{K}\nu^{*}(\eta)) - \mathcal{F}(\eta,\omega^{*}(\eta),\mathcal{K}\omega^{*}(\eta))]d\eta$$

$$\leqslant \mathcal{L}_{2}\mathbb{U}(t,0)(\nu^{*} - \omega^{*}) + \mathcal{L}_{1}\int_{0}^{t}\mathbb{U}(t,\eta)\Big[(\nu^{*}(\eta) - \omega^{*}(\eta)) + (\mathcal{K}\nu^{*}(\eta) - \mathcal{K}\omega^{*}(\eta))\Big]d\eta.$$

Since the positive cone is normal, therefore

$$\begin{aligned} \|\nu^* - \omega^*\| &\leq \mathcal{N} \bigg[\mathcal{L}_2 \mathcal{M} \|\nu^* - \omega^*\| + \mathcal{M} \mathcal{L}_1 b \bigg(\|\nu^* - \omega^*\| + \|\mathcal{K}\nu^* - \mathcal{K}\omega^*\| \bigg) \bigg] \\ &\leq \mathcal{N} \mathcal{M} \bigg[\mathcal{L}_2 + \mathcal{L}_1 b (1 + bK^*) \bigg] \|\nu^* - \omega^*\| = \Lambda_2 \|\nu^* - \omega^*\|. \end{aligned}$$

Since $\Lambda_2 < 1$, so $\|\nu^* - \omega^*\| = 0$, i.e. $\nu^*(t) = \omega^*(t)$, $\forall t \in J$. Thus $\nu^* = \omega^*$ is the unique mild solution for (8.1.1) in $[\omega_0, \nu_0]$.

8.4 Example

Now we consider an example to show how our abstract results can be applied to a concrete problem. Consider the following partial differential equation :

$$\begin{cases} x'(t,z) + a(t,z)\frac{\partial^2}{\partial z^2}x(t,z) = \frac{1}{25}\frac{e^{-t}}{1+e^t}x(t,z) + \int_0^t \frac{1}{50}e^{-s}x(s,z)ds, \\ z \in [0,\pi], \quad t \in J = [0,b], \\ x(t,0) = 0, \ x(t,\pi) = 0, \quad t \in J, \\ x(0,z) = \frac{e^{x(t,z)}}{1+e^{x(t,z)}} + x_0(z), \quad z \in [0,\pi], \end{cases}$$

$$(8.4.1)$$

where $\mathbb{X} = L^2([0, b] \times [0, \pi], \mathbb{R}), x_0(z) \in \mathbb{X}, a(t, z)$ is continuous function and satisfies uniform Hölder continuity in t. Define

$$\mathbb{A}(t)x(t,z) = a(t,z)\frac{\partial^2}{\partial z^2}x(t,z), \qquad (8.4.2)$$

with domain

$$\mathcal{D}(\mathbb{A}) = \{ w \in \mathbb{X} : w, \ \frac{\partial w}{\partial z} \text{ are absolutely continuous, } \ \frac{\partial^2 w}{\partial z^2} \in \mathbb{X}, \ w(0) = w(\pi) = 0 \}.$$

Then, $-\mathbb{A}(t)$ generates a positive evolution system of bounded linear operators $\mathbb{U}(t,s)$ on X and satisfies the conditions (A1)-(A3) (see [156]). Put

$$\begin{aligned} x(t)(z) &= x(t,z), \ t \in [0,b], \ z \in [0,\pi], \\ \mathcal{F}(t,x(t),\mathcal{K}x(t))(z) &= \frac{1}{25} \frac{e^{-t}}{1+e^{t}} x(t,z) + \int_{0}^{t} \frac{1}{50} e^{-s} x(s,z) ds, \\ (\mathcal{K}x(t))(z) &= \int_{0}^{t} \frac{1}{50} e^{-s} x(s,z) ds, \\ (\mathcal{G}x(t))(z) &= \frac{e^{x(t,z)}}{1+e^{x(t,z)}}. \end{aligned}$$
(8.4.3)

Then the system (8.4.1) can be rewritten into the abstract form of (8.1.1). Now, assume that $x_0(z) \ge 0$ for $z \in [0, \pi]$, and there exists a function $v(t, z) \ge 0$ such that

$$v'(t,z) + \mathbb{A}(t)v(t,z) \geq \mathcal{F}\left(t,v(t,z),\mathcal{K}v(t,z)\right), \quad t \in J, \ z \in [0,\pi],$$
$$v(t,0) = v(t,\pi) = 0, \quad t \in J,$$
$$v(0,z) \geq \mathcal{G}(v(z)) + x_0(z), \quad z \in [0,\pi].$$

From the above assumptions, we have $\omega_0 = 0$ and $\nu_0 = v(t, z)$ are the lower and upper solutions for the system (8.4.1). By (8.4.3), it is easy to verify that the assumptions (H1) and (H3) hold. Suppose $\{x_n\} \subset [\omega_0(t), \nu_0(t)]$ be a monotone increasing sequence. For $n \leq m$

$$\|\mathcal{F}(t, x_m, \mathcal{K}x_m) - \mathcal{F}(t, x_n, \mathcal{K}x_n)\| \leq \frac{1}{25} \left(\|x_m - x_n\| + \|\mathcal{K}x_m - \mathcal{K}x_n\| \right), \text{ hence}$$

$$\beta \left(\mathcal{F}(t, x_n, \mathcal{K}x_n) \right) \leq \frac{1}{25} \left(\beta(\{x_n\}) + \beta(\{\mathcal{K}x_n\}) \right).$$

Therefore, assumption (H2) is satisfied. So, by Theorem 8.3.1, we conclude that the minimal and maximal mild solutions for (8.4.1) exist between the lower solution 0 and upper solution v.

List of Publications

List of papers published/accepted in International Refereed Journals

- A. Meraj, D.N. Pandey, "Existence result for neutral fractional integrodifferential equations with nonlocal integral boundary conditions", *Malaya Journal* of Matematik, 6 (1) (2018), 21-27.
- A. Meraj, D.N. Pandey, "Existence of mild solutions for fractional non instantaneous impulsive integro-differential equations with nonlocal conditions", *Arab Journal of Mathematical Sciences*, DOI: 10.1016/j.ajmsc.2018.11.002.
- A. Meraj, D.N. Pandey, "Approximate controllability of fractional integrodifferential evolution equations with nonlocal and non-instantaneous impulsive conditions", *Journal of Fractional Calculus and Applications*, **10** (2) (2019), 3-17.
- 4. A. Meraj, D.N. Pandey, "Existence and uniqueness of mild solution and approximate controllability of fractional evolution equations with deformable fractional derivative", *Journal of Nonlinear Evolution Equations and Applications* (Accepted).
- A. Meraj, D.N. Pandey, "Monotone iterative technique for non-autonomous semilinear differential equations with nonlocal condition", *Demonstratio Mathematica*, **52** (1) (2019), 29-39.
- 6. A. Meraj, D.N. Pandey, "Existence and uniqueness of extremal mild solutions

for non-autonomous nonlocal integro-differential equations via monotone iterative technique", *Filomat* (Accepted).

 A. Meraj, D.N. Pandey, "Approximate controllability of non-autonomous Sobolev type integro-differential equations having nonlocal and non-instantaneous impulsive conditions", *Indian Journal of Pure and Applied Mathematics* (Accepted).

List of papers communicated in International Refereed Journals

 A. Meraj, D.N. Pandey, "Monotone iterative technique for non-autonomous semilinear differential equations with non-instantaneous impulses", communicated. (Springer Book Chapter through the ICAME'18)

Bibliography

- Abbas, S.: 2011, Existence of solutions to fractional order ordinary and delay differential equations and applications, *Elec. J. Diff. Eqn.* 9, 11 pp.
- [2] Abdeljawad, T.: 2015, On conformable fractional calculus, Journal of computational and Applied Mathematics 279, 57–66.
- [3] Abdeljawad, T., AL Horani, M. and Khalil, R.: 2015, Conformable fractional semigroups of operators, *Journal of Semigroup Theory and Applications* 2015.
- [4] Agarwal, R. P., Baleanu, D., Nieto, J. J., Torres, D. F. and Zhou, Y.: 2018, A survey on fuzzy fractional differential and optimal control nonlocal evolution equations, *Journal of Computational and Applied Mathematics* 339, 3–29.
- [5] Agarwal, S. and Bahuguna, D.: 2006, Existence of solutions to sobolev-type partial neutral differential equations, *International Journal of Stochastic Anal*ysis 2006.
- [6] Ahmad, B., Alsaedi, A., Assolami, A. and Agarwal, R. P.: 2013, A new class of fractional boundary value problems, *Adv. Diff. Equ* 273, 1–8.
- [7] Ahmad, B. and Nieto, J. J.: 2011, Riemann-liouville fractional integrodifferential equations with fractional nonlocal integral boundary conditions, *Bound. Value Probl.* 2011(1), 36.

- [8] Ahmad, B. and Ntouyas, S. K.: 2013, Existence results for higher order fractional differential inclusions with multi-strip fractional integral boundary conditions, *Electron. J. Qual. Theory Differ. Equ.* 20, pp–19.
- [9] Ahmad, B. and Ntouyas, S. K.: 2016, Some fractional-order one-dimensional semi-linear problems under nonlocal integral boundary conditions, *Revista de la Real Academia de Ciencias Exactas, Físicas y Naturales. Serie A. Matemáticas* 110(1), 159–172.
- [10] Akiladevi, K. S., Balachandran, K. and Kim, J.: 2014, Existence results for neutral fractional integro-differential equations with fractional integral boundary conditions, *Nonlinear Funct. Anal. Appl.* 19(2), 251–270.
- [11] Alam, A. and Imdad, M.: 2016, Monotone generalized contractions in ordered metric spaces, Bull. Korean Math. Soc 53(1), 61–81.
- [12] Alfaqih, W. M., Gubran, R. and Imdad, M.: 2018, Coincidence and common fixed point results under generalized (a, s) f-contraction, *FILOMAT* 32(7).
- [13] Arora, U. and Sukavanam, N.: 2017, Controllability of retarded semilinear fractional system with non-local conditions, IMA Journal of Mathematical Control and Information.
- [14] Atangana, A., Baleanu, D. and Alsaedi, A.: 2015, New properties of conformable derivative, Open Mathematics 13(1).
- [15] Avci, D., Eroglu, B. B. I. and Ozdemir, N.: 2017, Conformable heat equation on a radial symmetric plate, *Thermal Science* 21(2), 819–826.
- [16] Bahuguna, D. and Agarwal, S.: 2006, Approximations of solutions to neutral functional differential equations with nonlocal history conditions, J. Math. Anal. Appl. 317, 583–602.

- [17] Bai, L. and Nieto, J. J.: 2017, Variational approach to differential equations with not instantaneous impulses, *Applied Mathematics Letters* 73, 44–48.
- [18] Bai, L., Nieto, J. J. and Wang, X.: 2017, Variational approach to noninstantaneous impulsive nonlinear differential equations, J. Nonlinear Sci. Appl 10, 2440–2448.
- [19] Bai, Z. and Lu, H.: 2005, Positive solutions for boundary value problem of nonlinear fractional differential equation, J. Math. Anal. Appl. 311, 495–505.
- [20] Balachandran, K. and Dauer, J.: 2012, Elements of Control Theory, Alpha Science International.
- [21] Balachandran, K. and Park, J.: 2009, Controllability of fractional integrodifferential systems in banach spaces, Nonlinear Analysis: Hybrid Systems 3(4), 363– 367.
- [22] Balasubramaniam, P., Park, J. Y. and Kumar, A. V. A.: 2009, Existence of solutions of semilinear neutral stochastic functional differential equations with nonlocal conditions, *Nonlinear Anal.* **71**, 1049–1058.
- [23] Balasubramaniam, P., Vembarasan, V. and Senthilkumar, T.: 2014, Approximate controllability of impulsive fractional integro-differential systems with nonlocal conditions in hilbert space, *Numerical Functional Analysis and Optimization* 35(2), 177–197.
- [24] Banas, J. and Goebel, K.: 1980, Measures of noncompactness in Banach spaces, Lecture Notes in Pure and Applied Mathematics, 60. Marcel Dekker, Inc., New York.
- [25] Barenblat, G., Zheltor, J. and Kochiva, I.: 1960, Basic concepts in the theory of seepage of homogeneous liquids in fissured rocks, J. Appl. Mathe. Mech. 24, 1286–1303.

- [26] Barnett, S.: 1975, Introduction to mathematical control theory.
- [27] Bashier, E. B. and Patidar, K. C.: 2017, Optimal control of an epidemiological model with multiple time delays, *Applied Mathematics and Computation* 292, 47–56.
- [28] Bashir, A. and Nieto, J. J.: 2011, Riemann-liouville fractional integrodifferential equations with fractional nonlocal integral boundary conditions, *Bound. Value Probl.* 36, pp–9.
- [29] Bashirov, A. E. and Mahmudov, N. I.: 1999, On concepts of controllability for deterministic and stochastic systems, SIAM Journal on Control and Optimization 37(6), 1808–1821.
- [30] Batarfi, H., Losada, J., Nieto, J. J. and Shammakh, W.: 2015, Three-point boundary value problems for conformable fractional differential equations, *Journal of function spaces* 2015.
- [31] Benchohra, M., Henderson, J. and Ntouyas, S.: 2006, *Impulsive differential equations and inclusions*, Vol. 2, Hindawi Publishing Corporation New York.
- [32] Benchohra, M., Nieto, J. J. and Rezoug, N.: 2017, Second order evolution equations with nonlocal conditions, *Demonstratio Mathematica* **50**(1), 309–319.
- [33] Benchohra, M. and Slimani, B. A.: 2009, Existence and uniqueness of solutions to impulsive fractional differential equations, *Elec. J. Diff. Equ.* 10, pp–11.
- [34] Bensoussan, A., Da Prato, G., Delfour, M. C. and Mitter, S. K.: 1992, Representation and control of infinite dimensional systems, Vol. 1, Birkhäuser Boston.
- [35] Bernfeld, S. R., Ladde, G. S. and Lakshmikantham, V.: 1975, Existence of

solutions of two point boundary value problems for nonlinear systems, *Journal* of Differential Equations **18**(1), 103–110.

- [36] Bothe, D.: 1998, Multivalued perturbations of m-accretive differential inclusions, Israel Journal of Mathematics 108(1), 109.
- [37] Byszewski, L.: 1991, Theorems about the existence and uniqueness of solutions of a semilinear evolution nonlocal cauchy problem, *Journal of Mathematical* analysis and Applications 162(2), 494–505.
- [38] Byszewski, L. and Lakshmikantham, V.: 1990, Theorem about the existence and uniqueness of a solution of a nonlocal abstract cauchy problem in a banach space, *Applicable Anal.* 40, 11–19.
- [39] Cabada, A. and Wang, G.: 2012, Positive solutions of nonlinear fractional differential equations with integral boundary value conditions, J. Math. Anal. Appl. 389(1), 403–411.
- [40] Chaddha, A. and Pandey, D. N.: 2016, Faedo-galerkin approximation of solution for a nonlocal neutral fractional differential equation with deviating argument, *Mediterr. J. Math.* 13, 3041–3067.
- [41] Chadha, A. and Pandey, D. N.: 2014a, Existence of a solution for history valued neutral fractional differential equation with a nonlocal condition, J. Non. Evol. Eqn. and App 2, 13–28.
- [42] Chadha, A. and Pandey, D. N.: 2014b, Existence of the mild solution for implsive semilinear differential equation, *Inter. J. of Partial Diff. Eqn.* 2014, pp– 10.
- [43] Chadha, A. and Pandey, D. N.: 2014c, Existence results for an impulsive neutral fractional integrodifferential equation with infinite delay, *Int. J. Differ. Equ.* 2014, 10 pp.

- [44] Chadha, A. and Pandey, D. N.: 2014d, Existence results for an impulsive neutral integrodifferential equation with infinite delay via fractional operators, *Malaya J. Mat.* 3, 203–214.
- [45] Chadha, A. and Pandey, D. N.: 2015a, Existence and approximation of solution to neutral fractional differential equation with nonlocal conditions, *Comput. Math. Appl.* 69, 893–908.
- [46] Chadha, A. and Pandey, D. N.: 2015b, Mild solutions for non-autonomous impulsive semi-linear differential equations with iterated deviating arguments, *Electronic Journal of Differential Equations* 2015(222), 1–14.
- [47] Chadha, A. and Pandey, D. N.: 2015c, Periodic bvp for a class of nonlinear differential equation with a deviated argument and integrable impulses, *Cubo* (*Temuco*) 17(1), 11–27.
- [48] Chang, Y., Nieto, J. and Li, W.: 2009, Controllability of semilinear differential systems with nonlocal initial conditions in banach spaces, *Journal of Optimization Theory and Applications* 142(2), 267–273.
- [49] Chaudhary, R. and Pandey, D. N.: 2016, Monotone iterative technique for neutral fractional differential equation with infinite delay, *Mathematical Methods* in the Applied Sciences 39(15), 4642–4653.
- [50] Chaudharya, R. and Pandeya, D. N.: 2016, Existence results for nonlinear fractional differential equation with nonlocal integral boundary conditions, *Malaya J. Mat.* 4(3), 392–403.
- [51] Chen, P. and Li, Y.: 2011, Mixed monotone iterative technique for a class of semilinear impulsive evolution equations in banach spaces, Nonlinear Analysis: Theory, Methods & Applications 74(11), 3578–3588.

- [52] Chen, P. and Li, Y.: 2013, Monotone iterative technique for a class of semilinear evolution equations with nonlocal conditions, *Results in Mathematics* 63(3-4), 731–744.
- [53] Chen, P. and Li, Y.: 2014, Existence of mild solutions for fractional evolution equations with mixed monotone nonlocal conditions, *Zeitschrift für angewandte Mathematik und Physik* 65(4), 711–728.
- [54] Chen, P., Li, Y. and Yang, H.: 2013, Perturbation method for nonlocal impulsive evolution equations, Nonlinear Analysis: Hybrid Systems 8, 22–30.
- [55] Chen, P., Li, Y. and Zhang, X.: 2015, Existence and uniqueness of positive mild solutions for nonlocal evolution equations, *Positivity* 19(4), 927–939.
- [56] Chen, P. and Mu, J.: 2010, Monotone iterative method for semilinear impulsive evolution equations of mixed type in banach spaces., *Electronic Journal of Differential Equations (EJDE)*/*electronic only*/ 2010(149), 1–13.
- [57] Chen, P., Zhang, X. and Li, Y.: 2016, Existence of mild solutions to partial differential equations with non-instantaneous impulses, *Electron. J. Differ. Equ* 241, 1–11.
- [58] Chen, P., Zhang, X. and Li, Y.: 2017, Study on fractional non-autonomous evolution equations with delay, *Computers & Mathematics with Applications* 73(5), 794–803.
- [59] Chen, P., Zhang, X. and Li, Y.: 2018a, A blowup alternative result for fractional nonautonomous evolution equation of volterra type., *Communications on Pure* & Applied Analysis 17(5).
- [60] Chen, P., Zhang, X. and Li, Y.: 2018b, Fractional non-autonomous evolution equation with nonlocal conditions, *Journal of Pseudo-Differential Operators* and Applications pp. 1–19.

- [61] Chen, P., Zhang, X., Li, Y. et al.: 2017, Iterative method for a new class of evolution equations with non-instantaneous impulses, *Taiwanese Journal of Mathematics* 21(4), 913–942.
- [62] Curtain, R. F. and Zwart, H.: 2012, An introduction to infinite-dimensional linear systems theory, Vol. 21, Springer Science & Business Media.
- [63] Dabas, J. and Chauhan, A.: 2015, Existence and uniqueness of mild solution for an impulsive neutral fractional integro-differential equation with infinite delay, *Math. Comput. Modelling* 57, 754–763.
- [64] Dauer, J. and Mahmudov, N.: 2002, Approximate controllability of semilinear functional equations in hilbert spaces, *Journal of Mathematical Analysis and Applications* 273(2), 310–327.
- [65] De Andrade, B., Cuevas, C. and Soto, H.: 2016, On fractional heat equations with non-local initial conditions, *Proc. Edinb. Math. Soc.* 59, 65–76.
- [66] Deimling, K.: 2010, Nonlinear functional analysis, Courier Corporation.
- [67] Delbosco, D. and Rodino, L.: 1996, Existence and uniqueness for a nonlinear fractional differential equation, J. Math. Anal. Appl. 204, 609–625.
- [68] Deng, K.: 1993, Exponential decay of solutions of semilinear parabolic equations with nonlocal initial conditions, J. Math. Anal. Appl. 179, 630–637.
- [69] Du, S. W. and Lakshmikantham, V.: 1982, Monotone iterative technique for differential equations in a banach space, J. Math. Anal. Appl. 87, 454–459.
- [70] Dubey, S. A. and Bahuguna, D.: 2009, Existence and regularity of solutions to nonlocal retarded differential equations, *Appl. Math. Comput.* 215, 2413–2424.
- [71] Dubey, S. and Sharma, M.: 2014, Solutions to fractional functional differential equations with nonlocal conditions, *Fract. Calc. Appl. Anal.* 17, 654–673.

- [72] El-Borai, M., El-Nadi, K. E.-S., Ahmed, H., El-Owaidy, H., Ghanem, A. and Sakthivel, R.: 2018, Existence and stability for fractional parabolic integropartial differential equations with fractional brownian motion and nonlocal condition, *Cogent Mathematics & Statistics* 5(1), 1460030.
- [73] El-Borai, M. M.: 2004a, The fundamental solutions for fractional evolution equations of parabolic type, *International Journal of Stochastic Analysis* 2004(3), 197–211.
- [74] El-Borai, M. M.: 2004b, Semigroups and some nonlinear fractional differential equations, App. Math. Comp. 149, 823–831.
- [75] El-Borai; Mahmoud, M.: 2002, Some probability densities and fundamental solutions of fractional evolution equations, *Chaos Solitons Fractals* 14, 433– 440.
- [76] El-Sayed, A. M. A.: 1988, Fractional differential equations, *Kyungpook Math.* 28, 119–122.
- [77] El-Sayed, A. M. A.: 1992, On the fractional differential equations, Appl. Math. Comp. 49, 205–213.
- [78] El-Sayed, A. M. A.: 1996, Fractional order diffusion wave equation, Internat.
 J. Theoret. Phys. 35, 311–322.
- [79] Fattorini, H.: 1966, Some remarks on complete controllability, SIAM Journal on Control 4(4), 686–694.
- [80] Fattorini, H.: 1967, On complete controllability of linear systems, Journal of Differential Equations 3, 391–402.
- [81] Feckan, M., Zhou, Y. and Wang, J. R.: 2012, on the concept and existence

of solution for impulsive fractional differential equations, *Commun. Nonlinear Sci. Numer. Simul.* **17**, 3050–3060.

- [82] Friedman, A.: 2008, Partial differential equations of parabolic type, Courier Dover Publications.
- [83] Gautam, G. R. and Dabas, J.: 2015, Mild solutions for class of neutral fractional functional differential equations with not instantaneous impulses, *Appl. Math. Comput.* 259, 480–489.
- [84] Ge, F. D., Zhou, H. C. and Kou, C. H.: 2016, Approximate controllability of semilinear evolution equations of fractional order with nonlocal and impulsive conditions via an approximating technique, *Applied Mathematics and Computation* 275, 107–120.
- [85] George, R. K.: 1995, Approximate controllability of nonautonomous semilinear systems, Nonlinear analysis 9(24), 1377–1393.
- [86] Giona, M., Cerbelli, S. and Roman, H. E.: 1992, Fractional diffusion equation and relaxation in complex viscoelastic materials, *Physica A: Statistical Mechanics and its Applications* 191, 449–453.
- [87] Granas, A. and Dugundji, J.: 2003, Fixed point theory, Springer Monographs in Mathematics, Springer-Verlag, New York.
- [88] Gubran, R., Khan, I. A. and Imdad, M.: 2018, Common fixed point theorems for matkowski-type nonlinear contraction in ordered metric spaces, *Journal of Inequalities and Special Functions* 9(3), 34–47.
- [89] Guo, D. J. and Liu, X.: 1993, Extremal solutions of nonlinear impulsive integrodifferential equations in banach spaces, J. Math. Anal. Appl. 177, 538–552.

- [90] Haiping, Y., Jianming, G. and Yongsheng, D.: 2007, A generalized gronwall inequality and its application to a fractional differential equation, J. Math. Anal. Appl. 328, 1075–1081.
- [91] Haloi, R.: 2017, Approximate controllability of nonautonomous nonlocal delay differential equations with deviating arguments, *Electronic Journal of Differential Equations* 2017(111), 1–12.
- [92] Haloi, R., Pandey, D. N. and Bahuguna, D.: 2012, Existence, uniqueness and asymptotic stability of solutions to a non-autonomous semi-linear differential equation with deviated argument, *Nonlinear Dynamics and System Theory* 12(2), 179–191.
- [93] Hammad, M. A. and Khalil, R.: 2014, Abel's formula and wronskian for conformable fractional differential equations, *International Journal of Differential Equations and Applications* 13(3).
- [94] Heinz, H.-P.: 1983, On the behaviour of measures of noncompactness with respect to differentiation and integration of vector-valued functions, Nonlinear Analysis: Theory, Methods & Applications 7(12), 1351–1371.
- [95] Hernández, E. and O'Regan, D.: 2013, On a new class of abstract impulsive differential equations, *Proceedings of the American Mathematical Society* 141(5), 1641–1649.
- [96] Hernández, E., O'Regan, D. and Balachandran, K.: 2010, On recent developments in the theory of abstract differential equations with fractional derivatives, *Nonlinear Anal.* 73, 3462–3471.
- [97] Hernández, E. and Pierri, M.: 2018, On abstract neutral differential equations with state-dependent delay, *Journal of Fixed Point Theory and Applications* 20(3), 97.

- [98] Hristova, S. G. and Bainov, D.: 1989, Applications of monotone iterative techniques of v. lakshmikantham to the solution of the initial value problem for functional differential equations, *Le Mathematicae* 44, 227–236.
- [99] Hristova, S. G. and Bainov, D.: 1993, Applications of monotone iterative techniques of v. lakshmikantham to the solution of the initial value problem for impulsive differential-difference equations, *Rocky Moun. Jour. Math.* 23, 609– 618.
- [100] Jackson, D.: 1993, Existence and uniqueness of solutions to semilinear nonlocal parabolic equations, J. Math. Anal. Appl. 172, 256–265.
- [101] Jaiswal, A. and Bahuguna, D.: 2018, Semilinear conformable fractional differential equations in banach spaces, *Differential Equations and Dynamical Systems* pp. 1–13.
- [102] Ji, S.: 2014, Approximate controllability of semilinear nonlocal fractional differential systems via an approximating method, Applied Mathematics and Computation 236, 43–53.
- [103] Ji, S. and Li, G.: 2012, A unified approach to nonlocal impulsive differential equations with the measure of noncompactness, *Adv. Diff. Equ.* 182, pp–14.
- [104] Joshi, M. C. and George, R. K.: 1989, Controllability of nonlinear systems, Numerical Functional Analysis and Optimization 10(1-2), 139–166.
- [105] Kadalbajoo, M. K. and Patidar, K. C.: 2002a, Numerical solution of singularly perturbed non-linear two point boundary value problems by spline in compression, *International journal of computer mathematics* **79**(2), 271–288.
- [106] Kadalbajoo, M. K. and Patidar, K. C.: 2002b, Numerical solution of singularly perturbed two-point boundary value problems by spline in tension, *Applied Mathematics and Computation* 131(2-3), 299–320.

- [107] Kalman, R. E.: 1960, A new approach to linear filtering and prediction problems, Journal of basic Engineering 82(1), 35–45.
- [108] Kamaljeet and Bahuguna, D.: 2013, Controllability of the impulsive finite delay differential equations of fractional order with nonlocal conditions, Neural, parallel & scientific computations 21(3-4), 517–532.
- [109] Kamaljeet and Bahuguna, D.: 2015, Monotone iterative technique for nonlocal fractional differential equations with finite delay in banach space, *Electronic Journal of Qualitative Theory of Differential Equations* 2015(9), 1–16.
- [110] Kamaljeet and Bahuguna, D.: 2016, Approximate controllability of nonlocal neutral fractional integro-differential equations with finite delay, *Journal of Dynamical and Control Systems* 22(3), 485–504.
- [111] Kamaljeet, Bahuguna, D. and Shukla, R.: 2016, Approximate controllability of finite delay fractional functional integro-differential equations with nonlocal condition, *Differential Equations and Dynamical Systems* pp. 1–15.
- [112] Khalil, R., Al Horani, M., Yousef, A. and Sababheh, M.: 2014, A new definition of fractional derivative, *Journal of Computational and Applied Mathematics* 264, 65–70.
- [113] Khaliq, A. and ur Rehman, M.: 2018, On variational methods to noninstantaneous impulsive fractional differential equation, *Applied Mathematics Letters* 83, 95–102.
- [114] Kilbas, A. A., Srivastava, H. M. and Trujillo, J. J.: 2006, Theory and applications of fractional differential equations, North-Holland Mathematics Studies, 204, Elsevier Science B.V., Amsterdam.

- [115] Kumar, A., Muslim, M. and Sakthivel, R.: 2018, Controllability of the secondorder nonlinear differential equations with non-instantaneous impulses, *Journal* of Dynamical and Control Systems 24(2), 325–342.
- [116] Kumar, P., Haloi, R., Bahuguna, D. and Pandey, D.: 2016, Existence of solutions to a new class of abstract non-instantaneous impulsive fractional integrodifferential equations, Nonl. Dyn. and Systems Theory 73.
- [117] Kumar, P., Pandey, D. N. and Bahuguna, D.: 2014, On a new class of abstract impulsive functional differential equations of fractional order, J. Nonlinear Sci. Appl 7(2), 102–114.
- [118] Kumar, S. and Sukavanam, N.: 2013, Controllability of fractional order system with nonlinear term having integral contractor, *Fractional Calculus and Applied Analysis* 16(4), 791–801.
- [119] Ladde, G. S., Lakshmikantham, V. and Vatsala, A.: 1984, Existence of coupled quasi-solutions of systems of nonlinear elliptic boundary value problems, *Nonlinear Analysis: Theory, Methods & Applications* 8(5), 501–515.
- [120] Ladde, G. S., Lakshmikantham, V. and Vatsala, A. S.: 1985, Monotone iterative techniques for nonlinear differential equations, Monographs, Advanced Texts and Surveys in Pure and Applied Mathematics, Vol. 27, John Wiley and Sons, Inc., New York.
- [121] Lakshmikantham, V. and Du, S. W.: 1981, Monotone iterative technique for differential equations in a banach space, *Technical report*, University of Texas at Arlington.
- [122] Lakshmikantham, V., Simeonov, P. S. et al.: 1989, Theory of impulsive differential equations, Vol. 6, World scientific.

- [123] Li, F.: 2013, Existence for nonautonomous fractional integrodifferential equations with nonlocal conditions, *Abstract and Applied Analysis*, Vol. 2013, Hindawi.
- [124] Li, Y. and Liu, Z.: 2007, Monotone iterative technique for addressing impulsive integro-differential equations in banach spaces, *Nonlinear Analysis* 66, 83–92.
- [125] Liang, J., Liu, J. H. and Xiao, T.-J.: 2009, Nonlocal impulsive problems for nonlinear differential equations in banach spaces, *Mathematical and Computer Modelling* 49(3-4), 798–804.
- [126] Liang, S. and Mei, R.: 2014, Existence of mild solutions for fractional impulsive neutral evolution equations with nonlocal conditions, Advances in Difference Equations 2014(1), 101.
- [127] Liu, Z. and Li, X.: 2015, Approximate controllability of fractional evolution systems with riemann–liouville fractional derivatives, SIAM Journal on Control and Optimization 53(4), 1920–1933.
- [128] Lv, Z., Gong, Y. and Chen, Y.: 2017, Solvability for a class of evolution equations of fractional order with nonlocal conditions on the half-line, Advances in Difference Equations 2017(1), 382.
- [129] Mahmudov, N. and Denker, A.: 2000, On controllability of linear stochastic systems, *International Journal of Control* 73(2), 144–151.
- [130] Mahmudov, N. I.: 2001, Controllability of linear stochastic systems in hilbert spaces, Journal of mathematical Analysis and Applications 259(1), 64–82.
- [131] Mahmudov, N. I.: 2003, Approximate controllability of semilinear deterministic and stochastic evolution equations in abstract spaces, SIAM journal on control and optimization 42(5), 1604–1622.

- [132] Mahmudov, N. I. and Zorlu, S.: 2014, On the approximate controllability of fractional evolution equations with compact analytic semigroup, *Journal of Computational and Applied Mathematics* 259, 194–204.
- [133] Mahmudov, N. and McKibben, M.: 2015, On the approximate controllability of fractional evolution equations with generalized riemann-liouville fractional derivative, *Journal of Function Spaces* 2015.
- [134] Mahmudov, N. and Zorlu, S.: 2013, Approximate controllability of fractional integro-differential equations involving nonlocal initial conditions, *Boundary Value Problems* 2013(1), 118.
- [135] Mahto, L., Abbas, S. and Favini, A.: 2013, Analysis of caputo impulsive fractional order differential equations with applications, *Int. J. Diff. Equ.* 2013, 11 pp.
- [136] Mainardi, F.: 1997, Fractional Calculus: Some basic problems in continuum and statistical mechanics, in Fractals and Fractional Calculus in Continuum Mechanics, Springer, New York.
- [137] Malik, M., Abbas, S. and Kumar, A.: 2018, Existence and uniqueness of solutions of fractional order nonautonomous neutral differential equations with deviated arguments, *Journal of Nonlinear Evolution Equations and Applications ISSN* 2017(6), 81–93.
- [138] McKibben, M.: 2011, Discovering Evolution Equations with Applications: Volume 2-Stochastic Equations, Chapman and Hall/CRC.
- [139] McRae, F.: 2009, Monotone iterative technique and existence results for fractional differential equations, Nonlinear Analysis: Theory, Methods & Applications 71(12), 6093–6096.

- [140] Miller, K. S. and Ross, B.: 1993, An introduction to the fractional calculus and fractional differential equations, A Wiley-Interscience Publication. John Wiley and Sons, Inc., New York.
- [141] Mishra, I. and Sharma, M.: 2018, Approximate controllability of a non-autonomous differential equation, *Proceedings-Mathematical Sciences* 128(3), 34.
- [142] Mouffak, B., Henderson, J. and Ntouyas, S.: 2001, Impulsive neutral functional differential equations in banach spaces, *Applicable Analysis* 80(3-4), 353– 365.
- [143] Mu, J.: 2011, Monotone iterative technique for fractional evolution equations in banach spaces, *Journal of Applied Mathematics* 2011.
- [144] Mu, J.: 2012, Extremal mild solutions for impulsive fractional evolution equations with nonlocal initial conditions, *Boundary Value Problems* 2012(1), 71.
- [145] Mu, J. and Li, Y.: 2011, Monotone iterative technique for impulsive fractional evolution equations, *Journal of Inequalities and Applications* 2011(1), 125.
- [146] Murad, S. A. and Hadid, S. B.: 2012, Existence and uniqueness theorem for fractional differential equation with integral boundary condition, J. Frac. Calc. Appl. 3, pp–9.
- [147] Muslim, M., Kumar, A. and Fečkan, M.: 2018, Existence, uniqueness and stability of solutions to second order nonlinear differential equations with noninstantaneous impulses, *Journal of King Saud University-Science* **30**(2), 204– 213.
- [148] Naito, K.: 1987, Controllability of semilinear control systems dominated by the linear part, SIAM Journal on control and Optimization 25(3), 715–722.

- [149] Naito, K.: 1989, An inequality condition for approximate controllability of semilinear control systems, *Journal of Mathematical Analysis and Applications* 138(1), 129–136.
- [150] Nashine, H. K., Imdad, M. and Ahmadullah, M.: 2018, Common fixedpoint theorems for hybrid generalized (f, φ)-contractions under the common limit range property with applications, Ukrainian Mathematical Journal 69(11), 1784–1804.
- [151] Nieto, J. J. and Uzal, J. M.: 2018, Pulse positive periodic solutions for some classes of singular nonlinearities, *Applied Mathematics Letters* 86, 134–140.
- [152] Ntouyas, S. K.: 2013, Boundary value problems for nonlinear fractional differential equations and inclusions with nonlocal and fractional integral boundary conditions, *Opuscula Math.* 33, 117–138.
- [153] Nyamoradi, N., Baleanu, D. and Agarwal, R. P.: 2013, On a multipoint boundary value problem for a fractional order differential inclusion on an infinite interval, Adv. Math. Phys. Art. ID 823961, pp–9.
- [154] Ortigueira, M. D. and Machado, J. T.: 2015, What is a fractional derivative?, Journal of computational Physics 293, 4–13.
- [155] Park, J. Y., Han, H. and Kwun, Y.: 1998, Approximate controllability of second order integrodifferential systems, *Indian Journal of Pure and Applied Mathematics* 29, 941–950.
- [156] Pazy, A.: 1983, Semigroups of linear operators and applications to partial differential equations, Applied Mathematical Sciences, 44, Springer, New York.
- [157] Pierri, M., O'Regan, D. and Rolnik, V.: 2013, Existence of solutions for semilinear abstract differential equations with not instantaneous impulses, *Applied mathematics and computation* 219(12), 6743–6749.

- [158] Podlubny, I.: 1999, Fractional differential equations. An introduction to fractional derivatives, fractional differential equations, to methods of their solution and some of their applications, Mathematics in Science and Engineering, 198. Academic Press, Inc., San Diego, CA.
- [159] Qin, H., Zuo, X., Liu, J. and Liu, L.: 2015, Approximate controllability and optimal controls of fractional dynamical systems of order 1; q; $2 \ 1 < q < 2$ in banach spaces, *Advances in Difference Equations* **2015**(1), 73.
- [160] R., W. J., Feckan, M. and Zhou, Y.: 2011, on the new concept of solution and existence results for impulsive fractional evolution equations, *Dyn. Par. Diff. Equ.* 8, 345–361.
- [161] Radhakrishan, B. and Anukokila, P.: 2016, Controllability of second-order sobolev-type neutral impulsive integrodifferential systems in banach spaces, *Electronic Journal of Differential Equations* 2016(259), 1–18.
- [162] Ren, Y., Qin, Y. and Sakthivel, R.: 2010, Existence results for fractional order semilinear integro-differential evolution equations with infinite delay, *Int. Equ. Operator Theory* 67, 33–49.
- [163] Rouzkard, F., Imdad, M. and Gopal, D.: 2013, Some existence and uniqueness theorems on ordered metric spaces via generalized distances, *Fixed Point Theory and Applications* 2013(1), 45.
- [164] Sakthivel, R. and Anandhi, E. R.: 2010, Approximate controllability of impulsive differential equations with state-dependent delay, *International Journal* of Control 83(2), 387–393.
- [165] Sakthivel, R., Mahmudov, N. I. and Nieto, J. J.: 2012, Controllability for a class of fractional-order neutral evolution control systems, *Applied Mathematics* and Computation 218(20), 10334–10340.

- [166] Sakthivel, R., Ren, Y. and Mahmudov, N. I.: 2011, On the approximate controllability of semilinear fractional differential systems, *Computers & Mathematics with Applications* 62(3), 1451–1459.
- [167] Samko, S. G., Kilbas, A. A. and Marichev, O. I.: 1993, Fractional integrals and derivatives. Theory and applications, Gordon and Breach Science Publishers, Yverdon.
- [168] Samoilenko, A. M. and Perestyuk, N. A.: 1995, *Impulsive differential equa*tions, Monographs and Treatises, 14. World Scientific Publishing Co., Inc., River Edge, NJ.
- [169] Sharma, M. and Dubey, S.: 2015, Controllability of nonlocal fractional functional differential equations of neutral type in a banach space, International Journal of Dynamical Systems and Differential Equations 5(4), 302–321.
- [170] Shi, Y. et al.: 2016, A study on the mild solution of impulsive fractional evolution equations, Applied Mathematics and Computation 273, 465–476.
- [171] Shu, X.-B., Lai, Y. and Chen, Y.: 2011, The existence of mild solutions for impulsive fractional partial differential equations, *Nonlinear Analysis: Theory, Methods & Applications* 74(5), 2003–2011.
- [172] Shukla, A., Sukavanam, N., Pandey, D. and Arora, U.: 2016, Approximate controllability of second-order semilinear control system, *Circuits, Systems, and Signal Processing* 35(9), 3339–3354.
- [173] Sun, J. X.: 1990, Solutions of ordinary differential equations in banach spaces, Acta Math. Sinica 33, 374–380.
- [174] Sun, J. X. and Zhao, Z. Q.: 1992, Extremal solutions of initial value problem for integro-differential equations of mixed type in banach spaces, Ann. Differential Equations 8, 469–475.

- [175] Tai, Z. and Wang, X.: 2009, Controllability of fractional-order impulsive neutral functional infinite delay integrodifferential systems in banach spaces, Applied Mathematics Letters 22(11), 1760–1765.
- [176] Tamilalagan, P. and Balasubramaniam, P.: 2017, Approximate controllability of fractional stochastic differential equations driven by mixed fractional brownian motion via resolvent operators, *International Journal of Control* 90(8), 1713–1727.
- [177] Tisdell, C. C.: 2015, Maximal solutions to fractional differential equations, Fractional Differential Calculus 5(1), 79–85.
- [178] Toledano, J. M. A., Benavides, T. D. and Acedo, G. L.: 1997, Measures of Noncompactness in Metric Fixed Point Theory, Volume 99 of Operator Theory, Advances and Applications, Birkhäuser, Basel.
- [179] Triggiani, R.: 1975a, Controllability and observability in banach space with bounded operators, SIAM Journal on Control 13(2), 462–491.
- [180] Triggiani, R.: 1975b, On the lack of exact controllability for mild solutions in banach spaces, Journal of Mathematical Analysis and Applications 50(2), 438– 446.
- [181] Triggiani, R.: 1977, A note on the lack of exact controllability for mild solutions in banach spaces, SIAM Journal on Control and Optimization 15(3), 407–411.
- [182] Vijayakumar, V., Murugesu, R., Poongodi, R. and Dhanalakshmi, S.: 2017, Controllability of second-order impulsive nonlocal cauchy problem via measure of noncompactness, *Mediterranean Journal of Mathematics* 14(1), 3.
- [183] Wang, G., Ahmad, B., Zhang, L. and Nieto, J. J.: 2014, Comments on the concept of existence of solution for impulsive fractional differential equations,

Communications in Nonlinear Science and Numerical Simulation **19**(3), 401–403.

- [184] Wang, J., Fečkan, M. and Zhou, Y.: 2017, Approximate controllability of sobolev type fractional evolution systems with nonlocal conditions., *Evolution Equations & Control Theory* 6(3).
- [185] Wang, J. and Li, X.: 2014, Periodic bvp for integer/fractional order nonlinear differential equations with non-instantaneous impulses, *Journal of Applied Mathematics and Computing* 46(1-2), 321–334.
- [186] Xue, X.: 2005, Nonlinear differential equations with nonlocal conditions in banach spaces, Nonlinear Analysis: Theory, Methods and Applications 63, 575– 586.
- [187] Xue, X.: 2009, Nonlocal nonlinear differential equations with a measure of noncompactness in banach spaces, Nonlinear Analysis: Theory, Methods and Applications 70, 2593–2601.
- [188] Yakar, A. and Kutlay, H.: 2017, Monotone iterative technique via initial time different coupled lower and upper solutions for fractional differential equations, *Filomat* **31**(4), 1031–1039.
- [189] Yan, Z.: 2009, On solutions of semilinear evolution integrodifferential equations with nonlocal conditions, *Tamkang Journal of Mathematics* 40(3), 257– 269.
- [190] Yan, Z.: 2016, On a new class of impulsive stochastic partial neutral integrodifferential equations, Appl. Anal. 95, 1891–1918.
- [191] Yang, D. and Wang, J.: 2017, Integral boundary value problems for nonlinear non-instantaneous impulsive differential equations, *Journal of Applied Mathematics and Computing* 55(1-2), 59–78.

- [192] Yang, H., Agarwal, R. P. and Liang, Y.: 2017, Controllability for a class of integro-differential evolution equations involving non-local initial conditions, *International Journal of Control* 90(12), 2567–2574.
- [193] Yavuz, M. and Ozdemir, N.: 2018, A different approach to the european option pricing model with new fractional operator, *Mathematical Modelling of Natural Phenomena* 13(1), 12.
- [194] Yu, X. and Wang, J.: 2015, Periodic boundary value problems for nonlinear impulsive evolution equations on banach spaces, *Communications in Nonlinear Science and Numerical Simulation* 22(1-3), 980–989.
- [195] Zabczyk, J.: 2009, Mathematical control theory: an introduction, Springer Science & Business Media.
- [196] Zhang, X. and Chen, P.: 2016, Fractional evolution equation nonlocal problems with noncompact semigroups, *Opuscula Mathematica* 36.
- [197] Zhang, X., Li, Y. and Chen, P.: 2017, Existence of extremal mild solutions for the initial value problem of evolution equations with non-instantaneous impulses, *Journal of Fixed Point Theory and Applications* 19(4), 3013–3027.
- [198] Zhang, X., Zhu, C. and Yuan, C.: 2015, Approximate controllability of fractional impulsive evolution systems involving nonlocal initial conditions, Advances in Difference Equations 2015(1), 244.
- [199] Zhao, D. and Luo, M.: 2017, General conformable fractional derivative and its physical interpretation, *Calcolo* 54(3), 903–917.
- [200] Zhou, H. X.: 1983, Approximate controllability for a class of semilinear abstract equations, SIAM Journal on Control and Optimization 21(4), 551–565.

- [201] Zhou, H. X.: 1984, Controllability properties of linear and semilinear abstract control systems, SIAM Journal on Control and Optimization 22(3), 405–422.
- [202] Zhou, X. F., Yang, F. and Jiang, W.: 2015, Analytic study on linear neutral fractional differential equations, *Applied Mathematics and Computation* 257, 295–307.
- [203] Zhou, Y. and Jiao, F.: 2010a, Existence of mild solutions for fractional neutral evolution equations, *Comput. Math. Appl.* 59, 1063–1077.
- [204] Zhou, Y. and Jiao, F.: 2010b, Nonlocal cauchy problem for fractional evolution equations, Nonlinear Analysis: Real World Applications 11, 4465–4475.
- [205] Zulfeqarr, F., Ujlayan, A. and Ahuja, P.: 2017, A new fractional derivative and its fractional integral with some example, *arXiv preprint arXiv:1705.00962*